East African Scholars Multidisciplinary Bulletin

(An Open Access, International, Indexed, Peer-Reviewed Journal) A Publication of East African Scholars Publisher, Kenya www.easpublisher.com Engineering

Original Research Article

New Empirical Relations for Determination of Rock Slope Safety Factor in Fully Drained Conditions on Section RS06 of Block five in Sungun Copper Mine Mohammad Salehi Alashti^{1*}

¹Independent Researcher, Tehran, Iran

*Corresponding Author Mohammad Salehi Alashti Article History Received: 03.12.2018 | Accepted: 23.12.2018 | Published: 30.12.2018 DOI: 10.36349/easmb.2018.v01i03.005

Abstract: During the past centuries, more methods with many involved imperfections and limitations in various conditions suggested for slopes stability analysis specially based on different rock mass classification systems. However, RMi (rock mass index) classification system did not use for scrutinized assessments until now in accordance with its precise studies upon discontinuities system as joint characteristics, also block size as three-dimensional block volumes and intact rock strength to represent the rock mass uni-axial compressive strength. In this paper, for the first time, several empirical relations recommend only for the section RS06 in the block five of Sungun Copper Mine for 37° overall slope angle in fully drained conditions which presents some RMi applications in determining rock slopes safety factor in these situations to show multidimensional importance existence relations between factor of safety and RMi. Meanwhile, computed safety factors values from proposed equations compared with results of the limit equilibrium slope stability evaluation of SLOPE/W software. Using the earth's strength limit fundamental key parameter in offered formulas, which assays and indicates the major gravitational effects in slope stability procedures, increasing predictability of earth's materials behaviors that enabled us to calculate safety factor through RMi, simply, comprehensively and accurately in reasonable short-duration. Also, leads to better analyzing, showing moreover, understanding of the relationships between safety factor and RMi. Furthermore, this factor helps open new horizons and bridge between various sciences such as basis and applied physics, rock mechanics, rock engineering and engineering geology to indicating extended complicate correlations of all sciences' components that play role in the entire word wide. Research procedures in this article planned based on present mathematical relations via main properties of the slopes, earth and the universe. Low rootmean-square error, percent of variance, standard deviation and variation coefficient also great efficiency and correlation coefficient of calculated safety factor through new suggested empirical equations in its comparison with SLOPE/W results indicated high accuracy of proposed formulas.

Keywords: Strength Limit Number of Earth, Safety Factor Determination, RMi Classification System, Fully Drained conditions, Empirical Relations, Sungun Copper Mine, Slopes Stability Engineering, Earth Sciences

1. Introduction

Any part of existence such as key fundamental factors enable scientists to discover knowledge and solve the greatest challenges which faces to accelerate finding out new advanced intelligence researches, innovations and technologies that are changing lives every day.

The slope stability analysis has many applications in the engineering projects such as dams, roads, open pits, embankments, excavations, landfills and other engineering structures. These analyses studies carried out to prevent, eliminate and minimize the occurrences of slope failure, landslide or other results of instabilities. Slope stability investigation in open pit mines and design of these mines' walls at different stages of mining is important for the safe and economic mining operations.

The slope stability analyses usually implemented at the beginning, and sometimes throughout the lifetime of projects during planning, design, construction, improvement, rehabilitation, and maintenance by planners, engineers, geotechnical engineers, geologists, contractors, technicians, and maintenance workers, which become involved in this process [Adopted from 2].

Copyright @ 2018: This is an open-access article distributed under the terms of the Creative Commons Attribution license which permits unrestricted use, distribution, and reproduction in any medium for non commercial use (NonCommercial, or CC-BY-NC) provided the original author and source are credited.

Increasing construction of engineered cut and fill slopes in projects has necessitated to fully understanding analytical methods, investigative tools, and stabilization methods specialty construction techniques with contemplating limitations of them which must be modeled in realistic ways to solve slope stability problems have been faced.

Many scientists and researchers have focused on slopes stability assessment however, now slope instabilities problems are still presents an important and significant challenge for designers in mining, civil, rock mechanics, structural geology, geotechnical engineering and engineering geology [Adopted from 115, 166].

A common technique for the primary estimates of rock slopes stability even during implementing the work is empirical design using empirical methods and in the other words using of classification systems and empirical formulas [Adopted from 106]. Different rock mass classification systems used to assess the stability of slopes therewith for the design of slopes too. These systems worked based on empirical approaches, which means relations between rock mass parameters and various parameters of slopes such as height and dip of the slope [Adopted from 275]. Used calculation methods and parameters seem always not suited to any particular condition for slopes stability and should improve [Adopted from 106].

Construction materials commonly used in mining and civil engineering mostly characterized by their strength properties. This basic property of the material used in the engineering and design. In rock engineering, no such specific strength characterization of the rock mass is in common use. Most engineering carried out using various descriptions, classifications and un-quantified experience [221, 222]. Hoek and Brown (1980), Nieto (1983), Bieniawski (1984) and therefore several other authors have represented the necessities for rock masses' strength characterization. The Rock Mass index (RMi) developed to characterize the strength of the rock mass for construction purposes. The RMi is determined based on inherent parameters of the rock mass. Rock Mass index calculated by combining the intact rock compressive strength and jointing parameter which definitely and clearly represents the main jointing features, namely block volume (or density of joints) and joint characteristics as given by joint roughness, joint alteration and joint size. Using parameters in the RMi is an important issue in engineering [Adopted from 31, 117, 206, 221, 222].

In order to deficiency of different rock mass classification systems, must use a system which has a complete approach to slopes stability evaluation specialized in factor of safety determination also cover important and affecting items in calculations moreover emends limitations.

RMi classification system has many applications in different fields like rock mechanics and rock engineering as follows [Adopted from 221]:

- a) RMi parameters applied in practical design and engineering (rock design and engineering) such as: stability, rock support calculations, TBM progressive evaluation, rocks blasting and fragmentation.
- b) Application in systems for rock support evaluations such as: Q-system, RMR system and NATM.
- c) Applied to input in rock mechanics likes: hoek-brown failure criterion, numerical modeling, rock mass deformation modulus, ground response curves.
- d) Used as input data in other engineering methods.
- e) Applied in communication.

The Assessment of RMi classification system applicability in rock slopes stability is implementing for the first time by author. One part of these applications are presents related terms, relations and formulas between safety factors of slopes with rock mass index in fully drained conditions. In fact, the principle and particular goal of this scrutiny is present the relationship between rock slopes safety factor with RMi classification system in fully drained conditions. And present relevant equations and formulas for slope stability assessments only for section RS06 of block five of Sungun copper mine as the largest open-cast copper mine of Iran for 37° overall slope angle in response to: Does any relation between factor of safety and RMi?

Using of RMi classification system in these empirical formulas has some benefits [Adopted from 221, 223, 224]: (1) The three-dimensional block volume and joint parameters for rock masses uses in RMi determination that will generally meliorate the rock mass's characterization (description of the features of rock mass) and hence lead to better estimates.

- (2) Using of RMi classification system cover most important ground parameters that led to better assessments.
- (3) The RMi system covers a vast spectrum of rock mass variation. Therefore, mentioned system has eventualities for extensive usages rather than other today's rock mass classification and characterization systems.
- (4) The RMi can used for difficult estimates easily when access limited information about the ground condition.
- (5) The RMi suggests a suitable platform and framework for engineering judgments.
- (6) The RMi will give principal meaningful improvements in the use of geological input data.

(7) The RMi very likely describes also determine the qualities of a wider range of materials more than most classification systems.

(8) Relatively quick and inexpensive to implement.

Some of the main characteristics, which affecting on slopes failures that could considered for more comprehensive slopes' design are as follows:

1) Fundamental factors of nature: Gravity, etc.; 2) Geology: Geological Conditions; 3) State of all stresses: like as shear stresses, current stresses in the rock and soil; 4) State of strains: strain softening; 5) Geometrical data: Geometry of the slope: Height and angle of the slope, bench height and angle; 6) Slope Characteristics: Position of the slope; 7) Loads: External loading, applied loads, cyclic loading, additional loads at the top of the slope, seismic loadings (loads), groundwater load, soil net weight load, ongoing load on the surface, concentrated horizontal and vertical loads, loads coming from structures in the region, constant loads; 8) External forces: reinforcing elements, seismic accelerations; 9) Site characteristics: site surface and subsurface conditions, Site topography; 10) Geology: Geological information: properties of soil and rock mass (Soil parameters: soil cohesion, internal friction angle of the soil, the power functional relationship for the soil), Soil and rock strength (soil type, stratification); 11) Geology: Geological Discontinuities (the presence and of structural discontinuities) such as: Fault, Joint, bedding plane, crack (crack width); 12) Forming Materials: Slopes forming materials; 13) Characteristics of structural discontinuities: Discontinuity orientation, discontinuity spacing; 14) Alternation of materials by faulting, joint or discontinuity systems; 15) Movements and tension in joints; 16) Cracking; 17) Swelling; 18) Erosion;

19) Weathering, progressive weathering; 20) Leaching; 21) Decomposition of clayey rock fills; 22) Hydrology: Water: ground water conditions (a. Alter the cohesion and frictional parameters and b. Reduction of the normal effective stress), water content, drainage pattern, rainfall, permeability, aquifer, pore pressure, water pressure in cracks at the top of the slope, hydrostatic forces' effects; 23) Strength: shear strength (shear strength parameters), compressive strength, and tensile strength; 24) Geotechnical parameters: gran size, moisture content, atterberg limit, cohesion and unit weight etc.; 25) Method of construction: shovel, dumper, BWE or combination; 26) Possible effects of proposed construction; 27) Excavations: excavation at the bottom of the slope; 28) Disturbances: Excavation disturbances; 29) Critical geological conditions; 30) Dynamic forces: blasting; 31) Seismicity: Seismic velocity, seismic activities (earthquake activity); 32) Displacements: Displacements caused by earthquakes; 33) Temperature; 34) Creep: creep under sustained loads, progressive creep; 35) Brittle fracture; 36) Liquefaction: Liquefaction of weaker soil layers; 37) Environment: Environmental influencers; 38) Deformations: Internal deformation, deformation limit values; 39) Properties of other materials; 40) Lithology; 41) Geological Structure: Amount and direction of dip, Intra-formational shear zones, Joints and discontinuities (a. Reduction of shear strength, b. Change permeability, c. Act as sub surface drain and plains of failure), Faults (a. Weathering and alternation along the faults, b. Act as ground water conduits, c. Provides a probable plane of failure); 42) Planes of weakness; 43) Climate: Climatic conditions; 44) Mining Methods and Equipment; 45) Seepage; 46) Vegetation; 47) Intact rock strength ...

Because of the continuity of universe components there are different factors involve in cosmos and related process to them. Therefore, many parameters can use on FOS calculations, which act on a continuum procedure. In this paper, try to use some of the affecting and determining parameters in slope stability fields for FOS calculations. Strength limit factor of earth is one of these important parameters that help contemplate many other involving factors in scientific analysis potentially. One application of this factor will use to calculating safety factor of slopes on planets (on earth and space). All of the powers, coefficients and numbers serve in suggested formulas relevance with properties and characteristics of earth. In this paper, consider a new aspect in slope stability field that can extended these achieves to any sciences. Empirical relations suggested based on various parameters which shown in Table 8.

On the other hand, as the preliminary step toward to vast precise inter-disciplinary researches in sciences in the future, the main objectives in this study are showing the existence of relations between rock slopes engineering, rock mechanics and fundamental parameters and multi-disciplinary physics. Such as some of the existing relations between RMi rock mass classification system, factor of safety calculations and earth's strength limit key factor only for a defined section of one block of Sungun open pit mine under fully drained conditions.

The results provide a better understanding of the impact of strength limit of earth and RMi classification system in slope stability evaluations. In addition, results indicated that the suggested formulas have high accuracy in safety factor calculations, which contemplates safety, cost effectiveness, simplicity, time saving, widely adopting, easy understanding, training and calculating. Therewith, mentioned equations have no environmental impact also does not require to any equipment and assumptions. The obtained results by suggested relations for safety factor computation in fully drained conditions can then compare with any other assessment methods of slope stability even with result of any analysis

software. Using of RMi system in safety factor determination implies the unity, correlation and dependency all of the existence parts (like the universe).

In this paper, Section 2 introduces a background of previous slope stability studies. Section 3 illustrate limitations of current design approaches for slope stability. Section 4 presents brief specifications of Sungun copper mine. Section 5 demonstrates basilar theory and various definitions of strength limit factor of earth. In Section 6, express specifications of variable strength limit of earth. Section 7 explains determination of cohesion variable limit number for slopes. Section 8 represents new procedural empirical investigations of safety factor computation formulas by RMi classification system for fully drained condition. Section 9 clarifies the validity check of recommended equations in comparison with stability analysis results of limit equilibrium SLOPE/W software. Section 10 describes discussion and Section 11 explain conclusion about the relationships between the factor of safety and RMi classification system.

2. Previous Studies on Slopes Stability Analysis Procedures

2.1. Review of available slope stability researches approaches

2.1.1. Background of slope stability researches in rock masses

It noted that the jointed rock masses strength was difficult to evaluate notoriously. Generally, mentioned rock masses are expressive inhomogeneous, discontinuous media composed of rock material and naturally occurring discontinuities such as joints, fractures and bedding planes. It is very difficult to make any analysis in presence of these features using simple theoretical solutions, like the limit equilibrium method. Therewith, the displacement finite element method is not suitable for analyzing fractured, breached, interrupted and discontinuous rock masses without including special interface or joint elements [Adopted from 166]. Jaeger (1971) and Goodman and Kieffer (2000) have outlined several simple methods and corroborated their limitations to overcome the problem of rock slope strength estimating and stability due to complicated failure mechanisms. In addition, many criteria have proposed by Hoek and Brown (1980a), Yu et al. (2002), Grasselli and Egger (2003), Sheorey (1997), and Yudhbir et al. (1983) for estimating rock strength. Currently the Hoek-Brown failure criterion is one recognized approach, which used to rock mass strength estimation widely [116, 119]. Merifield et al. (2006) pointed out that the Hoek-Brown failure criterion is one of the few non-linear criteria, which used by engineers to estimate rock mass strength [Adopted from 100, 102, 116, 119, 132, 166, 184, 262, 326, 327].

2.1.2. Past slope stability researches with physical model tests

Ladanyi and Archambault (Ladanyi and Archambault, 1970; Ladanyi and Archambault, 1972) carried out some of the uttermost remarkable model tests with relevance to probable failure mechanisms in large-scale slopes. Two types of failures observed in the tests in presence of discontinuous joints. Shear failure along a well-defined failure surface was the first type and shear zone was the second type of formation. However, shear failure was the big difference, which did not occur along the concrete bricks' interfaces, but instead appeared as shear failure through the intact material [Adopted from 148, 149, 166].

Model experimentations using a mix of gypsum plaster, water and celite carried out by Einstein et al. (1970). The purpose was to simulate a brittle rock of relatively high strength, like as granite and quartzite. A considerable result from these tests was that the confining stress forcefully impressed the mechanisms of failure in the samples under triaxial stress loadings. Failure happened by the side of the pre-existing joints in low confining stress (less than 10 MPa). However, failure occurred principally all over the intact material in higher confining stress. The test conclusions moreover represented that the jointed samples' overall strength was lower than the model material's intact strength however, failure happened throughout the intact material [Adopted from 80, 166].

Stacey (1973) was tested several different joint configurations according to centrifuge tests. The results reveal that failure was not observed in the tests via the intact model material but failure happened as sliding along pre-existing joints. The above studies reported some variation in the results from different model tests. This fact can likely clarify with the differences in loading conditions and model material. The conditions of loading alter from alone gravitational loading to biaxial and triaxial loading [Adopted from 166, 282].

A review of the literatures shows that there are very few small-scale experimental results presented for rock slopes. This is probably to be happened/ true because Stewart et al. (1994) highlighted that hard rock modelling could necessitate large capacity centrifuge equipment. Stewart et al. (1994) in their study are used centrifuge modelling to investigate mechanisms of rock slope failure. The collapse mechanism evident in the model compared well with flexural toppling failure observed in the field. Furthermore, it found that the rock masses with high stiffness would fail in a brittle fashion [Adopted from 166, 287].

Adhikary et al. (1994) presented a set of rock slopes stability charts due to flexural toppling failures. These charts regulated based on the resultants from centrifuge tests and the limiting equilibrium method. However, Adhikary and Dyskin (2007) were found that the rock slope stability through the chart solutions tends to overestimate in the case of joint angles of less than 20°-25°. Furthermore, Adhikary and Dyskin (2007) demonstrated that the fractures could observe from the toe and (1) in high joint friction angle cases; fractures propagate back into the slope immediately and (2) in low joint friction angle cases, fractures distribute progressively back into the slope [Adopted from 3, 6, 166].

2.1.3. Past researches based on the limit equilibrium method

The limit equilibrium (LE) approach created and has developed since the 1930s, for circular and non-circular surfaces' slopes instabilities problems. Fellenius in 1936 presented the Swedish circle that can only use for circular slip surfaces. Bishop (1955) enlarged a revised method for circular slip surface assessment, which made better the accuracy of the FOS calculations and was suitable for automated computer methods. Janbu's method and Janbu's simplified method (1956) is usually used for non-circular slips surface. After that, the Morgenstern-Price (1965) contemplates the normal and tangential also the moment equilibrium for each slice in circular and non-circular slip surfaces. Janbu's Generalized Procedure of Slices (GPS) (1957, 1973) is applying vertical slices for any shape of slip-surface that consider vertical and horizontal force and moment equilibrium of the slices analysis also it is developed for non-circular slip surfaces through adopting a frictional center of rotation. Sarma's (1973) methods have since been proposed which accounted both force and moment equilibrium. These advanced LE methods improve the safety factor computations' exactitude [Adopted from 7, 39, 84, 134, 135, 136, 180, 194, 250, 280].

Currently, engineers trying to predict rock slopes stability using Hoek and Bray (1981) stability charts typically. These chart solutions take the water table in to account and are suited to uniform rock and rockfill slopes. In addition, Zanbak (1983) suggested a set of rock slopes stability charts suited for cases, which capable of accepting or permitting no resistance against toppling failure based on the limit equilibrium theorem. However, the conventional rock masses Mohr-Coulomb parameters (c' and ϕ ') or block interfaces are required as input for these two sets of chart solutions [Adopted from 115, 166, 328].

Sonmez et al. (1998) gained parameters of rock slope strength through back analysis of slope failures. In their study, rock mass classification's applicability and a practical procedure for estimating the mobilized shear strength based on the Hoek-Brown yield criterion clarified. They derived that determination of the shear strength is very difficult for jointed rock masses, particularly due to the scale effect [Adopted from 166, 278].

Chen et al. (2003) investigated rock slope stability under earthquake loadings through conducted a series of backanalyses for a case history. They found that could reasonably simulate the slope condition in the field when using the reduction factor of 2/3 for peak strength parameters. These peak strength parameters were attained from the laboratory testing in terms of c' and \emptyset' . Moreover, they indicated that the vertical ground acceleration was an important factor for inducing rockslide under near field conditions [Adopted from 52, 166].

Day and Seery (2007) in accordance to the failure mechanism's back-analysis emphasized that a major geological structure is the key factor, which controls slope failure. Thus, it cannot ignore that the slip surface follows the structural features in slope stability analysis. Harman et al. (2007) adopted a supposed case and investigated the permeability influences on the stability of rock slope. They found that the factor of safety increase with increasing permeability. As regards the fact that several types of rock have a low permeability, Harman et al. (2007) are indicated in their researches if these types to be and behave as intact rock then rock slopes should have no pore pressure and hydraulic continuity [Adopted from 67, 112, 166].

2.1.4. Past researches based on the numerical analysis

Zienkiewicz and Taylor (2000), Stead et al. (2001), Liu and Quek (2003), Hartmann and Katz (2007) efficiently contributed to cognizing theoretical background and the applications of numerical methods through computer software solutions [Adopted from 113, 171, 214, 286, 330]. Stead et al. (2001), Stacey et al. (2003), Singh (2011) and Nutakor (2012) applied numerical analysis for slope stability calculations [Adopted from 214, 215, 274, 283, 286].

Buhan et al. (2002) according to the numerical analysis found that the ultimate results of a slope stability analysis might affected by scale-effects in rock masses. Hoek et al. (2000), Wang C. et al. (2003), Eberhardt et al. (2004), and Stead et al. (2006) have been used a range of numerical methods in previous assays of progressive failures and/or rock slopes safety factor assessment. These include the continuum methods (finite element method and the finite difference method), the discontinuum methods (distinct element and discontinuous deformation analysis), and finite-discrete-

element codes. Specifically, Elmo et al. (2007) study modelled a large-scale open pit mine in 2D and 3D analyses using finite-discrete-element codes. It certified that 3D large scale analysis of the fracturing process is currently limited because of the weak memory and processing capacity of computer hardware [Adopted from 43, 78, 81, 122, 166, 285, 305].

Stewart et al. (1994) applied the finite difference method (FDM) to investigate rock slope stability. In their study, the strain softening found to be an important factor in some slope stability situations. Adhikary et al. (1995) obtained stress concentration pattern around the slope's toe for the cases of frictionless joints from the finite element analysis. Hence, the mechanism of failure is progressive for the slope with frictionless joints. In addition, they implied that the frictional sliding along the joints tended to re-propagate the instant stresses rather evenly over a large area. This area expands inside from the slope's toe and thus, the slope fails immediately. Adhikary and Dyskin (2007) in their study found that the joint friction plays the greatest importance role in the toppling failure mechanism. However, the joint cohesion does not have a same result on the mechanism of failure [Adopted from 4, 6, 166, 287].

Clough and Woodward (1967), Kulhawy and Duncan (1972), Adikari et al. (1982) and Veiga Pinto and Neves (1985) in their analyses was used the finite element method [Adopted from 5, 57, 147, 302].

2.1.5. Past researches according to limit analysis theorems

Siad (2003) created 2D charts in accordance to the upper bound approach, which can apply for rock slopes with earthquake effects. A range of parameters includes slope angle, joint inclination, shear strength of rock masses, joints, etc. considered in this analysis study. In addition, Chen et al. (2001a) investigated about 3D slope stability of rocks. In their study, the critical failure mode can be found by optimization routines; however, the failure surface requires still to be assumed in advance. Moreover, the presented solutions in the above studies need conventional Mohr-Coulomb soil parameters, cohesion (c') and friction angle (\emptyset '), as input data [Adopted from 54, 166, 265].

Tangential strength parameters (C_t and ϕ_t) from the $\sigma - \tau$ planes of nonlinear failure criteria adopted by Collins et al. (1988), Drescher and Christopoulos (1988) and Yang et al. (2004a) for slope stability estimations. After the Yang et al. (2004b) study, the latest version of the Hoek-Brown failure criterion used to conduct slope stability analyses. The effects of the seismic loadings [321-323] and pore pressure [324] on the rock slope stability considered. The studies of Yang et al. (2004b), Yang et al. (2004b) and Yang and Zou (2006) indicate the only endeavor at being preparing slope stability factors to estimating rock slope stability [Adopted from 59, 76, 166, 321, 322, 323, 324].

2.1.6. Previous slope stability researches by the pseudo static (PS) method

Seed (1979) tries to estimate stability of rock slope with contemplating of earthquake effects. Seed (1979) proposed the adopted seismic coefficient versus Richter's earthquake magnitude based on the PS analysis. Luo et al. (2004) found that ground water maybe considerably decrease stability of slopes during earthquakes which excite (or stimulate) surrounding ground of slopes where the maximum seismic coefficient obtained by up to 60% changes. Sepúlveda et al. (2005a) indicated that the topographic amplification effects such as slope orientation and seismic wavelength might influence the stability assessment of rock slopes. Chen et al. (2003) in their case study found the vertical ground acceleration as to be an important factor leading to rockslide under near field conditions [Adopted from 52, 166, 173, 251, 258].

PS method applied and extended to evaluate the induced ground movement through an earthquake by Newmark (1965). Sepúlveda et al. (2005b), Huang et al. (2001) and Ling and Cheng (1997) has been accepted this approach and extensively used to study earthquake triggered landslides and rockslides. Pradel et al. (2005) in particular, obtained a good agreement of slope crest displacement between the calculated and observed results. In their study, used strength parameters in analyses specified via repeated direct shear testing and back analysis [Adopted from 124, 166, 169, 203, 231, 259].

Newmark (1965), Seed (1979), Baker et al. (2006), Ling et al. (1997), and Loukidis et al. (2003) has been applied the PS approach in number of investigations, mainly due to its simplicity. Baker et al. (2006) and Loukidis et al. (2003) particularly provide chart solutions for soil slopes through have adopted the PS method respectively in limit equilibrium analysis and limit analysis. By using complicated dynamic response analysis coupled with appropriate constitutive laws, a more precise seismic evaluation for slopes can obtained. However, the PS method is still applicable and recommended as a screening procedure to identify any requirement for more sophisticated dynamic analyses. Although Cotecchia (1987) and Kramer (1996) represented that the PS approach has a number of limitations, generally conservative approach purposed as to be the considered methodology. This approach is the one most often used in current practice [Adopted from 16, 61, 146, 166, 170, 172, 203, 251].

The seismic coefficients in generally are determined from experience by using the maximum horizontal acceleration or peak ground acceleration of a designed earthquake. It should consider that, current design employing PS method is mostly according to a horizontal seismic coefficient (k_h). Therefore, this research is firstly centralizing on investigating the effects of earthquakes on stability of rock slopes by applying a range of horizontal seismic coefficients. Seed (1979) proposed that the PS analysis was applicable in evaluating the efficiency of embankments, which built from materials that do not endure considerable strength loss during earthquakes by reference to the magnitude of k_h [Adopted from 166, 251]. It is recommended to utilize k = 0.1 for earthquakes of Richter's magnitude 8.5. For both cases, a safety factor $F \ge 1.15$ is required for design [166, 251].

The Hynes-Griffin and Franklin's (1984) suggestion proposed one of the extensively used and accepted methods for determining a suitable value of k_h . They recommended that a PS analysis could apply for preliminary slope stability evaluation, where a seismic coefficient equal to one-half, the measured bedrock acceleration adopted. If the obtained safety factor is greater than 1.0, then the provided slope design can accept. Hynes-Griffin and Franklin (1984) proposed that a more thorough numerical analysis need to be performed for safety factors of less than 1. Because the magnitude of k_h is related to the measured bedrock acceleration as discussed above. However, the PS method may not account for the site amplification induced by the underlain stratum [28] or topography [259] etc. [Adopted from 28, 131, 166, 259].

The California Division of Mines and Geology (1997) summarized a diagram to select an appropriate PS coefficient for a given site. This provides the recommendations in regard to the seismic coefficient (k_h), versus a required factor of safety [Adopted from 44, 166]. From this diagram, it can recognize that the recommended k_h values do not exceed 0.375. Therefore, the range of the seismic coefficients adopted in the present study will be between $k_h = 0.0$ and $k_h = 0.375$ [44, 166].

2.1.7. Kinematic analysis

Hoek and Bray (1981), was explained the kinematic analysis method, which developed by Goodman (1989) and modified by Wyllie and Mah (2004). The investigation of potential planar, wedge and toppling failure conceives through this method via assumes that only friction involved in the sliding surface' shear strength and the cohesion is zero [Adopted from 99, 115, 316].

2.1.8. Three-dimensional analysis

Anagnosti (1969) more advanced the safety factor specification of the potential sliding mass for different shapes, which is a development of the 2-D Morgenstern-Price method (1967), via setting equilibrium equations for the series of thin vertical slices and assuming limit equilibrium conditions on sliding sides of each slice. Hovland (1977) approach is a development of the assumptions involved in the two-dimensional ordinary method of slices but columns used instead of slices. This is inaccurate because it assumes zero normal stress on vertical surfaces. Chen (1981) and Chen and Chameau (1982) according to extended Spencer and finite element methods and Chen and Chameau (1983) based on developed Spencer method expanded a pervasive study of the three-dimensional effects on the stability of slopes for vast types of soil parameters. Baligh and Azzouz (1975), also Baligh, Azzouz, and Ladd (1977) extended the concept of the twodimensional circular arc method for simple loaded slopes with revolution slip surfaces to evaluate the end effects of the three-dimensional slip surface developed in a cohesive slope. Azzouz and Baligh (1976) present design charts for threedimensional stability of cohesive slopes subjected to surcharge loads. Azzouz, et al. (Azzouz, Baligh, and Ladd) (1981) extended 3-D slope stability analysis based on extended Swedish circle for real embankments with slip surfaces of revolutions. Azzouz and Baligh (1983) developed 3-D stability analysis for slopes with loads on top in accordance to extended Swedish circle. Leshchinsky, Baker and Silver (1985) is according to limit equilibrium and variational analysis propound a 3-D mathematical approach for slope stability analysis, which represented by Kopacsy (1957). They proponed a quantified formulation of a given slope's safety margin relative to its available shear strength and therefore allowed a limiting condition's application (i.e., the adjusted Coulomb's failure criterion) for stable slopes. Hungr (1987) uses a microcomputer program (CLARA-3) (based on extended Bishop's modified) suggested a 3-D method which is a direct development of the assumptions accompanied by Bishop's (1954) 2-D simplified method. Hungr et al. (1989) applied a 3-D way, which was a development of the assumptions in Bishop's (1954) simplified and simplified twodimensional Janbu models, and they displayed comparisons for number of solutions. Other efforts on developing methods of three-dimensional analysis of slopes stability are as follows:

1) Giger and Krizek (1975) and Giger and Krizek (1976) based on upper bound theory of perfect plasticity. 2) Hutchinson and Sarma (1985) based on three-dimensional limit equilibrium. 3) Dennhardt and Forster (1985) based on assumed S on slip surface. 4) Ugai (1985), Leshchinsky and Baker (1986) and Baker and Leshchinsky (1987) based on limit equilibrium and variational analysis. 5) Cavounidis (1987) based on limit equilibrium. 6) Leshchinsky and Mullet (1988a, 1988b) based on limit equilibrium and variational analysis. 6) Gens et al. (1988) based on extended Swedish

circle. 7) Xing (1988a, 1988b) based on limit equilibrium. 8) Ugai (1988) based on extended ordinary method of slices. 9) Bishop's modified, Janbu and Spencer, Michalowski (1989) based on kinematic theorem of limit plasticity. 10) Seed et al. (1990) based on Ad hoc 2-D and 3-D. 11) Leshchinsky (February 1992) and Leshchinsky and Huang (October 1992, November 1992) based on limit equilibrium and variational analysis. 12) Cavounidis and Kalogeropoulos (1992) based on 3D method. 13) Azzouz and Baligh (1978) and Lam and Fredlund (1993) based on 2D general limit equilibrium. 14) Yamagami and Jiang (1996, 1997) based on Simplified Junbu's methods (1954). 15) Huang and Tsai (2000) based on limit equilibrium and two-directional factor of safety. 16) Huang et al. (2002) based on Junbu's method and two-directional factor of safety. 17) Chen et al. (2003) based on Spencer's method. 18) Jiang and Yamagami (2004) based on Spencer's method according to variational analysis. 19) Cheng and Yip (2007) based on Bishop's, Junbu's and Morgenstern- Price's methods. 20) Zheng (2009) based on limit equilibrium. 21) Sun et al. (2012) based on Morgenstern-Price's method (for generalized slip surface) [Adopted from 7, 8, 9, 10, 11, 12, 15, 17, 18, 38, 45, 46, 48, 49, 50, 53, 55, 75, 91, 93, 94, 123, 125, 126, 128, 129, 130, 133, 137, 139, 144, 150, 158, 159, 160, 161, 162, 163, 164, 185, 195, 252, 289, 295, 296, 317, 318, 319, 320, 329].

2.1.9. Other investigations

Goldscheider et al. (2010), Martens et al. (2011), Petri and Stein (2012) and Nutakor (2012) consider the slope stability analysis using slices [Adopted from 96, 179, 214, 228].

Baker and Gaber (1978) applied the variational calculus to locating the critical slip surface. Baker (1980) utilized dynamic programming for this purpose. Celestino and Duncan (1981) also Li and White (1987) implemented critical noncircular slip surface through alternating variable ways. The Monte Carlo technique used by Greco (1996) and Malkawi et al. (2001) to characterize the critical slip surface. Goh (1999), McCombic and Wilkinson (2002), Das (2005), and Zolfaghari et al. (2005) represented the application of genetic algorithm for slope stability analysis in critical surface recognition [Adopted from 13, 14, 47, 66, 95, 103, 138, 167, 174, 182, 331].

Yarahmadi Bafghi and Verdel (2005) and Hack et al. (2003) was applied the probabilistic analytical method to identify the rock slope potential failure key-group and estimate the probability of failure. It must be attended that rock slope stability evaluations via the Slope Stability Probability Classification that proposed by Hack et al. (2003) does not require cohesion and friction angle factors as input for assessment. Furthermore, Wang et al. (2000) and Hack et al. (2007) predicted the failure risk of rock slopes and investigated the influence of earthquakes on rock slope stability through reliability analysis [Adopted from 107, 109, 166, 306, 325].

2.2. Empirical modeling

Analytical, empirical, and numerical are three design strategies in rock engineering. Empirical methods (i.e. rock mass classification,) are widely applied for feasibility and pre-design studies, also mostly used for the final design [Adopted from 36, 106, 225, 226, 238, 268, 269].

2.2.1. Rock Classification Systems

Rock mass classification systems applied in various engineering design and stability analysis projects. These systems created based on relations between rock mass parameters and engineering applications, such as tunnels, slopes, foundations, and other excavations empirically. The marcher step gaits in rock mass classification as the first system in geotechnical engineering field for steel set support of tunnels that suggested in 1946 [Adopted from 36, 106, 225, 226, 238, 268, 269].

In order to rock mass complexity, many researchers try to involve rock mass parameters in rock slopes designs and generate specific relation between them. Many empirical methods (rock mass classification systems) adjusted and generally these are using in primary designs techniques and sometimes use in final design such as: RMR, GSI, SMR and CSMR

Some of these classification systems, basically developed for underground excavations, also have been used for slopes (e.g., Q and RMR system) or have been modified for slopes (e.g., the RMS, SMR, SRMR and CSMR systems comprise modifications of the RMR system) [Adopted from 226].

In the recent several decades, rock mass classification systems have been propounded in accordance to cuttings with high risk of failure are recognized, also in this status preventative measures prioritized effectively [Adopted from 226].

Classification systems have played an indispensable role in engineering for centuries [29, 36]. The rock mass classification today forms an essential part of the most common design approach, the empirical design methods.

However, modern rock classifications have been never intended as the ultimate solution to design problems, but only a means towards this end [Adopted from 29, 36, 279].

Field engineers through the years have been attempting to describe the ground condition using the rock or rock mass properties such as petrologic descriptions, general rock type, or one or a few of the physiomechanical (physico-mechanical) properties. As a result, several methods have come into usage describing the same rock in different ways. Most of the earlier systems were "intact rock classifications", that is, systems based on laboratory properties determined on a sample of rock. On the other hand, "rock mass classifications" consider discontinuities and large-scale ground features [279]. The rock mass classifications main objectives (After Bieniawski 1989) categorized to [Adopted from 279]:

- 1. Find the most significative parameters which affecting on rock mass behavior [Adopted from 279].
- 2. Divide a particular rock mass formation into groups of similar behavior, that is, rock mass classes of varying quality [279].
- 3. Provide a basis for understanding the characteristics of each rock mass class [279].
- 4. Relate the experience of rock conditions at one site to the conditions and experience encountered at others [279].
- 5. Derive quantitative data and guidelines for engineering design [279].
- 6. Provide common fundament to communicate between engineers and geologists [Adopted from 279].

In addition, the rock mass classifications' principal advantages are as follows [Adopted from 36, 106, 225, 226, 238, 268, 269]:

• Making better the qualitative conditions of site investigations by using the least input information as classification factors.

- Preparing quantitative data for design aims.
- Enabling better engineering judgement
- Providing more efficient communicating in engineering projects.

Rock mass classification is a tool to deliberate the performance of rock cut slopes based on the most important inherent, fundamental and structural parameters [Adopted from 101]. The classification systems must be providing flexible consistent means to describing the rock mass conditions qualitatively and quantitatively even under uncertainty conditions. A detailed list of the existing developed empirical rock mass classification methods presented in Table 1.

Deere et al. (1967) developed the Rock Quality Designation index (RQD) [Adopted from 72, 304].

Bieniawski presented a rock mass classification system for stability assessment of cuttings in 1974. In this work, a "rough" adjustment factor which represents discontinuity orientation was added to the basic Rock Mass Rating (RMR) system's five pre-existing parameters to make it appropriate for slope stability problems. Since then, a number of rock mass classification systems have been proposed [Adopted from 30, 36, 226].

RMR had deficiencies in assessment of very poor quality rock masses. Stille et al. (1982) offered a modification of the RMR system that named as the rock mass strength (RMS) classification. Romana (1985) was introduced Slope Mass Rating (SMR) to assess slopes' behaviour. Hoek et al. (1995) was reported the geological strength index (GSI) as a complement of their generalized rock failure criterion. Hoek et al. (2002) was implemented several minor revisions for the original GSI [Adopted from 119, 120, 239, 288, 304].

In addition, Varnes proposed landslide classification that modified by Cruden and Varnes has been adopted [Adopted from 64, 65, 226, 299, 300].

Rock mass classification systems use particularly factors relevant to the condition of cuttings. Hence, further comparison with any rating systems of rock fall hazard or risk assessment system would not be appropriate, as the first consist of both hazard and consequence factors and the second estimates risk for failure instead of hazards for failure [226]. Substantial caution can exercise in using RMi classification system (same as others) to all rock-engineering problems such as slope stability problems accordance to above review.

Table 1: Detailed list of the existing developed empirical rock mass classification methods - modified and
augmented [Adopted from 226 and modified by 60, 105, 106, 223, 224, 235, 249].

Name of the system	Abbreviation	Authors	Application	Comments	References
-	-	Ritter	Tunnels	The preliminary endeavor to formalize an empirical approach for the design of tunnel (1879).	Pantelidis (2009); Ritter (1879); Grasselli and Egger (2003).
Rock load	-	Karl Von Terzaghi	Tunnels (Tunnels with steel Support)	The rudimental resource to the application of rock mass classification to tunnel support design (1946).	Adopted from Pantelidis (2009); Terzaghi (1946); Kulhawy and Duncan (1972); Hack (1998).
Stand-up time	-	Lauffer	Tunnels	Linked to an unsupported tunnel excavation's stand- up time (1958), Applications in tunneling design.	Adopted from Pantelidis (2009); Lauffer (1958); Pantelidis (2009); Pantelidis (2009); Pantelidis (2009); Hack (1998).
Rock Quality Designation	RQD	De Deere et al.	General , core logging, tunneling	Incorporator parameter in many classification systems (1967, 1988, 1989).	Adopted from Pantelidis (2009); Pantelidis (2009); Deere (1963); Deere (1964); Deere (1989); Deere DU and Deere DW (1988); Deere, Hendron, Patton and Cording (1967); Deere and Miller (1967); Pantelidis (2009); Pantelidis (2009); Hack (1998).
New Austrian Tunneling Method	NATM	Rabcewicz, Muller and Pacher, Rabcewicz and Golser, Pacher et al, Pacher, Muller, Kovari	Tunnels	The system solely designed for tunneling (Rabcewicz, 1964/1965, 1972; Muller and Pacher 1964; Pacher et al., 1974; Muller, 1978; Kovari, 1993). (Rabcewicz and Golser 1973; Pacher, 1975), Uses in excavation and design of incompetent (overstressed) ground.	Hack (1998); Hack (1998); Hack (1998); Kovari (1993); Müller (1978a); Müller (1970); Muller and Pacher (1964); Pacher (1975); Pacher, Rabcewicz and Golser (1974); Rabcewicz (1964); Rabcewicz and Golser (1973); Hack (1998); Hack (1998).
Rock classification for rock mechanical purposes	-	Patching and Coates	General	Application as input in rock mechanics. Descriptive (1968).	Patching and Coates (1968).
The unified classification of rocks(and soils)	-	Deere et al.	General	Descriptive. According to particles and blocks for communication (1969).	Deere, Merritt and Coon (1969).
Rock Structure Rating (Rock Structure concept)	кэк	wickham et al.	Small tunnels	First rock masses rating system Developed by Wickham, Tiedemann and Skinner in the 1970s (1972,	Adopted from Pantelidis (2009); Wickham, Tiedemann and

Size Strength Classification of rock masses (Strength Hlock size)FranklinTunnel stability, MiningDevelopment of a slope strength (classification) rock masses. (Strength Hlock size)FranklinTunnel stability and potential failure or cox strength and block (1970); Franklin (1970); Franklin (1971); Franklin (1972); Franklin (1974), intersection added in the 1979 version added in the 1979 version added in the 1979 version (1980).A rum rating regulation, not fite King steam (1972); Bieniawski (1973); Bieniawski (1973); 					1974).	Skinner (1972);
Size Strength Classification for rock masses (Strength Block size)-FranklinTunnel stability, MiningDevelopment of a slope stability dassification system for runnel stability, and potential failure mechanism assessment (1971a); Cottiss, 1071a); Cottiss, 1071a); Cottiss, 1071a); Cottiss, 1071a); Cottiss, 1071a); Cottiss, 1071a); Cottiss, 1071a); Cottiss, 1071b; Franklin (1971b; Franklin (1971b; Franklin, (1973); Franklin (1973); Franklin (1973); Franklin, (1974); Franklin, (1975); Franklin, (1974); Franklin, (1975); Franklin, (1974); Franklin, (1975); Franklin, (1974); Franklin, (1975); Franklin, (1974); Franklin, (1974); Franklin, (1974); Fr						Wickham,
Size Strength Classification for rock masses (Strength - Block size)FranklinTunnel stability, MiringDevelopment of a slope study leassification mechanisms assessment, (1970): Franklin, 1980). Accords. Dowel and Franklin (1970): Franklin, 1980). Accords (1970): Franklin, 1980). Accords (1970): Franklin, 1980). Accords (1970): Franklin, 1980). Accords (1970): Franklin, 1980). Accords (1971): Franklin, 1980). Accords (1976): Barnowski (1973); Franklin, 1980). Accords (1976): Barnowski (1973); Franklin, 1980). Accords (1976): Barnowski (1973); Breinwski (1973); Breinwski (1973); Breinwski (1973); Breinwski (1976); Barnowski (1976); Barnowski (1976); Barnowski (1988); Breinwski (1976); Barnowski (1976						Tiedemann and
Size Strength Classification for rock masses (Strength - Block size)FranklinTurnel stability, MiningDevelopment of a slope stability classification system for tunnel stability, and potential failure metchanism savessment (1971b); Franklin (1971b); Franklin (1972b); Franklin (1972b); Franklin (1972b); Franklin (1972b); Franklin (1972b); Franklin (1972b); Franklin (1972b); Franklin (1972b); Franklin (1972b); Franklin (197						Skinner (1974); Hack
Size Strength rock masses (krength - Block - Franklin Tunnel stability wing Development of a slope system for tunnel stability apport of a slope to rock strength and block different for akkin (970); Franklin (970); Franklin (971); Franklin (973); Franklin (973); Franklin (974); Breniwski (973); Breniwski (976); Breniwski (976); Breni						(1998).
Classification for rock masses (Strength - Block size)Near end (Strength - Block size)Miningstability classification (Strength - Block size)Nowel and Franklin (1971a); Cratkin, Pranklin (1970); Franklin, (1970); Franklin, (1970); Franklin, 1971; Franklin, 1971; Franklin, 1971; Franklin, 1971; Franklin, 1975; Franklin, 1976; Franklin, 1970; (1976); Franklin, 1971; Franklin, 1971; Franklin, 1971; Franklin, 1974; Fran	Size Strength	-	Franklin	Tunnel stability,	Development of a slope	Hack (1998); Cottiss,
rock masses (Strength - Block Size) - Block Size	Classification for			Mining	stability classification	Dowel and Franklin
(Strength - Block size)and potential failure nerock strength and book (1971); Franklin, 1970; (1972); Franklin, 1970; (1973); Franklin, 1970; (1974); Franklin, 1970; (1974); Franklin, 1970; (1975); Franklin, 1970; (1975); Franklin, 1970; (1975); Franklin, 1970; (1974); Franklin, 1970; (1975); Franklin, 1970; (1974); Franklin, 1970; (1974); (1974); Franklin, 1970; (1974); (1974); (1974); Franklin, 1970; (1974); (1975); (1976); (1976); (1976); (1976); (1976); (1976); (1976); (1976); (1976); (1976); (1976); (1976); (1976); (1976); (1976); (1976); (1976); (1976); (1976); (1977); (1976); (1977); (1976); (1977); (1976); (1977); (1977); (1976); (1977); (1977); (1976); (1977); (1977); (1976); (1977); (1977); (1977); (1976); (1977); (1977); (1977); (1976); (1977); (1977); (1976); (1977); (1977); (1976); (1977); (19	rock masses				system for tunnel stability	(1971a); Cottiss,
size)size)and back in the second secon	(Strength - Block				and potential failure	Dowel and Franklin
Rock Mass RatingRMRBieniawskiTunnels, cuttings, nodification and correction pranklin 1975; Franklin, 1975; Franklin, 1971; Franklin, Broch and Walton 1971; Franklin, Broch and Walton 1971; Franklin, Louis and Masure 1974; Louis and Masure 1974; 	size)				mechanisms assessment	(1971b); Franklin
Image: Section of the section of th					(Franklin 1986). According	(1970); Franklin
Rock Mass RatingRMRBieniawskiTunnels, cuttings, Mines, Stopes and Poundations DesignA raw rating regulation, modification and correction possible that he 1979 version discussion of the KMR system (1973); Bieniawski (1976); Bieniawski (1978); Bieniawski (1978); Barton et al. ChambersThey are the most commonly used classification systems for underground excavations. Barton (1976); Barton and Grimstad modification 2020; Jeast Barton (1976); Barton and Grimstad modification 2020; Jeast Barton (1978); Barton and Grimstad modification 2020; Least Barton (2020; Barton and Grimstad modification 2020; Least Barton (2020; Barton and Grimstad modification 2020; Least Barton (2020; Barton and Grimstad (2041); Barton, Lien and Lunde (1974); Barton, Lien and Lunde (1974); Barton, Lien and Lunde (1974)					to rock strength and block	(1975); Franklin
Rock Mass RatingRMRBieniawskiTunnels, cuttings, Mines, Slopes and Poundations DesignAnsware 1974; Franklin, 1975; Franklin, 1976; Franklin, 1976; Franklin, 1976; Franklin, 1976; 1980; Last modification of the RNR system (1973); Bieniawski (1976); Bieniawski (1976); Bieniawski (1976); Bieniawski (1976); Bieniawski (1978); Bieniawski (1978); Bieniawski (1988); Hack (1988); Hack (1988); Hack (1988); Hack (1988); Hack (1988); Barton (1986); Barton (1986); Barton (1986); Barton (1986); Barton (1986); Barton (1997); Bieniawski (1988); Hack (1998), Barton (1998); Barton (1998); Barton (1999); Barton (1990); Laubscher					diameter. (Franklin, 1970;	(1976); Franklin
Rock Mass RatingRMRBieniawskiTunnels, cutings, Mines, Slopes and Foundations DesignA raw rating regulation, modification and correction for discontinuity orientation of the RMR system (1974); Bieniawski (1976); Bieniawski (1988); Bieniawski (1988); <td></td> <td></td> <td></td> <td></td> <td>Cottiss, Dowel and Franklin</td> <td>(1986); Franklin,</td>					Cottiss, Dowel and Franklin	(1986); Franklin,
Rock Mass RatingRMRBieniawskiTunnels, cutings, Mines, Slopes and posingA raw rating regulation, 					1971; Franklin, Broch and	Broch and Walton
Rock Mass RatingRMRBieniawskiTunnels, cutturgs, Mines, Slopes and Poundations DesignA raw rating regulation, modification and correction for discontinuity orientative to apply in slopes that was added in the 1979 version of the RMR system (1973, Bieniawski (1974); Bieniawski (1975); Bieniawski (1976); Bieniawski (1976); Bieniawski (1976); Bieniawski (1976); Bieniawski (1976); Bieniawski (1976); Bieniawski (1978); Bieniawski (1978); Barton (1978); Barton (1988); Barton (1994); Barton (1994); Barton and Lunde (1974a); Barton, Lien and Lunde (1974a); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1975).Master and Taylor (1975).RMR system extension-KeaverMinesBased on RMR (1973), (1977, 1981, 1984, 1990), (1977, 1981, 1984, 1990), Laubscher (1977); Laubscher (1977); Laubscher (1977); Laubscher (1977); Laubscher (1977); Laubscher (1977); Laubscher (1977); Laubscher (1978), Coppin; Laubscher (1978), Laubscher (Walton 1971; Franklin,	(1971); Franklin,
Rock Mass RatingRMRBieniawskiTunnels, cuttings, Mines, Stopes and Foundations DesignA raw rating regulation, modification and correction for discontinuity orientation bieniawski (1973); Bieniawski (1973); Bieniawski (1974); Bieniawski (1974); Bieniawski (1974); Bieniawski (1974); Bieniawski (1975); Bieniawski (1976); Bieniawski (1976); Bieniawski (1978); Bieniawski (1988); Batron (1976); Batron (1988); Batron (1976); Batron (1976); Batron (1988); Batron (1998); Batron (1999); Batron and Grimstad (1999); Batron and Grimstad (2004); Batron, Loset, Lien and Lunde (1974a); Batron, Loset, Lien and Lunde (1974b); Batron, Loset, Lien and Lunde (1975).WeaverRippability Mining(1975).Veaver (1975).RMR system extension-VeaverRippability Mining(1976), Based on RMR (1973), Laubscher (1977); Laubscher (1978); Laubscher (1978); Laubscher (1978); Laubscher (1978); Laubscher (1978);					Louis and Masure 1974;	Louis and Masure
Rock Mass RatingRMRBieniawskiTunnels, cutting Tunnels, cutting regulation, DesignAraw rating regulation, modification and correction for discontinuity orientation bieniawski (1973); Bieniawski (1974); Bieniawski (1974); Bieniawski (1976); Bieniawski (1976); Bieniawski (1976); Bieniawski (1976); Bieniawski (1976); Bieniawski (1976); Bieniawski (1976); Bieniawski (1976); Bieniawski (1976); Bieniawski (1978); Bieniawski (1978); Bieniawski (1978); Bieniawski (1978); Bieniawski (1989), Last modification systems for underson (1970); Barton (1976); Barton (1978); Barton (1976); Barton (1978); Barton (1976); Barton (1976); Barton (1978); Barton (1976); Barton (1976); Barton (1976); Barton (1976); Barton (1976); Barton (1978); Barton (1978); Barton (1978); Barton (1978); Barton, Lien and Lunde (1974); Barton, Lien and Lunde (1974); Barton, Lien and Lunde (1974); Barton, Lien and Lunde (1975).Weaver (1975).RMR system extension-LaubscherMinesBased on RMR (1973), (1977, 1981, 1984, 1990), Laubscher (1977a); Laubscher (19					Franklin 1975; Franklin	(1974).
Rock Mass RatingRMRBieniawskiTunnels, cuttings, Mines, Slopes and DesignA raw rating regulation, modification and correction for discontinuity orientation to apply in slopes that was added in the 1979 version of the RMR system (1988); Barton I (1984); Bieniawski (1976); Bieniawski (1978); Bieniawski (1988); Bieniawski (1988); Bieniawski (1988); Bieniawski (1988); Bieniawski (1988); Bieniawski (1988); Bieniawski (1988); Bieniawski (1988); Barton (1976); Barton (1998); Barton (1976); Barton (1976); Barton (1976); Barton (1976); Barton (1976); Barton (1976); Barton (1976); Barton (1976); Barton and Grimstad (1994); Barton, Lien and Lunde (1974a); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1977b); Laubscher (1977b); Laubsc					1976; Franklin, J. A., 1986).	
RatingMines, Stopes and Foundationsmodification adcorrection iscontinuity orientationsPratelidis (2009); Beiniawski (1974); added in the 1979 version of the RMR system (1973, 1974, 1976, 1984, 1988, Bieniawski (1976b); Bieniawski (1976b); Bieniawski (1978b); Bieniawski (1988); Bieniawski (1988); Barton (1976); Barton (1976); Barton (1976); Barton (1976); Barton (1976); Barton (1976); Barton (1976); Barton (1974); Barton (1974); Barton (1974); Barton, Lien and Lunde (1974a); Barton, Lien and Lunde (1974a); Barton, Lien and Lunde (1974a); Barton, Lien and Lunde (1974a); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1974b); Laubscher (1977), Laubscher (1977), Laubscher (1977), Laubscher (1977), Laubscher (1977), Laubscher (1977); Laubscher (1977); Laubscher (1977); Laubscher (1977); Laubscher (1977); Laubscher (1977); Laubscher (1977); Laubscher (1977); Laubscher (1977); Laubscher (1978); Laubscher (1978); Laubscher (1978); Laubscher (1978); Laubscher (1978); Laubscher (1978); Laubscher (1978); Laubscher (1978); Laubscher (1978)	Rock Mass	RMR	Bieniawski	Tunnels, cuttings,	A raw rating regulation,	Adopted from
Poundations Designfor discontinuity operation to apply in slopes that was added in the 1979 version of the RMR system (1973, 1974, 1976, 1984, 1988, 1989), Last modification (1989), Last modification (1989), Last modification (1989), Last modification (1989), Last modification systems for tunnels (1974, 1976, 1984, 1988, Berniawski (1989); Hack (1998); Barton (1976); Barton (1977); Barton, Lien and Lunde (1974a); Barton, Lien and Lunde (1974a); Barton, Lien and Lunde (1974a); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1975).RMR system extension-WeaverRippability(1975).Weaver (1975).RMR system extension-LaubscherHard Rock Mining Mines(1976).Laubscher and Taylor (1976).RMR system extension-LaubscherMinesBased on RMR (1973), Laubscher (1977); Laubscher (1977); Laubscher (1977); Laubscher (1977); Laubscher (1977); Laubscher (1977); Laubscher (1978); Laubscher (1978); Laubscher (1978); Laubscher (1978); Laubscher (1978); Laubscher (1978); Laubscher (1978); Laubscher (1978); Laubscher (1978); Laubscher (Rating			Mines, Slopes and	modification and correction	Pantelidis (2009);
DesignLo spip in stops added in the 1979 version of the RMR system (1973); Bieniawski (1974); Bieniawski (1976a); Bieniawski (1976a); Bieniawski (1976a); Bieniawski (1976a); Bieniawski (1976a); Bieniawski (1976a); Bieniawski (1976a); Bieniawski (1988); Bieniawski (1976a); Bieniawski (1976b); Bieniawski (1976b); Bieniawski (1988); Bieniawski (1988); Barton (1976); Barton (1976); Barton (1976); Barton (1976); Barton (1976); Barton (1976); Barton (1978); Barton (1988); Barton (1978); Barton (1988); Barton, Lien and Lunde (1974a); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1974b); Laubscher and Page (1976).Partelidis (2009); Laubscher (1977b); Laubscher (1977b); Laubscher (1977b); Laubscher (1977b); Laubscher (1977b); Laubscher (1977b); Laubscher (1977b); Laubscher (1977b); Laubscher (1977b); Laubscher (1977b				Foundations	for discontinuity orientation	Bieniawski (1973) ;
Added in the RMR system Quality)Definition of the RMR system Bernitowski (1976a); Bienitowski (1976b); Bienitowski (1976b); Bienitowski (1976b); Bienitowski (1976b); Bienitowski (1976b); Bienitowski (1976b); Bienitowski (1976b); Bienitowski (1978b); Bienitowski (1978b); Bienitowski (1978b); Bienitowski (1978b); Bienitowski (1989); Bienitowski (1989); Bienitowski (1980); Bienitowski (1980); Bienitowski (1980); Bienitowski (1980); Barton (1988); Barton (1989); Barton (1999); Barton (1990); Barton Addition (1999); Barton (1990); Barton Addition (1980); Hack (1998).RMR system extension-LaubscherHard Rock Mining(1976).Kalescher (1977b); Laubscher (1978), Laubscher (1998); Ther typological classification-Matula and HolzerDescriptive, (1978). Hack (1998); Hack (1998); Hack (1998);				Design	added in the 1979 version	Bieniawski (1974); Bieniawski (1984);
Rock Tunneling Quality Index (Rock Mass Quality)Q 					of the RMR system (1973	Bienjawski (1976a):
Rock Tunneling Quality Index (Rock Mass Quality)QBarton et al.Tunnels, mines, foundations, ChambersThey are the most commonly used classification systems for tunnels (1974, 1976, 1980), Last modification systems for tunnels (1974, 1976, 1980), Barton al (2002); Barton al (2002); Barton and Cimstad (1994); Barton al (2004); Barton and Cimstad (1994); Barton and Lunde (1974a); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1980); Hack (1998),RMR system extension-LaubscherHard Rock Mining(1975).Weaver (1975).RMR system extension-LaubscherMiningBased on RMR (1973), (1977, 1981, 1984, 1990).Pantelidis (2009); Laubscher (1977b); Laubscher (1977b); Laubscher (1977b); Laubscher (1984); Laubscher (1986); Hack (1988);The typological classification-Matula and HolzerDescriptive, (1978).Matula and Holzer (1978).Terrain Index Cisope Stability-VecchiaStability of hillikides andA simple terrain index for the stability assessement of Vecchia (1978).					1974 1976 1984 1988	Bierntowski (1976h)
Rock Tunneling Quality)QBarton et al.Tunnels, mines, foundations, ChambersThey are the most classification systems for unnels (1974, 1976a, 1980), Barton (1976); Barton (1976); Barton (1976); Barton (1976); Barton (1976); Barton (1978); Barton (1978); Barton (1978); Barton (1974); Barton, Lien and Lunde (1974a); Barton, Lien and Lunde (1974a); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1976).Pantelidis (2009); Course, Lien and Lunde (1980); Hack (1988).RMR system extension-LaubscherMinesBased on RMR (1973), (1977, 1981, 1984, 1990).Pantelidis (2009); Laubscher (1977b); Laubscher (1978).Matula and Holzer (1978).The typological classification-Matula and HolzerDescriptive, (1978).Matula And Holzer (1978).<					1989). Last modification	Bieniawski (1979):
Rock Tunneling Quality Index (Rock Mass Quality)Q Barton et al.Tunnels, mines, foundations, ChambersThey are the most commonly used classification systems for unnels (1974, 1976a, 1980, 1988, 1999, 2002). Last modification 2002. Use in design of support in underground excavations.Barton (1976); Barton (1988); Barton and Grimstad (1994); Barton and Grimstad (2004); Barton, Lien and Lunde (1974a); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1980); Hack (1988).RMR system extension-WeaverRippability Mining(1975).Weaver (1975).RMR system extension-LaubscherHard Rock Mining(1976).Laubscher and Taylor (1976).Mining Rock Mass RatingMRMR LaubscherLaubscherMinesBased on RMR (1973), (1977, 1981, 1984, 1990).Pantelidis (2009); Laubscher (1977a); Laubscher (1977a); Laubscher (1977b); Laubscher (1977b); Laubscher (1977b); Laubscher (1977b); Laubscher (1977a); Laubscher (1977b); Laubscher (1977b); Laubscher (1977a); Laubscher (1977b); Laubscher (1978).The typological classification-Matula and HolzerDescriptive, (1978).Matula and Holzer (1978).The typological classification-VecchiaStability of hillisides andA simple terrain index for the stability assessment of hillisides andMatula and Holzer (1978).					(1989).	Bieniawski (1988):
Rock Tunneling Quality Index (Rock Mass Quality)QBarton et al.Tunnels, mines, foundations, ChambersThey are the most commonly used classification systems for tunnels (1974, 1976a, 1980, 1999), 2002). Last modification 2002. Use in (1999); Barton and Grimstad (1994); Barton and Grimstad (2004); Barton, Lien and Lunde (1974a); Barton, Lien and Lunde (1974a); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1976b).Laubscher Taylor (1975).RMR system extension-LaubscherHard Rock Mining(1976).Laubscher and Taylor (1976).Mining Rock Mass RatingMRMRLaubscherMinesBased on RMR (1973), (1977, 1981, 1984, 1990).Pantelidis (2009); Laubscher (1977b); Laubscher (1977b); Laubscher (1977b); Laubscher (1978), (2009); Hatek (1998).The typological classification-Matula and HolzerDescriptive, (1978).Matula and Holzer (1978), (2009); Hatek (1998); (2009); Hate						Bieniawski (1989);
Rock Tunneling Quality Index (Rock Mass Quality)QBarton et al.Tunnels, mines, foundations, ChambersThey are the most commonly used classification systems for tunnels (1974, 1976a, 1980, 1988, 1999, 2002). Last modification 2002): Last modification 2002. Use in design of support in underground excavations.Pantelidis (2009); Barton (1976); Barton and Grimstad (1994); Barton alton (1994); Barton diffication 2002. Use in design of support in underground excavations.Pantelidis (2009); Barton (1988); Barton and Grimstad (1994); Barton alton (1974a); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1974b); Barton, Loset, Lien and Lunde (1974b); Barton, Loset, Lien and Lunde (1975).WeaverRippability(1975).Weaver (1975).RMR system extension-LaubscherHard Rock Mining(1976).Laubscher and Taylor (1976).Laubscher (1977b); Laubscher (1990); Hack (1998).The typological classification-Matula and HolzerGeneral BartonDescriptive, (1978).Matula and Holzer (1978).The typological classification-VecchiaStability of hillsides andA simple terrain index for the stability assessment of Vecchia (1978).						Hack (1998).
Quality Index (Rock Mass Quality)Antiper sector (Rock Mass Quality)Barton (1976); (Lassification systems for tunnels (1974, 1976a, 1980, 1988, 1999, 2002). Last modification 2002. Use in design of support in underground excavations.Barton (1988); Barton (1994); Barton and Grimstad (1994); Barton (1999); Barton and Grimstad (2004); Barton, Lien and Lunde (1974a); Barton, Lien and Lunde (1974a); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1976).Weaver (1975).RMR system extension-LaubscherHard Rock Mining(1976).Weaver (1975).RMR system extension-LaubscherMinesBased on RMR (1973), (1977, 1981, 1984, 1990).Pantelidis (2009); Laubscher (1977a); Laubscher (1977a); Laubscher (1977b); Laubscher (1978).Matula and Holzer (1978).The typological classification-Matula and HolzerGeneral Batability of hillsides andDescriptive, (1978).Matula and Holzer (1978).The typological classification-VecchiaStability of hillsides andA simple terrain index for the stability as	Rock Tunneling	Q	Barton et al.	Tunnels, mines,	They are the most	Pantelidis (2009);
(Rock Mass Quality)Image: ChambersClassification systems for tunnels (1974, 1976a, 1980, 1988, 1999, 2002). Last modification 2002. Use in design of support in underground excavations.Barton (1988); Barton (1998); Barton and Grimstad (1994); Barton (1999); Barton and Grimstad (2004); Barton, Lien and Lunde (1974a); Barton, Lien and Lunde (1974b); Barton, Loest, Lien and Lunde (1980); Hack (1998).RMR system extension-KeaverRippability(1975).Weaver (1975).RMR system extension-LaubscherHard Rock Mining(1976).Laubscher and Taylor (1976).Mining Rock MiningMRMRLaubscherMinesBased on RMR (1973), (1977, 1981, 1984, 1990).Pantelidis (2009); Laubscher (1977b); Laubscher (1977b); Laubscher (1990); Laubscher (1998); C(2009); Hack (1998);The typological classification-Matula and HolzerDescriptive, (1978).Matula	Quality Index	-		foundations,	commonly used	Barton (1976);
Quality)LungerLungerLunnels (1974, 1976a, 1980, 1988, 1999, 2002). Last modification 2002. Use in design of support in underground excavations.Barton (2002); Barton and Grimstad (1994); Barton and Grimstad (1994); Barton and Grimstad (2004); Barton, Lien and Lunde (1974a); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1970b); Mining Rock Mass Rating-WeaverRippability(1975).Weaver (1975).Mining Rock Mass RatingMRMR HaubscherLaubscherMinesBased on RMR (1973), (1977, 1981, 1984, 1990).Laubscher (1977h); Laubscher (1977h); Laubscher (1977b); Laubscher (1977b); Laubscher (1984); Laubscher (1987b);Matula and Holzer (1987); (1977, 1981, 1984, 1990).The typological class classification-Matula and HolzerGeneral HolzerDescriptive, (1978).Matula and Holzer (1978).Terrain Index (Slop Stability divides-VecchiaStability of hillsides andA simple terrain index for the stability assessment of Vecchia (1978).Hack (1998); Vecchia (1978).	(Rock Mass			Chambers	classification systems for	Barton (1988);
Image: stabilityImage: stabilityImage	Quality)				tunnels (1974, 1976a, 1980,	Barton (2002);
modification 2002. Use in design of support in underground excavations.(1994); Barton and (1999); Barton and Grimstad (2004); Barton, Lien and Lunde (1974a); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1980); Hack (1998).RMR system extension-WeaverRippability(1975).Weaver (1975).RMR system extension-LaubscherHard Rock Mining(1976).Laubscher and Taylor (1976).Mining Rock Mass RatingMRMRLaubscherMinesBased on RMR (1973), (1977, 1981, 1984, 1990).Pantelidis (2009); Laubscher (1977b); Laubscher (1977b); Laubscher (1997b); Laubscher (1997b); Laubscher (1997b); Laubscher (1997b); Laubscher (1990); Laubscher (1997b); Laubscher (1990); Laubscher (1998); (1978).The typological classification-Matula and HolzerDescriptive, (1978).Matula and Holzer (1978).Terrain Index (Slop Stability-VecchiaStability of hillsides andA simple terrain index for Hack (1998); Vecchia (1978).					1988, 1999, 2002). Last	Barton and Grimstad
Image: Section of the section of th					modification 2002. Use in	(1994); Barton
Image: Second statistic stateImage: Second stateImage: Second stateGrimstad (2004); Barton, Lien and Lunde (1974a); Barton, Lien and Lunde (1974b); Barton, Loset, Lien and Lunde (1980); Hack (1988).RMR system extension-WeaverRippability(1975).Weaver (1975).RMR system extension-LaubscherHard Rock Mining(1976).Laubscher and Taylor (1976).Mining Rock Mining Rock Miss RatingMRMRLaubscherMinesBased on RMR (1973), (1977, 1981, 1984, 1990).Pantelidis (2009); Laubscher (1977a); Laubscher (1977b); Laubscher (1977b); Laubscher (1997b); Laubscher (1990); Pantelidis (2009); Hack (1998).The typological classification-Matula and HolzerGeneralDescriptive, (1978).Matula and Holzer (1978).Terrain Index (Slope Stability-VecchiaStability of hillsides andA simple terrain index for the stability assessment of Vecchia (1978).Hack (1978).					design of support in	(1999); Barton and
Additional and barton, Lien and Lunde (1974a); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1974b); Barton, Lien and Lunde (1980); Hack (1998).RMR system extension-WeaverRippability(1975).Weaver (1975).RMR system extension-LaubscherHard Rock Mining(1976).Laubscher and Taylor (1976).Mining Rock Mass RatingMRMRLaubscherMinesBased on RMR (1973), (1977, 1981, 1984, 1990).Pantelidis (2009); Laubscher (1977a); Laubscher (1977b); Laubscher (1976).The typological classification-Matula and HolzerGeneralDescriptive, (1978).Matula and Holzer (1978).The typological classification-Matula and HolzerGeneralDescriptive, (1978).Matula and Holzer (1978).Terrain Index (Slope Stability-VecchiaStability of hillsides andA simple terrain index for the stability assessment of Vecchia (1978).					underground excavations.	Grimstad (2004);
RMR system extension-WeaverRippability(1975).Lunde (1974a); Barton, Lien and Lunde (1974b); Barton, Loset, Lien and Lunde (1980); Hack (1998).RMR system extension-LaubscherRippability(1975).Weaver (1975).RMR system extension-LaubscherHard Rock Mining(1976).Laubscher and Taylor (1976).Mining Rock Mass RatingMRMRLaubscherMinesBased on RMR (1973), (1977, 1981, 1984, 1990).Pantelidis (2009); Laubscher (1977a); Laubscher (1977b); Laubscher (1977b); Laubscher (1990); Laubscher (1990); Laubscher (1990); Laubscher (1990); Laubscher (1990); Laubscher (1990); Laubscher (1990); Laubscher (1990); Laubscher (1984); Laubscher (1990); Laubscher (1993).The typological classification-Matula and HolzerGeneralDescriptive, (1978).Matula and Holzer (1978).Terrain Index (Slope Stability-VecchiaStability of hillsides andA simple terrain index for the stability assessment of Vecchia (1978). <td></td> <td></td> <td></td> <td></td> <td></td> <td>Barton, Lien and</td>						Barton, Lien and
Barton, Len and Lunde (1974b); Barton, Loset, Lien and Lunde (1980); Hack (1980); Laubscher and Taylor (1976).RMR system extension-LaubscherHard Rock Mining(1976).Laubscher and Taylor (1976).Mining Rock Mass RatingMRMRLaubscherMinesBased on RMR (1973), (1977, 1981, 1984, 1990).Laubscher (1977a); Laubscher (1977b); Laubscher (1977b); Laubscher (1977b); Laubscher (1970); Laubscher (1990); Laubscher and Page (1990); Pantelidis (2009); Hack (1998).The typological classification-Matula and HolzerGeneralDescriptive, (1978).Matula and Holzer (1978).Terrain Index (Slope Stability-VecchiaStability of hillsides andA simple terrain index for the stability assessment of Vecchia (1978).						Lunde (19/4a);
RMR system extension-WeaverRippability(1975).Barton, Loset, Lien and Lunde (1980); Hack (1998).RMR system extension-LaubscherHard Rock Mining(1975).Weaver (1975).RMR system extension-LaubscherHard Rock Mining(1976).Laubscher and Taylor (1976).Mining Rock Mass RatingMRMRLaubscherMinesBased on RMR (1973), (1977, 1981, 1984, 1990).Pantelidis (2009); Laubscher (1977a); Laubscher (1977b); Laubscher (1977b); Laubscher (1984); Laubscher (1990); Pantelidis (2009); Pantelidis (2009); Pantelidis (2009); Hack (1998).The typological classification-Matula and HolzerGeneralDescriptive, (1978).Matula and Holzer (1978).Terrain Index (Slope Stability-VecchiaStability of hillsides andA simple terrain index for the stability assessment of Vecchia (1978).Hack (1998);						Lundo (1074b):
RMR system extension-WeaverRippability(1975). and Lunde (1980); Hack (1998).RMR system extension-LaubscherHard Rock Mining(1976).Laubscher and Taylor (1976).Mining Rock Mining RockMRMRLaubscherMinesBased on RMR (1973), (1977, 1981, 1984, 1990).Pantelidis (2009); Laubscher (1977a); Laubscher (1977b); Laubscher (1977b); Laubscher (1990); Laubscher (1998).The typological classification-Matula and HolzerDescriptive, (1978).Matula and Holzer (1978).Terrain Index (Slope Stability-VecchiaStability of hillsides and the stability assessment of the stability assessment ofHack (1998).						Barton Loset Lian
RMR system extension-WeaverRippability(1975).Hack (1998). Hack (1995).RMR system extension-LaubscherHard Rock Mining(1976).Laubscher and Taylor (1976).Mining Rock Mass RatingMRMRLaubscherMinesBased on RMR (1973), (1977, 1981, 1984, 1990).Pantelidis (2009); Laubscher (1977a); Laubscher (1977b); Laubscher (1977b); Laubscher (1984); Laubscher (1990); Laubscher (1990); Laubscher (1984); Laubscher (1990); Laubscher (1990); Laubscher (1990); Laubscher (1990); Laubscher (1984); Laubscher (1998).The typological classification-Matula and HolzerGeneral HolzerDescriptive, (1978). (1978).Matula and Holzer (1978).Terrain Index (Slope Stability-VecchiaStability of hillsides andA simple terrain index for the stability assessment ofHack (1998).						and Lunde (1980).
RMR system extension-WeaverRippability(1975).Hack (1976).RMR system extension-LaubscherHard Rock Mining(1976).Laubscher and Taylor (1976).Mining Rock Mass RatingMRMRLaubscherMinesBased on RMR (1973), (1977, 1981, 1984, 1990).Pantelidis (2009); Laubscher (1977a); Laubscher (1977b); Laubscher (1977b); Laubscher (1977b); Laubscher (1990); Laubscher (1990); Laubscher (1990); Laubscher (1990); Laubscher (1990); Laubscher (1990); Laubscher (1990); Laubscher (1998).The typological classification-Matula and HolzerGeneralDescriptive, (1978). (1978).Matula and Holzer (1978).Terrain Index (Slope Stability-VecchiaStability of hillsides andA simple terrain index for the stability assessment of Vecchia (1978).Hack (1998).						Hack (1998)
extension-LaubscherHard Rock Mining(1976).Laubscher and Taylor (1976).Mining Rock Mass RatingMRMRLaubscherMinesBased on RMR (1973), (1977, 1981, 1984, 1990).Pantelidis (2009); Laubscher (1977a); Laubscher (1977b); Laubscher (1977b); Laubscher (1984); Laubscher (1990); Laubscher and Page (1990); Pantelidis (2009); Hack (1998).The typological classification-Matula and HolzerGeneralDescriptive, (1978).Matula and Holzer (1978).Terrain Index (Slope Stability-VecchiaStability of hillsides andA simple terrain index for the stability assessment ofHack (1998).	RMR system	-	Weaver	Rippability	(1975).	Weaver (1975).
RMR system extension-LaubscherHard Rock Mining(1976).Laubscher and Taylor (1976).Mining Rock Mass RatingMRMRLaubscherMinesBased on RMR (1973), (1977, 1981, 1984, 1990).Pantelidis (2009); Laubscher (1977a); Laubscher (1977b); Laubscher (1977b); Laubscher (1984); Laubscher (1990); Laubscher and Page (1990); Pantelidis (2009); Hack (1998).The typological classification-Matula and HolzerGeneralDescriptive, (1978).Matula and Holzer (1978).Terrain Index (Slope Stability-VecchiaStability of hillsides andA simple terrain index for the stability assessment ofHack (1998); Vecchia (1978).	extension					
extensionMiningTaylor (1976).Mining Rock Mass RatingMRMRLaubscherMinesBased on RMR (1973), (1977, 1981, 1984, 1990).Pantelidis (2009); Laubscher (1977a); Laubscher (1977b); Laubscher (1977b); Laubscher (1990); Laubscher (1990); Laubscher (1990); Laubscher and Page (1990); Pantelidis (2009); Hack (1998).The typological classification-Matula and HolzerGeneralDescriptive, (1978).Matula and Holzer (1978).Terrain Index (Slope Stability-VecchiaStability of hillsides andA simple terrain index for the stability assessment ofHack (1998); Vecchia (1978).	RMR system	-	Laubscher	Hard Rock	(1976).	Laubscher and
Mining Rock Mass RatingMRMRLaubscherMinesBased on RMR (1973), (1977, 1981, 1984, 1990).Pantelidis (2009); Laubscher (1977a); Laubscher (1977b); Laubscher (1970); Laubscher (1990); Laubscher (1990); Laubscher (1990); Laubscher (1990); Laubscher (1990); Laubscher (1990); Laubscher (1990); Laubscher (1990); Laubscher (1990); Hack (1998).The typological classification-Matula and HolzerGeneralDescriptive, (1978).Matula and Holzer (1978).Terrain Index (Slope Stability-VecchiaStability of hillsides andA simple terrain index for the stability assessment ofHack (1998); Vecchia (1978).	extension			Mining		Taylor (1976).
Mass RatingImage: Construction of the stability of the stability assessment of the stabil	Mining Rock	MRMR	Laubscher	Mines	Based on RMR (1973),	Pantelidis (2009);
Image: space of the stabilityImage: space of the stability assessment of the stability as	Mass Rating				(1977, 1981, 1984, 1990).	Laubscher (1977a);
Laubscher, (1984); Laubscher (1990); Laubscher and Page (1990); Pantelidis (2009); Hack (1998).The typological classification-Matula and HolzerGeneralDescriptive, (1978).Matula and Holzer (1978).Terrain Index (Slope Stability-VecchiaStability of hillsides andA simple terrain index for the stability assessment ofHack (1998).						Laubscher (1977b);
Laubscher (1990); Laubscher and Page (1990); Pantelidis (2009); Hack (1998).The typological classification-Matula and HolzerGeneralDescriptive, (1978).Matula and Holzer (1978).Terrain Index (Slope Stability-VecchiaStability of hillsides andA simple terrain index for the stability assessment ofHack (1998); Vecchia (1978).						Laubscher, (1984);
Laubscher and Page (1990); Pantelidis (2009); Hack (1998).The typological classification-Matula and HolzerGeneralDescriptive, (1978).Matula and Holzer (1978).Terrain Index (Slope Stability-VecchiaStability of hillsides andA simple terrain index for the stability assessment ofHack (1998); Vecchia (1978).						Laubscher (1990);
The typological classificationMatula and HolzerGeneralDescriptive, (1978).Matula and Holzer (1978).Terrain Index (Slope Stability-VecchiaStability of hillsides andA simple terrain index for the stability assessment ofHack (1998); Vecchia (1978).						(1000): Deptalidia
The typological classification-Matula and HolzerGeneralDescriptive, (1978).Matula and Holzer (1978).Terrain Index (Slope Stability-VecchiaStability of hillsides andA simple terrain index for the stability assessment ofHack (1998); Vecchia (1978).						(1990), Failtenuis (2009): Hack (1008)
InterpropriationInternationOctoberDescriptive, (1978).International and HolzerclassificationHolzer(1978).(1978).Terrain Index-VecchiaStability of hillsides andA simple terrain index for the stability assessment ofHack (1998); Vecchia (1978).	The typological	_	Matula and	General	Descriptive (1978)	Matula and Holzer
Terrain Index-VecchiaStability of hillsides andA simple terrain index for the stability assessment ofHack (1998); Vecchia (1978).	classification	-	Holzer	Guiuai	Descriptive, (1770).	(1978)
(Slope Stability hillsides and the stability assessment of Vecchia (1978).	Terrain Index	-	Vecchia	Stability of	A simple terrain index for	Hack (1998):
	(Slope Stability			hillsides and	the stability assessment of	Vecchia (1978).

Mohammad Salehi Alashti.; East African Scho	lars Multidiscip Bull; V	vol-1, Iss-3 (Dec,	2018): 61-121
---	--------------------------	--------------------	---------------

System of			scarps - Natural	hillsides and scarps. A	
Vecchia			slopes	classification system	
			1	designed to quantify the	
				stability of a hillside and	
				scarp e a patural slopes	
				(1078)	
DIAD			XX7 (1 1 1)	(1978).	01: : (1070)
RMR system	-	Olivier	Weatherability	(1979).	Olivier (1979).
extension					
Rock Mass	RMS	Selby, Moon and	Cuttings	In accordance to natural	Pantelidis (2009);
Strength		Selby		slope database (1980,	Hack (1998); Moon
(Geomorphic				1982), (1990).	and Selby (1990);
Rock Mass					Selby (1980a): Selby
Strength					(1980b): Selby
Classification)					(1982a); Selby
Clussification)					(1982h); Selby
					(19820), Beloy
Unified Deals	LIDCC	W/:11:	Concert	The immediate the second and	(1907). Williamaan (1090).
Unified Rock	URCS	Williamson	General	The input to the system	Williamson (1980);
Mass				mainly is based on	Williamson (1984);
Classification				descriptions. Main	Williamson and
System (Unified				applications in	Kuhn (1988).
classification)				communication, (1980,	
				1984, 1988).	
Basic	BGD	Brown	General	Use for general	Brown (1981).
geotechnical				applications, (1981).	
classification					
RMR system	_	Ghose and Rain	Coal mining	(1981)	Ghose and Rain
extension	-	Onose and Raju	Coar mining	(1901).	(1981)
DMD system		Morono Tollon	Tunnalina	(1082)	(1701). Morene and Talian
KIVIK System	-	Moreno Tanon	Tunnening	(1982).	
extension	DIG	0.11		(1000)	(1982).
Rock Mass	RMS	Stille et al.	Metal mining	(1982).	Stille, Groth and
Strength					Fredriksson (1982).
Q-system	Q	Kirsten	Excavatability,	Applications as	Kirsten (1982).
development				Excavatability (1982).	
Q-system	Q	Kirsten	Tunneling	Applications as Tunneling	Kirsten (1983)
development				(1983).	
RMR system	-	Kendorski and	Hard Rock	(1983).	Kendorski.
extension		Cummings	Mining		Cummings
entenoron		C unining 5			Bieniawski and
					Skinner (1983)
DMD system		Nelveo et el	Tunnaling	(1082)	Nekeo Jihoshi and
	-	INAKAO et al.	Tunnening	(1983).	V_{axa0} , (1082)
extension			D	(1000)	Koyama (1985).
RMR system	-	Serafim and	Foundations	(1983).	Serafim and Pereira
extension		Pereira			(1983).
RMR system	-	Gonzalez de	Tunneling	(1983, 1985).	Gonzalez de Vallejo
extension		Vallego			(1983); González de
					Vallejo (1985).
RMR system	-	Ünal	Coal mine roof	(1983).	Unal (1983).
extension			bolting		
Slope Mass	SMR	Romana, Romana	Cuttings	Based on RMR (1979). The	Adopted from
Rating	Sint	et al	Cuttings	utmost widespread	Pantelidis (2009)
Ruting		et ul.		classification system that	Pomana (1085) :
				used for slopes (1095–1001	$\mathbf{D}_{\text{omans}} (1001).$
				1002 1005 2002)	$\frac{1991}{5}$
				1995, 1995, 2005).	Komana (1993);
					котапа (1995а);
					Romana (1995b);
					Romana, Seron and
					Montalar (2003);
					Hack (1998).
RMR system	-	Newman	Coal mining	(1985, 1986).	Newman (1985);
extension					Newman (1985);

					Newman and Bieniawski (1986).
RMR system	-	Sandbak	Boreability	(1985).	Sandbak (1985).
RMR system extension	-	Smith	Dredgeability	(1986).	Smith (1986).
RMR system extension	-	Venkateswarlu	Coal mining	(1986).	Venkateswarlu (1986).
Slope Rock Mass Rating	SRMR	Robertson	Cuttings, Slope Stability	Based on RMR. The classification provided for weak altered rock mass materials from drill-hole cores (1988).	Pantelidis (2009); Robertson (1988); Hack (1998).
Communication Weakening Coefficient System	WCS	Singh	Coal mining	(1986, 1987, 1988, 1989).	Brown, Denby and Singh (1988); Brown and Singh (1987); Singh, Brown, Denby and Croghan (1986); Singh, Denby and Brown (1985); Singh and Gahrooee (1989); Singh, Reed and Hughes (1987).
Rockfall Hazard Rating System	RHRS	Developed by Pierson, Davis, and Van Vickle	Rock fall hazard assessment of slopes	One of the most widely recognized, accepted and used methods for assessment of slopes' rockfall hazards such as sides of highways (1990).	Adopted from Pierson, Davis and Van Vickle (1990); Hack (2002); Federal Highways Administration (1993).
Haines System (Slope Stability System of Haines - Modified Laubscher)	-	Haines & Terbrugge	Rock slopes stability in open pits	A rock mass classification system for introductory estimation of the stability of rock slopes (1991).	Adopted from Hack (1998); Hains and Terbrugge (1991).
Rock Engineering Systems	RES	Hudson	Natural slopes instability assessment	Rock mass characterization applied to assess natural slopes instability assessment (1992).	Hack (1998); Hudson (1992).
Coal Mine Roof Rating	CMRR	Molinda and Mark	Coal mining	Based on RMR format (1994).	Molinda and Mark (1994).
Natural Slope Methodology	NSM	Shuk	Natural slopes stability	Statistical analysis of existing natural slopes to predict rock mass and soil parameters, and the probability of slopes stability (1994a, 1994b, 1994c, 1994d).	Hack (1998); Shuk (1994).
Chinese Slope Mass Rating	CSMR	Chen	Cuttings	Adjustment parameters have been used to the SMR system for the discontinuity condition and slope height (1995).	Adopted from Chen (1995); Pantelidis (2009).
The Rock Mass index system	RMi	Palmström	General, Design of support, TBM progress, Rock engineering, Characterization	The numerical input parameters are ratings according to their character. Application in general characterization, design of support, TBM progress,	Palmstrom (1995).

Mohammad Salehi Alashti.; East African Scholars Multidiscip Bull; Vol-1, Iss-3 (Dec, 2018): 61-121

				(1995).	
Modified Rock	M-RMR	UnaI	Mines, Weak	For weak, stratified,	Pantelidis (2009);
Mass Rating			rock, coal	anisotropic and clay bearing	Ünal (1996).
				rock masses, (1996).	
-	-	Mazzoccola and	Natural slopes	An example for determining	Adopted from
		Hudson		natural slopes instability	Mazzoccola and
				following the rock	Hudson (1996);
				engineering system	Pantelidis (2009);
				methodology; A rock mass	Hack (1998).
				characterization method to	
				show instabilities of natural	
				slopes (1996).	
Geological	GSI	Hoek, Hoek,	General	Based on RMR (1976).	Hoek (1994); Hoek
Strength Index		Kaiser and		Data for correlation with	and Brown (1997);
		Bawden, Hoek		modulus of deformation	Hoek, Kaiser and
		and Brown		(1997). Hoek 1994, Hoek,	Bawden (1995);
				Kaiser and Bawden 1995,	Pantendis (2009).
				Hoek and Brown 1997.	
Rock slope	RDA	Nicholson and	Cuttings	For shallow, weathering-	Nicholson (2000);
Deterioration		Hencher,		related breakdown of	Nicholson (2002);
Assessment		Nicholson et al.,		excavated rock slopes.	Nicholson (2003);
		Nicholson		(1997, 2000, 2002, 2003,	Nicholson (2003);
				2004).	Nicholson (2004);
					Nicholson and
					Hencher (1997);
					Nicholson, Lumsden
					and Hencher (2000);
					Pantelidis (2009).
Geological	GSI	Hoek et al.,	General	Use for non-structurally	Hoek, Marinos and
Strength Index		Marinos and		controlled failures. For all	Benissi (1998);
		Hoek, Marinos et		underground excavations,	Hoek, Read and
		al.		1998, 2000, 2001, 2005.	Karzulovic (2000);
				Application as design of	Marinos and Hoek
				support in underground	(2000); Marinos and
				excavations. Rock mass	Hoek (2001);
				characterization. Hoek,	Marinos V., Marinos
				Marinos and Benissi, 1998,	P. and Hoek (2005);
				Marinos and Hoek, 2000,	Pantelidis (2009).
				2001. Marinos V., Marinos	
				P. and Hoek 2005.	
Slope Stability	SSPC	Hack, Hack et al.	Cuttings	Probabilistic evaluation of	Adopted from
Probability				independently various	Pantelidis (2009);
Classification				mechanics of failure, (1998,	Hack (1998); Hack
				2003).	and Price (1993);
					Hack, Price and
	DO	.		(1000)	Rengers (2003).
Index of rock	ВQ	Lin	-	(1998)	Lin (1998).
mass basic					
quality		Dantan	TDM town alter a	(1000, 2000)	Dantan (2002).
Q-system	Q	Barton	I BM tunneling	(1999, 2000)	Barton (2002) ;
development					Barton (1999);
					(2004)
Volonia Deals	VDECD	Sinch and	Cuttings	For volcenia reals alongs to	(2004).
Volcanic Rock	VKFSK	Singn and	Cuttings	For voicanic rock slopes to	Adopted from Deptedidie (2000)
Rating		Connony	excavations	excavations on construction	Singh and Connelly
i i i i i i i i i i i i i i i i i i i			encurations	sites (2003)	(2003)
Falling Dook	FDU ^{1a}	Singh	Cuttings	Enlarged for stable	(2005). Adopted from
Hazard Index	I'KIII	Singli	Cuttings [temporary	excavations to distinguish	Pantelidia (2000)
Hazalu Index			excavations	the dangers degree for	Singh (2009) ;
1	1	1	sites anonol	the dangers degree 101	5mgn (2004).

Mohammad Salehi Alashti.; East African Scholars Multidiscip Bull; Vol-1, Iss-3 (Dec, 2018): 61-121

				workers (2004).	
Deutsche Steinkohle	-	Witthaus	Coal mining	(2006).	Witthaus (2006).
Rock Mass Excavability	RME	Bieniawski et al.	TBM tunneling	(2006).	Bieniawski, Celada, Galera Fernández and Hernández Álvarez (2006).
RMR system extension	-	Pakalnis et al.	Weak rock mining	(2007).	Pakalnis, Brady, Hughes, Caceres, Ouchi and MacLaughlin (2007).
Slope Stability Rating classification system	SSR	Taheri and Tani	Rock slopes sites	A rock mass classification system for anticipatory slope stability assessment (2010).	Adopted from Rai; Rai, Taheri and Tani (2010); Taheri and Tani (2007).
Rock Slope Rating	RSR	-	Rock slopes stability	A system that evaluates the probability of failures for plane, wedge sliding, toppling and circular failures.	Rai.
Mine Dump Slope Classification (Dump Mass Rating or Dump Slope Rating)	DSR	Rahul et al.	Coal mine waste dump slope	A dump classification system for coal mine waste dump slope (2010, 2011, 2017).	Rai; Rahul (2011); Sharma and Rai (2017).

^a Based on Singh, prior to the use of FRHI, it is essential to proceed for rock slope stability assessment using the previously suggested VRFSR system.

3. Limitations of Current Slope Stability Design Approaches

Empirical methods (models and formulas) of analysis such as rock mass classification; conventional methods of analysis such as basic analysis (Lorimer's method), stereo-graphic kinetic analysis and kinematic analysis tools, limit equilibrium (Swedish Circle ($\phi = 0$) method; logarithmic spiral method; friction circle method; method of slices: such as ordinary method of slices, simplified bishop method, Janbu's simplified method, Janbu's generalized procedure of slices (GPS), Spencer's method, Morgenstern and Price's method, Sarma's method), limit analysis, rockfall simulators; numerical methods of analysis such as continuum modeling; the dis-continuum modeling; hybrid/coupled modeling; statistical & probabilistic methods; Observational and analytical techniques and tools; new methods such as intelligent models, fuzzy logics, artificial neural network, genetic algorithm, slope stability radar and laser systems such as the 3-D laser mapping; the 3-D terrestrial laser scanning & ground penetrating radar, etc. are various slope stability analysis design approaches and modelling methods of slope behaviors.

It is very consequential which designers perceive and comprehend the assumptions and limitations before applying a design method and approach. The relative deserves and deficiencies of the currently existing design methods are summarized as follows [Adopted from 166]:

(1) Limit equilibrium methods (LEM) include the drawbacks, which are the assumptions of (i) The soil or rock masses behaving and acting as a rigid material and (ii) the shear strength being mobilized at the same time along the entire failure surface.

(2) The true failure load can bound by using both of the upper and lower bound limit analysis. However, in this condition the displacement of the slope cannot predict.

(3) Numerical modeling in rock mechanics analyses via standard software does not contemplate fracture propagation via intact material. Also, new developments in this field have not yet reached full maturity for practical applications in slope design. Sjöberg (1999) found that it was not possible to simulate smaller block sizes, as the models were very (computer) memory consuming and took a long time to run. His study also indicated that a reduced block or element size alone might not be sufficient to increase the ability of the rock mass to fracture [263].

(4) Empirical design charts only provide general design advice and process too. Because of limited data, the establishment of more detailed design rules is not possible.

(5) Although physical model tests can be useful for determining fundamental failure mechanisms and for the verification of analytical and numerical methods, they are not a true design method for simulating the correct loading conditions and modeling rock mass properties accurately. Centrifuge testing of rocks also requires somewhat larger

model dimensions, compared to soil testing in order to include discontinuities in the model. Larger model dimensions require a centrifuge that besides a high acceleration, also can handle a large mass. Unfortunately, these two objectives do not easily meet simultaneously.

(6) Probabilistic methods necessitate very extensive numbers of input values and suppositions regarding the distribution functions. Furthermore, probabilistic design methods worked often based on the limit equilibrium methods. Thus, these are exposing to the same limitations as limit equilibrium methods (LEM).

Integrating costs into design methods through probabilistic methods can be some extent to accomplish. Nevertheless, the large amount of required input data has difficultly rendered the use of these cost-benefit-methods in practical applications [Adopted from 166].

4. Study Area: Sungun Copper Mine

4.1. General characteristics of Sungun copper mine

Sungun copper deposit is located in east Azarbaijan province of Iran in mountainous area and 75 km North West of the provincial town of Ahar at Latitude $38^{\circ} 41' 34''$ north and Longitude $46^{\circ} 41' 54''$ east (Fig. 1) [281]. Mine linked to Tabriz city via approximately a 125-km road. The ore body is situated straightly and nearly west of the Sungun River [Adopted from 281] a deeply incised valley in mountainous country with topography locally ranging from 1700 meters to 2450 meters. This deposit is in the middle of Qarabagh Mountains that highest altitude of the area from open sea is about 2390 meters [281]. Pakhir and Sungun Rivers are streaming through the mine's area after joining to Mian-cafe River which they make Ilgene-chai River. Climate condition in winter is cold. In the summer, the weather status is moderate. The area is compactly covered by plants and jungle as a result of its humidity. Maximum temperature in summer is 33° and minimum is -16° in winter [Adopted from 186].



Figure 1. Location of Sungun copper mine: map and satellite view, A. Overall map view, B. Detailed map view, C. Detailed satellite view [175].

Historical evidence and documents confirm long-term and ancient mining traditional activities in this area. NICICO Company (Taskmaster is National Iranian Copper Industries Company) in Sungun Copper Project.) is operating the Sungun Copper mine. That is a world-class project of great magnitude and complexity. The Sungun copper porphyry deposit would exploit through this open pit mine which will produce in four or five pushbacks. This roughly semicircular pit has high walls on three sides and a low wall to the east in the Sungun Valley. The ultimate open pit's initial design indicated a maximum slope height of 765 meters. The interim pit approximately has a maximum height of 675 meters. Therefore, this pit categorized to be a high to a very high slope and considered too. Height of the bench is about 12.5 meters, and width of main road is 30 meters; safety bench width is 11 meters, bench slope angle is 70° and final pit slope angle is 37° that design for different walls [Adopted from 1, 111, 281, 292].

4.2. Geological settings of Sungun copper mine

Mohammad Salehi Alashti.; East African Scholars Multidiscip Bull; Vol-1, Iss-3 (Dec, 2018): 61-121

The Azarbaijan application area is part of the comprehensive and universal copper belt of ALP-HIMALIA and the orebody surrounding region placed in the Alpine-Himalayan metallogenic belt. Location of the project in major porphyry copper-gold occurrences belt in Iran shown in Fig. 2. Sub-volcanic rocks have intruded into cretaceous limestone and andesite volcanics. Mineralization in monzonite and volcanic host rocks is disseminated, and stock works in nature. Characteristic vertical zoning of porphyry deposits as Leached, Supergene, Hypogene and Skarn zones have been recognized as being significant in regard to grade. Three ages of dykes exist and injected into the ore-body that considered waste [Adopted from 82, 111, 292].

In briefly, Sungun copper ore deposit is Porphyry that encompass via dikes. Mine's area in the lithological approach divided in to Sungun Porphyry ore body, dyke A, dyke B and skarn. Type of rock in porphyry ore body is Monzonite up to Quartz-Monzonite. Rock type in dyke A is Monzonite and in dyke B is Diorite. Fig. 3 presents a snap shot of the 3-dimensional dyke model. Green represents type 1A dykes (DK1A), whereas brown indicates DK1B and DK1C dykes. There are kinds of alteration like Phyllic (sericite, kaolinite, quartz and pyrite), Potassic (potassic feldspar, quartz and secondary biotite), Propylitic (epidote, chlorite and calcite), Argillic (kaolinite, clays, sericite) and Silicic (mainly fine-grained quartz) in the mine's area [Adopted from 111]. Mine's area divided in to six blocks that Situation of these blocks of Sungun copper mine presents in Fig. 4. In addition, general features of block five of Sungun mine shown in Table 2.

Block five of Sungun copper mine is selected for this purpose because only is suitable for fully drained conditions and other blocks face with slightly, partially or whole considerable water problems (partially drained, un-drained conditions) which a set of water influence factor should be contemplated in study formulas. Other conditions will consider in other articles. Table 3 shown the lithological log summary of bore hole 15 (GT15) in the section RS06 of block five of Sungun copper mine.

Table 2: General leatures of block live of Sungun copper line.									
Block	Lithology	Average of rock	Average of	Average of	Average of	Internal			
number		mass	rock mass	rock mass	cohesion	friction			
		Compressive	tensile	tensile young		angle(°)			
		strength (MPa)	strength modulus						
			(MPa)	(MPa)					
5	Monzonite and	4.184	0.007	1683	220-240	33-35			
	Diorite Dykes								

Table 2: General features of block five of Sungun copper mine.

Table 3: Summary of lithological log of bore hole15 (GT15) in section RS06 of block five of Sungun c	copper
project.	

	LITHOL	OGY	1	ALTERA	TION	M	NERALI	ZATION		ZONE		
FRO	ТО	TYPE	FRO	ТО	ТҮР	FRO	ТО	ТҮРЕ	FROM	ТО	TYPE	
Μ			Μ		Е	Μ						
0.00	1.60	CORE	0.00	1.60	-	0.00	1.60	-	0.00	1.60	-	
		LOSS										
1.60	5.50	SP	1.60	30.90	PHY	1.60	43.10	PYY-FEX-MNX	1.60	5.50	LEA	
5.50	6.10	DK1a	30.90	41.20	-	43.10	52.40	PYY-FEX-CHA	5.50	6.10	DY	
6.10	30.90	SP	41.20	43.10	PHY	52.40	78.00	PYY-FEX	6.10	12.00	OXI	
30.90	41.20	Ex-Ep	43.10	52.40	-	78.00	122.0	PYY-FEX-CHA	12.00	30.90	LEA	
							0					
41.20	43.10	SP	52.40	162.0	PHY	122.00	144.0	PYY-FEX-CHA-	30.90	41.20	-	
				0			0	CPY-BOR-MDL				
43.10	52.40	Ex-Ep	162.00	192.7	POT	144.00	192.7	PYY-CPY-MDL	41.20	43.10	LEA	
				0			0					
52.40	192.7	SP	192.70	194.6	PHY	192.70	194.6	-	43.10	52.40	-	
	0			5			5					
192.70	194.6	DK1b	194.65	198.9	POT	194.65	198.9	PYY-CPY-MDL	52.40	88.00	LEA	
	5			5			5					
194.65	198.9	SP	198.95	203.3	PHY	198.95	203.3	-	88.00	122.00	SUP	
	5			5			5					
198.95	203.3	DK1b	203.35	300.0	POT	203.35	210.0	PYY-CPY-MDL-	122.00	142.00	SUP-	
	5			0			0	BOR			HYP	
203.35	300.0	SP				210.00	300.0	PYY-CPY-MDL	142.00	192.00	HYP	
	0						0					
									192.00	194.00	DY	
									194.00	198.95	HYP	
									198.95	203.35	DY	
									203.35	300.00	HYP	

Note 1. TYPE OF LITHOLOGY- SP: Sungun Porphyry (Monzonite to Quartz Monzonite); DK1a: Dyke 1a (Diorite Porphyry, Very altered, Similar to Sungun Porphyry); Ex-Ep: Exoscarn-Epidote Zone; DK1b: Dyke 1b (Diorite Porphyry, Altered, Argillic Alteration).

Note 2. TYPE OF ALTERATION- POT: Potassic Altered; PHY: Phyllic (quartz-sericite) altered.

Note 3. TYPE OF MINERALIZATION- PYY: Pyrite; FEX: Iron Oxide; MNX: Manganese Oxide; CHA: Chalcocite; CPY: Chalcopyrite; MDL: Molibdonite.

Note 4. TYPE OF ZONE- LEA: Leached; DY: Dyke; OXI: Oxide; SUP: Supergene; SUP-HYP: Supergene-Hypogene; HYP: Hypogene.



Figure 2. Location of project in major porphyry copper-gold occurrences belt in Iran - modified [199].



Figure 3. Three-dimensional lithological model of dykes (DK1A, DK1B and DK1C) in mine area looking northwest [281].



Figure 4. Domains, Pit Sectors and Cross Section Locations of Sungun Copper Mine [281].



Figure 5. General layouts of Sungun copper mine [281].

General layout of Sungun copper mine shown in Fig. 5. Also, geological section of borehole 15 (GT15) in block five of Sungun copper mine presented in Fig. 6. LS in Fig. 6 means loss of core.



Figure 6. Geological section for borehole 15 (GT15) in block five of Sungun copper mine.

5. Basic Theory and Calculation: Strength Limit of Earth; a New Basis Parameter

5.1. Strength limit of earth's explanation

Earth's strength limit is a key basis parameter for earth perceptions and interpretations. Materials' cohesion between particles varies through different reasons like structural problems, tectonic actions, geological structures and engineering geology problems. If these values are less than about, 65 percent (exactly equal to 64.712157172243419172410449319661 %) of own first value therefore instability changes begin. In such condition, micro cracks create and develop. Then, finally different results of instability such as failures occur, e.g., in mines' slopes and on side of the roads. Also in luminaries or other structures. Strength limit of earth represented that how much value remain active from one MPa or one percent of any strength or set of all resisting forces. Strength limit is the cohesion value of different materials, particles, elements, factors and parameters on earth and the stability limit of earth and allconstitutive particles and elements of earth and earth's visible or invisible constitutive components in presence of all disturbance factors such as different stresses. This Actual Multi-Interaction Strength (A.M.I.S) in existence illustrates the equilibrium and equilibrium limit between resisting and disturbance forces acting on earth during universe stability. Cohesion limit number is constant ($X_{C-Constant}$ or X_{CCLN}) instantly and on the other hand is variable that called cohesion variable limit number ($X_{C-variable}$ or X_{CVLN}). When strength limit parameter decreases continuously and reach to $[(X_{C-Earth}) \times 100]\%, ([\frac{X_{C-Earth}+G.S.Ns}{2}] \times 100)\%$ and (Golden section numbers× 100) % of theirs first own value then instabilities of structures like earth begin. The Golden section number is 0.6180339887. Thus, in critical stages' procedures strength limit parameter pass from [(1-G.S.Ns) × 100] % and $\left(\left[1 - \frac{X_{c-Earth} + G.S.Ns}{2}\right] \times 100\right)$ % of its primary value and ruptures begin in structures. Therefore, complete decomposing times and procedures occur during and after that strength limit parameter reduce yet and its value pass from [(1-X_{C-Earth}) ×100] % of preliminary value. Variations diagram of this interactional resistance realizes from nature and exactly from shape of the mountains. Strength limit can define for every structure and everything in various fields of science [Adopted from 246].

5.2. Definitions of strength limit of earth $(X_{C-Earth})$

Scrutinizing researches about gravitation's effects, its relations and influences on other sciences are major works, which required for future of sciences perfection. One branch of this basis project should be carried out with respect to

connect strength limit factor such as Earth's Strength Limit and rock strength especially strength of various rock(s) types like as intact, highly strength also jointed and altered rocks. Strength Limit of Earth is an identifier parameter, which linked other sciences to rock engineering and geology sciences also slope engineering too, to illustrate extensible detailed complicated relations of all sciences' parts, which acquit roles in the entire existence.

Determining the strength limit of earth parameter can organize, also adjusts and completes more theories. Therefore, this essential and efficient factor helps response to different unsolved questions in various fields of sciences.

5.2.1. Independent determination of strength limit by area density and standard gravity for earth

One of the principal basilar tensional forces that affect materials in the world is gravity. Gravitational effects are one of the regulator forces of all things in the universe. Gravity of earth denotes to space and time with fundamental parameters. Strength limit factor of earth is the strength of gravity forces also is the structural resistance of gravitational fields. Mainly, strength limit of earth indicates the equilibrium or equilibrium limit between resisting and disturbing forces in the universe gravitational field and plays considerable and indispensable key roles in cosmos universal stability field that is the reason to applying this factor in slope stability engineering analysis. In addition, Napier's or Euler's constant is a key number in universe explanation. Therefore, gravity and Napier's constant can use in definitions of earth's strength limit. Strength limit number of the earth also is the cohesion constant limit number of the earth that characterized through standard gravity and Euler's constant by below equation [Adopted from 246]:

$$X_{C-Constant} = X_{C-earth} = \frac{g_0}{e^e} = g_0 \times \frac{1}{\exp^e} = gravity \times density = 0.64712157172243419172410449319661 \text{ (MPa or \%)}$$
(1)

Where g_0 is the standard gravitational acceleration or Earth's standard surface gravity that's equal to 9.80665 m/s² [284, 294], e is the simply Euler's constant or Napier's constant, $X_{C-earth}$ [Adopted from 246] is the cohesion constant limit number of earth or strength limit parameter or strength of existence global gravitation (in MPa), e is Napier's number, $\frac{1}{e^e}$ is area density (A.D or D_A) or surface density (S.D or D_S) and equal to 0.06598803584531253707679018759685 Ggr/m² 65.98803584531253707679018759685 ton/m².

 e^e is equal to 15.15426224147926418976043027263 m^2/Ggr which means every value of gravitation is effect on e^e square meter of every part of earth or everything near the earth with one Ggr mass. In-fact e^e is the impact surface area of gravity on one Ggr mass. Surface density means every value of gravitation is effects on approximately 0.066 Ggr in one square meter of every part of earth or everything near the earth. In fact, area density is 65.98803584531253707679018759685 ton in one square meter also equal to 1000 tons (1 Ggr) in 15.15426224147926418976043027263 square meter (m^2) . Then, area density defined by [Adopted from 246]:

$$A.D = Area \ Density = S.D = Surface \ Density = \frac{1}{e^e} = \frac{1}{\sqrt{e^e} \times \sqrt{e^e}} = \frac{X_{C-earth}}{g_0} = \frac{Strength \ Limit \ of \ Earth}{Standard \ Gravity}.$$
(2)

Mathematical and numerical roots of strength limit of earth have different meanings. Eq. (1) represent that strength limit of earth approximately is equal to 0.65 (MPa or %) from every 1 (MPa or %) unit scale of any strength forces or set of all kinds of resisting forces. However, this parameter approximately is equal to 64.7122 (MPa or %) in every 100 (MPa or %) unit scale(s) of any strength forces or set of all kinds of resisting forces. In other words, earth's strength limit is equal to 0.647122 MPa or 64.7122 % approximately from any strength forces or set of all resisting forces [Adopted from 246].

Why 32 decimal digits used for some key numbers? "Because" an extremely small discrepancy in key numbers causes making great differences in features of various sciences [Adopted from 247].

5.2.2. Specification of strength limit by Newton's second law of motion and Newton's law of universal gravitation and standard gravity

Briefly, Definition of Newton's Second Law is the involved net force F of a body is directly equal to the scalar multiplication of the mass m and the acceleration ("a") of mentioned body [Adopted from 204, 205].

According to Newton's second law of motion and universal gravitation law below relation is feasible based on Eq. (1) [Adopted from 58,79,141,165,187,188,189,190,204,230,308]:

$$F = ma = mg = m\frac{X_{C-earth}}{A.D} = \frac{M \times G \times E}{R^2} , \qquad (3)$$

The mass of the earth is equal to [Adopted from 246]:

$$E = ED \times V_{Earth} \tag{4}$$

Then, cohesion constant limit number of the earth with impact of earth density according to Eqs. (3) and (4) written as [Adopted from 246]:

Mohammad Salehi Alashti.; East African Scholars Multidiscip Bull; Vol-1, Iss-3 (Dec, 2018): 61-121

$$X_{C-Earth} = \frac{M}{m} \times \frac{G \times E}{R_{Earth}^2 \times e^e} = \frac{M}{m} \times \frac{G \times ED \times V_{Earth}}{R_{Earth}^2 \times e^e} = \frac{M}{m} \times \frac{G \times ED \times V_{Earth} \times A.D}{R_{Earth}^2} \cong 0.647122$$
(5)

Many fundamental laws can be deriving and clarifying from substantial sciences by Strength limit factor of earth such as laws which put parallel beside Newton's notable laws [Adopted from 246].

Then, standard gravity expressed by below relation without deleting m and M from two sides of Eq. (3) [Adopted from 247]:

$$g_{0} = \frac{F}{m} = \frac{M}{m} \times \frac{G \times E}{R^{2}} = R^{'} \times \frac{G \times E}{R^{2}} = \frac{G \times E}{R^{'}} = \frac{X_{C-Earth}}{A.D} = \frac{X_{C-earth}}{\frac{1}{exp.^{\epsilon}}}.$$
(6)

Furthermore, follow relation deduced via the Eq. (6) [Adopted from 247]:

$$\frac{g_0 \times R^2}{G} = E \times R^{\dagger} = ED \times V_{Earth} \times R^{\dagger} = 242892793676594.55573513891084111 \text{ Kg.m.}$$
(7)

The ratio of every point mass M to mass m in calculations is equal to the distance between the masses only from quantitative aspect and can be written as [Adopted from 247]:

$$\frac{M}{m} = R^{'} = \frac{R_{Earth}^{2}}{R^{'}} = \frac{g_{0} \times R_{Earth}^{2}}{G \times E} = 40656957195372.527825506161635217.$$
(8)
$$R^{'} = \frac{R_{Earth}^{2}}{R^{'}} = \frac{R_{Earth}^{2}}{\frac{M}{m}}.$$
(8.1)

Where.

 $R' = R_{Earth-Dimensionless} = The Quantitative Value of R = Dimensionless R$

Cohesion limit number of the earth with impact of earth density clearly expressed as [Adopted from 247]:

$$X_{C-Earth} = \frac{M}{m} \times \frac{G \times E}{R_{Earth}^2} \times A.D = \frac{G \times E}{R^* \times exp.^e} = \frac{G \times ED \times V_{Earth}}{R^*} \times A.D = 0.64712157172243419172410449319661.$$
(9)

$$X_{C-Earth} = G \times ED \times V_{Earth} \times \frac{R_{Earth} - Dimensionless}{R_{Earth}^2} \times A.D = G \times ED \times V_{Earth} \times \frac{R_{Earth} - Dimensionless}{R_{Earth}(inm) \times R_{Earth}(inm)} \times A.D = \frac{G \times ED \times V_{Earth}}{R_{Earth}} \times A.D.$$
(10)

Where [Adopted from 247] F is the force between the masses (in N), G is the Newtonian gravitational constant or Newtonian constant of gravitation that's approximately equal to $6.67384 \times 10^{-11} \pm 0.00080 \times 10^{-11} \text{ m}^3 \text{kg}^{-1} \text{s}^{-2}$ [104] or 6.67384×10⁻¹¹ N m² kg⁻² [187, 188, 189, 190], **E** (also identified by \mathcal{M}_{\oplus}) is the mass of the earth that's equal to $\mathcal{M}_{\oplus} = 5.9742 \times 10^{24} Kg$ [56], **M** is the every point mass (in Kg), **R**_{Earth} or **R** is the distance between the masses (in m) which equal to 40656957195372.527825506161635217 m, ED is the earth density (in Kg/m³) that's equal to 5.5152740465837649209294596615615 gr/cm³, V_{earth} is the volume of earth (in m³) and equal to 1.08321×10^{12} km³ or 1.08321×10^{27} cm³ [56] (2.59876×10¹¹ cu mi [311]), **R'** or **R**_{Earth-Dimensionless} is the ratio of every point mass M to mass m or is the quantitative radius value of earth without unit (dimensionless) and equal to 40656957195372.527825506161635217 (dimensionless), \mathbf{R}'' is equal to 40656957195372.527825506161635217 m², $\mathbf{R'}_{Earth}$ is equal to 40656957195372.527825506161635217 m².

6. Variable Strength Limit of Earth (X_{C-variable} or X_{CVLN})

In the Theory section, it is necessary to give a theoretical or methodical basis that will be required for obtaining results of the paper.

The universe has been expanding and earth has twitching therefore density, volume, mass and radius of earth varying. In addition, Earth's gravity, mass, density, volume and strength limit of earth have direct proportion together. Thus, cohesion variable limit number for earth (variable strength limit number of earth) must be defined. Variable strength limit number of earth changing according to earth's variations of features defined by below equations:

$$X_{C-variable} = \frac{M}{m} \times \frac{G \times E_{V-Earth}}{R_{V-P}} = \frac{M}{m} \times \frac{G \times ED_{V-Earth} \times V_{V-Earth}}{R_{V-P}} = \frac{M}{m} \times \frac{G \times ED_{V-Earth} \times V_{V-Earth} \times A.D}{R_{V-P}}.$$
 (11)

$$X_{CVLN} = X_{CCLN} \times \left(\frac{r_e}{r_e + h}\right)^2 \,. \tag{12}$$

Therefore, according to Eq. (1) will have:

$$X_{C-variable} = X_{CVLN} = g_V \times A.D = Variable Gravity \times Area Density$$
(13)

Where X_{CVLN} or $X_{C-variable}$ is the variable strength limit of earth or cohesion variable limit number of earth (in MPa), \mathbf{X}_{CCLN} is the cohesion constant limit number (in MPa), \mathbf{g}_V is the variable gravity (in m/s²), $\mathbf{E}_{V-Earth}$ is the variable mass of the earth (in Kg), $\mathbf{R}_{V-Earth}$ is the variable distance between the masses (in m), $\mathbf{ED}_{V-Earth}$ is the variable earth density (in Kg/m³), $V_{V-Earth}$ is the variable volume of earth (in m³).

Area density of earth and gravitational factor would change in accordance to changing the earth's features if the changes did not perform with constant proportion. Hence, variable strength limit based on Eqs. (11) and (13) defined as:

(8.1)

Mohammad Salehi Alashti.; East African Scholars Multidiscip Bull; Vol-1, Iss-3 (Dec, 2018): 61-121

$$X_{C-variable} = \frac{M}{m} \times \frac{G_{V} \times E_{V-Earth} \times AD_{V}}{R_{V-Earth}^{2}} = \frac{M}{m} \times \frac{G_{V} \times ED_{V-Earth} \times V_{V-Earth} \times AD_{V}}{R_{V-Earth}^{2}} = \frac{M}{m} \times \frac{G_{V} \times ED_{V-Earth} \times V_{V-Earth} \times AD_{V}}{R_{V-Earth}^{2}}.$$

$$X_{C-variable} = X_{CVLN} = g_{V} \times AD_{V} = Variable Gravity \times Variable Area Density$$
(15)

Where \mathbf{G}_{V} is the gravitational variation, \mathbf{g}_{V} is variable standard gravity, \mathbf{AD}_{V} is variable area density of earth.

7. Definition of Cohesion Variable Limit Number for Slope

For the reason of importance, complicating, different and wide applications of X_C parameter in existence descriptions like earth and various sciences, in this article only presents a specific definition of this parameter about slope stability field. X_C parameter according to one approach is variable for different parts of earth like a slope that's call cohesion variable limit number for the slope ($X_{CV-Slope}$ or $X_{CV(Slope)}$) or $X_{CVLN-Slope}$), even for different parts of a mine (or different blocks of a mine). In the other words, if a mine is divided into eight blocks this parameter is different for every mine's block also is variable from one block to another block. Earth has a constant or standard strength limit number as same as its standard gravity. In addition, different parts of earth have different and variable strength limit number as same as various things near or on earth which has variable gravity in diverse situations. Therefore, several parts of slopes have varied strength limit number. Thus, variable strength limit number for a slope defined by:

 $X_{\text{CVLN-Slope}} = X_{\text{CV-Slope}} = A - B + C - X_1 + D - E = (A + C + D) - (B + X_1 + E)$ (16)

A, B, C, D and E variables in above relation are as follows:

$$\mathbf{A} = \frac{\overline{\emptyset}_i \times C_{imin}}{\overline{C}_i \times \emptyset_{imin}} \tag{17}$$

$$B = \left(MEARS \times \left(\frac{\overline{C}_i \times \emptyset_{imin}}{\overline{\emptyset}_i \times C_{imin} \times C_{imax}}\right)\right) = \left(MEARS \times \left(\frac{1}{A \times C_{imax}}\right)\right)$$
(18)

$$B = \left(10 \times \left(\frac{\overline{C}_i \times \emptyset_{imin}}{\overline{\emptyset}_i \times C_{imin} \times C_{imax}}\right)\right) = \left(10 \times \left(\frac{1}{A \times C_{imax}}\right)\right)$$
(19)

$$C = \left(\frac{1}{100} \times \frac{\overline{C}_i \times \emptyset_{imin}}{\overline{\emptyset}_i \times C_{imin}}\right) = \left(\frac{1}{100} \times \frac{1}{A}\right)$$
(20)

$$D = \frac{1}{1000} \left(\overline{\mathbf{V}} \times \left(\frac{\overline{C}_i \times \mathcal{O}_{imin} \times \mathcal{O}_{imax}}{\overline{\mathcal{O}}_i \times C_{imin} \times C_{imax}} \times \text{MEARS} \right) \right) = \frac{1}{1000} \left(\overline{\mathbf{V}} \times \left(\frac{1}{A} \times \frac{\mathcal{O}_{imax}}{C_{imax}} \times \text{MEARS} \right) \right)$$
(21)

$$D = \frac{1}{1000} \left(\overline{\mathbf{V}} \times \left(\frac{\overline{C}_i \times \mathcal{O}_{imin} \times \mathcal{O}_{imax}}{\overline{\mathcal{O}}_i \times C_{imin} \times C_{imax}} \times 10 \right) \right) = \frac{1}{1000} \left(\overline{\mathbf{V}} \times \left(\frac{1}{A} \times \frac{\mathcal{O}_{imax}}{C_{imax}} \times 10 \right) \right) = \frac{1}{1000} \left(\overline{\mathbf{V}} \times \left(\frac{1 \times \mathcal{O}_{imax}}{A \times C_{imax}} \right) \right)$$
(22)

$$E = \left[(10^{-2} \times \left(\overline{T} + \frac{H+D}{2} + \frac{\beta}{100} \right) \times (H+D + \frac{\beta}{100}) \right] = \left[\left(10^{-2} \times \left(\overline{T} + \frac{100(H+D) + 2\beta}{200000} \right) \right) \times \left(H+D + \frac{\beta}{100} \right) \right]$$
(23)

$$E = \left(\frac{(2 \times 10^5 \times \overline{T}) + 100(H + D) + 2\beta}{2 \times 10^7}\right) \times (H + D + \frac{\beta}{100})] = \left(\frac{\left[(2 \times 10^5 \times \overline{T}) + 100(H + D) + 2\beta\right] \times [100H + 100D + \beta]}{2 \times 10^9}\right]$$
(24)

Then, simple forms of Eq. (13) according to Eqs. (16), (17), (18), (20), (21) and (23) are as below:

$$X_{\text{CVLN-Slope}} = X_{\text{CV(Slope)}} = A + \frac{1}{A} \left[\left(\frac{1}{100} \right) + \left(\frac{\overline{\text{V}} \times \emptyset_{imax} \times \text{MEARS}}{1000 \times C_{imax}} \right) - \left(\frac{\text{MEARS}}{C_{imax}} \right) \right] - X_1 - E$$
(25)

$$\mathbf{X}_{\text{CVLN-Slope}} = \mathbf{X}_{\text{CV(Slope)}} = A + \frac{1}{A} \left[\left(\frac{1}{100} \right) + \left(\frac{\overline{\mathbf{V}} \times \boldsymbol{\varnothing}_{imax}}{100 \times C_{imax}} \right) - \left(\frac{10}{C_{imax}} \right) \right] - X_1 - E$$
(26)

Where *MEARS* is the minimal effective action and reaction unit of strength and equal to 10 MPa, [Adopted from 246] $X_{CVLN-Slope}$ or $X_{CV-Slope}$ or $X_{CV(Slope)}$ is the cohesion variable limit number for a slope (in MPa), $\overline{V}_{Assessing Mine Block} = \overline{V} = \overline{V}_{MB}$ is the rock mass index values average to sum of the rock mass index values of the assessing mine block ratio, X_1 is the detector parameter (dimensionless), \overline{C}_i is the Cohesion values average of slope components (in kPa), C_{imax} is the Maximum Cohesion value of slope components (in kPa), C_{imax} is the Maximum internal friction angle of slope components (in degree), ϕ_{imax} is the Maximum internal friction angle of slope components (in degree), ϕ_{imax} is the Slope components (in MPa), \overline{T} is the average of the rock mass tensile strength of the assessing mine block (in MPa), H is the seismic acceleration, D is the disturbance factor (dimensionless), β is the overall slope angle (in degree).

 X_1 in Eqs. (16), (25) and (26) is a detector parameter that is given by below equations:

$$\overline{\mathbf{V}} = \frac{\overline{RMI}}{RMI} \tag{27}$$

Available Online: http://www.easpublisher.com/easmb/

$$\exp^{\overline{V}} = 1 + \overline{V} + X_1 \tag{28}$$

(29)

Then, X₁ factor obtained by:

For example:

$$IF\overline{V} = 0.0042 \Longrightarrow e^{\overline{V}} = exp^{.0.0042} = 1 + 0.0042 + 0.000008832 \Longrightarrow X_1 = 0.000008832$$

In Eq. (28) e is the Napier number. In addition, the minimal effective action and reaction unit of strength obtained by [Adopted from 246]:

$$MEARS = \frac{b}{Decay Constant(yr^{-1})} = \frac{b}{D.C(yr^{-1})} = 10 MPa.$$
(30)

 $X_1 = exp.\overline{V} - \overline{V} - 1$

Where [Adopted from 246] *MEARS* is the minimal effective action and reaction unit of strength and equal to 10 MPa, **D**. **C** is the decay constant of gravitational particles and currently exact value for the decay constant of gravitational particles which close to the decay constant of ⁸⁷Rb and is equal to 1.4221904973467736128057926319335 × 10⁻¹¹ yr⁻¹, b is strength variations coefficient in 10¹⁰ years and equal to b = 1.4221904973467736128057926319335 × 10⁻¹⁰ in $\frac{MPa}{Year} = \frac{MN}{Year} = \frac{Ggr}{Year m s^2}$.

Strength variations coefficient calculated through one of the relations between age of the earth and strength limit number of earth accordance to Eq. (2) as below equations [Adopted from 246]:

$$X_{C-earth} = AOE \times p.$$
(31)

$$b = \frac{X_{C-\text{earth}}}{AOE} = \frac{g_0 \times \frac{1}{\exp^{\circ}}}{AOE} = \frac{g_0 \times A.D}{AOE}.$$
 (32)

Where [Adopted from 246] $X_{c-earth}$ is strength limit number of earth and equal to 0.64712157172243419172410449319661 MPa (this value calculated according to old standard gravity (g_{0-0ld}) which is equal to 9.80665 $\frac{m}{s^2}$), *AOE* is the age of the earth's abbreviation and old age of the earth according to old standard gravity is equal to 4.5501750498945017745329167415034 Billion Years.

Age of the earth according to old standard gravity (AOE_{old}) is equal to 4.5501750498945017745329167415034 Billion Years also the strength limit parameter of the earth according to old standard gravity ($X_{C-Earth,old}$) is equal to 0.64712157172243419172410449319661 MPa [Adopted from 246].

Strength limit factor of earth changed by changing gravity of earth. Above values for $X_{c-earth}$ and AOE computed according to old standard gravity. Accurate or exact age of the earth according to present standard gravity at present time (AOE_{Exact}) is equal to 4.7303652362904 Billion Years. Then, strength limit parameter of earth or strength of existence global gravitation according to present standard gravity ($X_{C-Earth,Now}$) is equal to 0.6727480488031732255208148942682 MPa. The present value of standard gravity (g_{0-Now}) is equal to 10.195000354006777301014825930477 m/s² [Adopted from 246].

What are the meanings of old and present values of gravity, age and strength limit of the earth?

Creation procedures of the earth started in 4.7303652362904 Billion Years ago with 0.673 MPa strength limit value and 10.2 m/s² gravity. Earth created completely (fully whole) in 4.5501750498945017745329167415034 Billion Years ago with approximately 0.65 MPa value of strength limit and 9.80665 m/s² of the gravity in whole creation point. Gravity value along the 180190186.4 years' creation process of the earth from 4.7304 Billion Years ago to 4.56 Billion Years ago was higher than now and equal to 10.2 m/s². These momentous and epochal discoveries destine also desquamates the covered key of the sciences' fast perfections.

Meanwhile, all above formulas true based on old and present values of standard gravity, strength limit and age of the earth therefore according to Eqs. (31) and (32) have:

1

$$X_{C-Earth,Old} = AOE_{Old} \times p.$$
(33)

$$X_{C-Earth,Now} = AOE_{Exact} \times p.$$
(34)

$$b = \frac{X_{C-Earth,Old}}{AOE_{old}} = \frac{g_{0-Old} \times \frac{1}{exp.^{e}}}{AOE_{old}} = \frac{g_{0-Old} \times A.D}{AOE_{old}}.$$
(35)

$$b = \frac{X_{C-Earth, Now}}{AOE_{Exact}} = \frac{g_{0-Now} \times \frac{1}{exp.^{e}}}{AOE_{Exact}} = \frac{g_{0-Now} \times A.D}{AOE_{Exact}}.$$
(36)

Earth's historical features involve in calculation of variable strength limit for slopes and its component factors. In addition, we have below equations in accordance to Eqs. (18), (16), (30), (35), and (36):

Mohammad Salehi Alashti.; East African Scholars Multidiscip Bull; Vol-1, Iss-3 (Dec, 2018): 61-121

$$B = \left(\frac{b = \frac{X_{C-Earth,Old}}{AOE_{Old}} = \frac{g_{0-Old} \times \frac{1}{exp.^{e}}}{AOE_{Old}} = \frac{g_{0-Old} \times A.D}{AOE_{Old}}}{Decay Constant(yr^{-1})} \times \left(\frac{\overline{C}_{i} \times \emptyset_{imin}}{\overline{\emptyset}_{i} \times C_{imin} \times C_{imax}}\right)\right)$$
(37)

$$B = \left(\frac{b = \frac{X_{C-Earth, Now}}{AOE_{Exact}} = \frac{g_{0-Now} \times \frac{1}{exp^{\epsilon}}}{AOE_{Exact}} = \frac{g_{0-Now} \times A.D}{AOE_{Exact}}}{Decay Constant (yr^{-1})} \times \left(\frac{1}{A \times C_{imax}}\right)\right)$$
(38)

$$D = \frac{1}{1000} \left(\overline{\mathbf{V}} \times \left(\frac{\overline{C}_{i} \times \emptyset_{imin} \times \emptyset_{imax}}{\overline{\emptyset}_{i} \times C_{imin} \times C_{imax}} \times \text{MEARS} \right) \right) = \frac{1}{1000} \left(\overline{\mathbf{V}} \times \left(\frac{1 \times \emptyset_{imax}}{A \times C_{imax}} \times \frac{b = \frac{\mathbf{X}_{\text{C-Earth, Old}}}{AOE_{\text{Old}}} = \frac{\mathbf{g}_{0-\text{Old}} \times \frac{1}{\text{AOE}_{\text{Old}}}}{AOE_{\text{Old}}} = \frac{\mathbf{g}_{0-\text{Old}} \times A.D}{AOE_{\text{Old}}} \right) \right)$$
(39)

$$D = \frac{1}{1000} \left[\overline{V} \times \left(\frac{1}{A} \times \frac{\emptyset_{imax}}{C_{imax}} \times \frac{b = \frac{X_{C-Earth,Now}}{AOE_{Exact}} = \frac{g_{0-Now} \times \frac{1}{exp.^{e}}}{AOE_{Exact}} = \frac{g_{0-Now} \times A.D}{AOE_{Exact}}}{Decay \operatorname{Constant}(yr^{-1})} \right) \right]$$
(40)

Therefore, variable strength limit for slopes in accordance to Eqs. (25), (27), (28), (35), and (36) and also based on old and present value of earth's strength limit with RMi effects obtained by:

$$X_{CVIN-Slope} = A + \frac{1}{A} \left[\left(\frac{1}{100} \right) + \left(\frac{\overline{RMI}}{\overline{RMI}_{Sum}} \times \bigotimes_{imax} \times \frac{b = \frac{X_{C-Earth,Oid}}{AOE_{Oid}} = \frac{g_{0-Oid} \times AD}{AOE_{Oid}}}{1000 \times C_{imax}} \right) - \left(\frac{\frac{X_{C-Earth,Now}}{AOE_{East}}}{C_{imax}} \times \frac{g_{0-Slow} \times AD}{AOE_{Oid}}}{\frac{Decay Constant(yr^{-1})}{C_{imax}}} \right) \right] - \left(\frac{\frac{X_{C-Earth,Now}}{AOE_{East}}}{C_{imax}}}{C_{imax}} \right) - \left(\frac{\frac{X_{C-Earth,Now}}{AOE_{East}}}{C_{imax}}}{C_{imax}} \right) \right] - exp^{\overline{V}} - \overline{V} - 1 - E$$

$$X_{CVIN-Slope} = \frac{\overline{O}_{i} \times C_{imin}}{\overline{C}_{i} \times \bigotimes_{imin}} + \frac{1}{\overline{O}_{i} \times C_{imin}}} \left[\left(\frac{1}{100} \right) + \left(\frac{\overline{RMI}}{RMI_{siom}} \times \bigotimes_{imax} \times \frac{b = \frac{X_{C-Earth,Oid}}{AOE_{Oid}}}{Decay Constant(yr^{-1})}} \right) - \left(\frac{\overline{C}_{C-Earth,Now}}{AOE_{Cout}} - \frac{g_{0-Now} \times AD}{AOE_{Oid}}}{\frac{AOE_{Oid}}{AOE_{Oid}}} \right) - \left(\frac{X_{C-Earth,Now}}{AOE_{East}} - \frac{g_{0-Now} \times AD}{AOE_{East}}}{Decay Constant(yr^{-1})}} \right) - exp^{\overline{V}} - \overline{V} - 1 - E$$

$$(41)$$

$$X_{CVIN-Slope} = \frac{\overline{O}_{i} \times C_{imin}}{\overline{C}_{i} \times \bigotimes_{imin}}} - \frac{\overline{RMI}}{\overline{C}_{i} \times \bigotimes_{imin}} - \frac{\overline{RMI}}{\overline{C}_{i} \times \widehat{O}_{imin}}}{\frac{\overline{C}_{i} \times \overline{O}_{imin}}}{2 \times 10^{2}}} + \frac{\overline{C}_{imin} \times (1 - 1)}{2 \times 10^{2}} + \frac{\overline{C}_{imin} \times (1 - 1)}{2 \times 10^{2}} + \frac{\overline{C}_{imin} \times \overline{O}_{imin}} + \frac{\overline{O}_{imin}}}{\overline{C}_{imin}} + \frac{\overline{O}_{imin}}{\overline{C}_{imin}} + \frac{\overline{O}_{imin}}{\overline{O}_{imin}} + \frac{\overline{O}$$

8. Procedures of Empirical Investigations: New Empirical Relations for Calculation of the Safety Factor in Fully Drained Conditions

Various relations between involved factors in existence such as rock engineering factors could use to slope stability analysis procedures.

The height of the slopes has a direct proportion to the changes of the stress levels in the slopes' rock mass also has face-to-face influence on it [Adopted from 106]. A high slope may also let more opportunities to evince of discontinuities associated failures to act as instabilities because the discontinuities' quantity which intersected by the slope is larger. High stress levels may lead to the slopes failures due to failure of the intact rock compared to the intact rock strength (Gama, 1989). Although slope's height is momentous in slope stability systems, but only the Haines and Shuk [264] system includes the slope height [Adopted from 106, 264]. High slope height included high-pressure causes instabilities to the slope structure, which needs to considered in a slope stability's assessment system. The overall slope angle, number of slope components and slope's characteristics and configurations are related to dewatering and drainage proportionally as a result of slope stability analysis. The overall slope angle has great importance application in drained, partially drained, fully drained even in un-drained conditions therefore play its roles in any conditions also slope configurations too.

If the slopes are not dewatered effectively, therefore, the overall slope angles will reduce significantly. The influence of ground water is often crucial to the stability of large open pit mine slopes [281]. Overall slope angle is an intra-factor, which carries the conditions of ground and underground water inside self and illustrates it too. The

disturbance parameter (D) is one of the strength reduction factors, which accounts for the disturbance and strength reduction of the in-situ rock mass by blasting or stress relaxation.

A D factor of less than or equal to 0.7 is applicable for free dig material. A D factor of one indicates maximum disturbance by blasting and is appropriate for open pit conditions [281]. Similar to many other porphyry deposit regions, the Sungun Copper Mine situated in an area with high seismicity. The relative importance of seismic accelerations is very significant in terms of the stable slope angles. Therefore, it is important to determine the appropriate earthquake accelerations for design's input [281]. A horizontal seismic acceleration coefficient indicates the one of the critical conditions for the open pit mines particularly for Sungun copper mine to specify damages during an earthquake in the lifetime of the mine. Thus, this specialty must consider precisely and carefully.

This factor in Sungun seismic study obtained from a re-interpretation of a seismic hazard evaluation report. Seismic hazard analysis for Sungun mine carried out based on relevant seismo-tectonic information for determine the peak ground acceleration (PGA) at an appropriate design level. The most stringent seismic design criterion is the Maximum Credible Level (MCL), and in further decreasing, order the levels are: Maximum Design Level (MDL), the Design Basis Level (DBL), and the Construction Level (CL)) [Adopted from 281]. Overall slope height (h_{OS}) is required to estimate the weight of the rock column overlying the estimated failure surface. This is also required to estimate σ_3 to calculate various parameters of the Hoek-Brown equation and for calculation of equivalent Mohr-Coulomb strength parameters, cohesion (c) and friction angle (phi) [281].

In presented equations in this part, the parameters requiring for safety factor calculation shown in Table 4:

Factors	Explanations	Units
FS _{fd}	Factor of safety for fully drained conditions.	Dimensionless
$OSH = H_{OS} = h_{OS}$	Overall slope height or is the maximum overall slope height.	Meter (m)
n	Number of components in a slope (number of slope's	Dimensionless
	components).	
\overline{RMi} or $RMI_{(av)}$	Average of rock mass index values for assessing mine block.	MPa
RMi _{Sum}	Sum of rock mass index values obtained from every approved	MPa
	components of a discovery boreholes in a slope for assessing	
	mine block.	
e	Napier number.	Dimensionless
Н	Seismic acceleration.	Dimensionless
D	Disturbance factor.	Dimensionless
$OSA = \beta$	Overall slope angle.	Degree
$IRA = \alpha$	Inter ramp angle.	Degree
$BFA = \Delta = \delta i$	Bench face angle.	Degree
$BFH = H_{BF}$	Bench face height.	Meter (m)
$SP.BW = W_{SP.B}$	Spill berm width.	Meter (m)
$SBW = W_{SB}$	Safety berm width.	Meter (m)
$IRH = H_{IR}$	Inter ramp height.	Meter (m)
$GBW = W_{GB}$	Geotechnical berm width.	Meter (m)
$\bar{C}_{components of slopes (c of s)}$	Average of cohesion values of slope components (average of a	MPa
	slope components cohesion values).	
C _{imin}	Minimum cohesion value of slope components.	kPa
$\overline{\emptyset}_{c \ of \ s}$	Average of internal friction angle values of slope components.	Degree
\overline{T}	Average of rock mass tensile strength in assessing mine block.	MPa
γ	Unit weight of rock in assessing mine block.	MN
		m^3
MEARS	Minimal effective action and reaction unit of strength.	MPa
X _{CVLN}	Cohesion variable limit number.	MPa

8.1. Safety factor computation via RMi and slope's properties using strength limit with an independent approach

Earth's strength limit factor can help identify and solve slope stability engineering problems. This feature of earth shows the equilibrium limit and limit equilibrium between resisting and disturbing forces in universal gravitational field and universe's overall stability field. Therefore, more required detailed data involved in this parameter that will uncover as a hidden potential. In addition, strength limit by these uncapping can apply the included equivalency between more influencer factors on slopes failures engineering analysis to solving slope stability problems without doing excessive long time-consuming laboratory and field researches.

Numbers like strength limit and Napier are Principal Fundamental Pivot Numbers (PFPN), which shows the equivalency and continuity in the entire world. Also, functions like e^x and e^{2x} are linear independents on every interval. Therefore, this number can be applying for setting the equivalency between factors of rocks slopes stability engineering. Euler's constant in fact derived from earth characteristics and overhand. Napier pivotal and fundamental number (e) reach science to finding the balance point between earth's features (in accordance to Eq. (1)) also, slope properties, average of rock mass index and variable strength limit of the slope, seismic acceleration and disturbance factor as an importance result which are obtained as follow:

$$e = ln\left(\frac{g_0}{X_{C-\text{earth}}}\right) = ln\left(\frac{1}{A.D}\right) = \sqrt{\frac{h_{\text{OS}} \times X_{\text{CVLN}} \times \left(H + D + \frac{\beta}{100}\right) \times (H + D)}{\overline{RMi} \times \text{FS}_{\text{fd}} \times \left[\frac{(\alpha + \beta + \delta)}{100} + \frac{(H_{IR} + H_{BF})}{100} + \frac{(W_{GB} + W_{SF,B} + W_{SB})}{100}\right]}$$
(42)

Strength limit show how properties of slopes, RMi, seismic acceleration and Napier number can play role parallel for safety factor estimation independently like as:

$$FS_{fd} = \frac{h_{OS}}{\overline{RMi} \times exp^{2} \times [\frac{(\alpha + \beta + \delta)}{100} + \frac{(H_{IR} + H_{BF})}{100} + \frac{(W_{GB} + W_{SP, B} + W_{SB})}{100}]} \times X_{CVLN} \times \left(H + D + \frac{\beta}{100}\right) \times (H + D)$$
(43)

Eq. (43) determined from results of Napier's number definition based on Eq. (42). Eq. (43) can use to check the reliability of designed slopes' stabilization. Unit equilibrium in Eq. (43) display as:

$$\frac{m}{MPa \times m} \times MPa = dimensionless$$

8.2. Calculation of the safety factor using impact of the unit weight of rock in assessing mine block

Strength limit of earth is a phenomenal parameter that enables us to compute essential factors with minimum data such as rock engineering properties. One of the applications of this factor is calculating the unit weight of rock in assessing mine's block (γ) as:

$$2 \times \gamma = \frac{X_{C-earth}}{g_0} \times \left[\frac{\left(H + D + \frac{\beta}{100}\right)}{n} + \frac{IRA}{100} \right] = \frac{1}{exp.^e} \times \left[\frac{100 \times \left(H + D + \frac{\beta}{100}\right) + \left(\ln x RA\right)}{100n} \right] = Area Density \times \left[\frac{\left(H + D + \frac{\beta}{100}\right)}{n} + \frac{IRA}{100} \right]$$
(44)

Hence, the unit weight of rock according to Eqs. (4) and (44) given by:

$$\gamma = \frac{\frac{1}{exp.^{e}} \times \left[\frac{\left(H + D + \frac{\beta}{100}\right)}{n} + \frac{IRA}{100} \right]}{2} = \frac{A.D \times \left[\frac{\left(H + D + \frac{\beta}{100}\right)}{n} + \frac{IRA}{100} \right]}{2}$$
(44.1)

$$4\gamma^{2} = (A.D)^{2} \times \left[\frac{\left(H + D + \frac{\beta}{100}\right)}{n} + \frac{IRA}{100} \right]^{2}$$
(44.2)

Eq. (44.2) defined in accordance to Eq. (44.1) and below procedures of Eq. (2):

$$\frac{1}{\exp^{2^e}} = \frac{1}{\exp^e \times \exp^e} = A.D^2$$
(45)

$$exp.^{2e} = \frac{1}{A.D^2}$$
 (45.1)

$$e = \frac{ln(\frac{1}{A.D^2})}{2}$$
(45.2)

$$e = ln(\frac{1}{A.D}) \tag{45.3}$$

$$exp.^{2} = \left(ln(\frac{1}{A.D})\right)^{2}$$
(45.4)

Hack et al. (1982), Hack et al. (1990), Cervantes (1995) and Hack (1998) imply that the behavior of a seismic wave in a rock mass and the relationships between the rock mass parameters and the seismic parameters not known in all details. Consequently, the interpretation is often ambiguous. Strength limit of earth, area density and gravity uncover the relationship between rock mass parameters like unit weight of rock and the seismic parameters such as seismic acceleration on a slope (Eqs. (44) and (44.1)). Inter ramp angle, average of rock mass tensile strength, the sum of rock mass index values obtained from every approved component of discovery boreholes in a slope, unit weight of rock mass and minimum and average of cohesion values of slope components in assessing mine's block prescribed other definitions of Euler's basis number as:

$$e = \frac{ln \left[\frac{h_{os}^{2} \times X_{CVLN} \times \left[\frac{(H+D+\frac{\beta}{100})}{n} + \frac{lRA}{100}\right]^{2} \times (H+D) \times \overline{T} \times 0.888}{2}\right]}{2}{ln \left[\frac{h_{os}^{2} \times X_{CVLN} \times \left[\frac{(H+D+\frac{\beta}{100})}{n} + \frac{lRA}{100}\right]^{2} \times (H+D) \times \overline{T} \times 0.888}{RMi \times FS_{id} \times RMi_{Sum} \times \overline{C}_{components of slopes}(c \circ f s) \times \overline{\emptyset}_{c \circ f s} \times C_{imin}}\right]} = \left[\frac{h_{os}^{2} \times X_{CVLN} \times \left[\frac{(H+D+\frac{\beta}{100})}{n} + \frac{lRA}{100}\right]^{2} \times (H+D) \times \overline{T} \times 0.888}{RMi \times FS_{id} \times RMi_{Sum} \times \overline{C}_{components of slopes}(c \circ f s) \times \overline{\emptyset}_{c \circ f s} \times C_{imin}}\right]}{2}\right]^{\frac{1}{2e}}$$

$$e = \frac{ln \left[\frac{h_{os}^{2} \times X_{CVLN} \times \frac{4\gamma^{2}}{AD^{2}} \times (H+D) \times \overline{T} \times 0.888}{RMi \times FS_{id} \times RMi_{Sum} \times \overline{C}_{components of slopes}(c \circ f s) \times \overline{\emptyset}_{c \circ f s} \times C_{imin}}\right]}{2} = \left[\frac{h_{os}^{2} \times X_{CVLN} \times \frac{4\gamma^{2}}{AD^{2}} \times (H+D) \times \overline{T} \times 0.888}{RMi \times FS_{id} \times RMi_{Sum} \times \overline{C}_{components of slopes}(c \circ f s) \times \overline{\emptyset}_{c \circ f s} \times C_{imin}}\right]^{\frac{1}{2e}}$$

$$(46.1)$$

Therefore, safety factor based on another key specification of Euler's determinant number with respect to slope stability properties can discovered from the following relations according to Eq. (46):

$$FS_{fd} = \frac{h_{OS}^{2}}{\overline{RMi} \times exp.^{2e} \times RMi_{Sum} \times \overline{C}_{components of slopes (c of s)} \times \overline{\emptyset}_{c of s} \times C_{imin}} \times X_{CVLN} \times \left[\frac{(H+D+\frac{p}{100})}{n} + \frac{IRA}{100}\right]^{2} \times (H+D) \times \overline{T} \times 0.888$$

$$(47)$$

Moreover, safety factor in accordance to Eqs. (2), (44.1), (46.1) and (47) given by:

$$FS_{id} = \frac{h_{OS}^{2}}{\overline{RMi} \times exp.^{2e} \times RMi_{Sum} \times \overline{C}_{components of slopes (c of s)} \times \overline{\emptyset}_{c of s} \times C_{imin}} \times X_{CVLN} \times \frac{4\gamma^{2}}{A.D^{2}} \times (H+D) \times \overline{T} \times 0.888$$
(47.1)

$$FS_{fd} = \frac{h_{OS}^{2}}{\overline{RMi} \times RMi_{Sum} \times \overline{C}_{components of slopes(c of s)} \times \overline{\emptyset}_{c of s} \times C_{imin}} \times X_{CVLN} \times 4\gamma^{2} \times (H+D) \times \overline{T} \times 0.888$$

$$(47.2)$$

The quantitative factor (0.888) in Eqs. (46), (46.1), (47), (47.1) and (47.2) determined by:

$$0.888 = \left(X_1 \times 10^5\right) + \left(\frac{\emptyset_{i\max}}{10^4}\right) \cong \left(X_1 \times 10^5\right) + \overline{V}$$
(47.3)

Other approximations based on Eqs. (47), (47.1) and (47.2) for safety factor determination given by:

$$FS_{tsl} = \frac{h_{cs}^2}{\overline{RMi} \times exp^{2e} \times RMi_{Sum} \times \overline{C}_{components of sloper(cof s)} \times \overline{\mathcal{Q}}_{cof s} \times C_{smin} \times \left(H + D + \frac{\beta}{100}\right)} \times X_{CVLN} \times \left[\frac{\left(H + D + \frac{\beta}{100}\right)}{n} + \frac{RA}{100}\right]^2 \times (H + D) \times \overline{T}$$
(48)

$$FS_{id} = \frac{h_{OS}^{2}}{\overline{RMi} \times e^{2\epsilon} \times RMi_{Sum} \times \overline{C}_{components of slopes(cofs)} \times \overline{\mathcal{O}}_{cofs} \times C_{imin} \times \left(H + D + \frac{\beta}{100}\right)} \times X_{CVLN} \times 4\gamma^{2} \times exp^{2\epsilon} \times (H + D) \times \overline{T}$$
(48.1)

$$FS_{id} = \frac{h_{cs}^{2}}{\overline{RMi} \times RMi_{Sum} \times \overline{C}_{components of slopes}(cof s) \times \overline{O}_{cof s} \times C_{imin} \times \left(H + D + \frac{\beta}{100}\right)} \times X_{CVLN} \times 4\gamma^{2} \times (H + D) \times \overline{T}$$
(48.2)

Design of pit slope of the section RS06 with this slope's specifications illustrated in Fig. 7. Number of slope components factor (n) which used in Eqs. (47) and (48) showed based on Fig. 7.



Figure 7. Pit slope design of north sector (section RS06) of Sungun copper project, scale 1:5000. 8.3. Empirical definition of safety factor via RMi classification system using limit equilibrium analysis approach

The simplex form of limit equilibrium analysis conventional methods only satisfied the equilibrium of the known forces. The sum of the known forces acting to persuade sliding of slope's parts compared with the sum of the identified forces available to resist against instabilities such as failure. The ratio of the total resisting actions to the sum of driving actions (ratio between two sums) defined as the factor of safety that present as Eq. (49):

$$FS = \frac{\sum (resisting \ actions)}{\sum (driving \ actions)}$$
(49)

Limit equilibrium can't become efficient if the slope failure occurs by complex mechanisms such as internal deformation, progressive deformation, brittle fracture, progressive creep, liquefaction of weaker soil layers, discontinuity orientation, progressive weathering, excavation disturbances, extensive internal disruption of slope mass, etc. [Adopted from 77]. In these conditions engineer trepan to more sophisticated numerical modelling techniques. Toward resolving these inadequacies, this simple definition of the factor of safety can interpreted in many modes, types, forms and ways. It could explain in terms of loads, forces, moments, works, action and reaction, etc. This concept of above definition also can use as a ratio of one influencing unit of total resisting forces to one unit of the total disturbing forces. The safety factor for fully drained condition given by below formulas:

$$FS_{fd} = \frac{reaction of resisting forces}{action of disturbing forces} = \frac{\frac{h_{OS}^{n-2}}{RMi_{Sum} \times \overline{\varnothing}_{c of s}} \times \left(\overline{C}_{components of slopes(c of s)}\right)^{2}}{1MN}$$
(50)

Eq. (50) indicate that what values of resisting forces react when 1 MN of disturbing forces act equivalently and simultaneously. Unit's equilibrium of Eq. (50) given as follows:

$$\frac{\frac{m^2}{MPa}(MPa)^2}{MN} = \frac{m^2 MPa}{MN} = \frac{m^2 \frac{MN}{m^2}}{MN} = \frac{MN}{MN} = Undimensioned$$

In addition, safety factor for fully drained condition with impact of the unit weight of rock in assessing mine's block and effective unit of strength given by below formula:

$$FS_{fd} = \frac{\left[\frac{n_{OS}}{RMi_{Sum} \times \overline{C}_{components of slopes(cof s) \times \overline{\mathcal{O}}_{cof s} \times 10^3}\right] \times \gamma_{ROCK}^2 \times MEARS}{1MN}$$
(51)

Equilibrium of units in Eq. (51) is as follows:

$$-\frac{\frac{m^{4}}{MPa \times MPa} \left(\frac{MN}{m^{3}}\right)^{2} \times MPa}{1MN} = \frac{\frac{m^{4}}{MPa \times MPa} \left(\frac{MN}{m^{2}} \times \frac{1}{m}\right)^{2} \times MPa}{1MN} = \frac{\frac{m^{4}}{MPa \times MPa} \left(MPa \times \frac{1}{m}\right)^{2} \times \frac{MN}{m^{2}}}{1MN} = \frac{MN}{MN} = dimensionless$$

8.4. Safety factor computing through disturbing work in a defined slope using limit equilibrium approach

This approach clarifies what values of resisting forces react based on 1 Mj disturbing work of disturbing forces (or what values of resisting works is carried out reactionary and responsively when 1 Mj of disturbing work is carried out). On the other words, this scrutiny shows what values of resisting works of resisting forces react when 1 Mj disturbing work of disturbing forces act in slope stability procedures. Thus, safety factor value obtained based on involved work approach in a slope stability action and reaction as:

$$FS_{fd} = \frac{reaction works value of resisting forces}{action works value of disturbing forces} = \frac{\frac{h_{OS}^{n}}{RMi_{Sum} \times \overline{C}_{components of slopes(cofs)} \times \overline{\emptyset}_{cofs} \times \left(H + D + \frac{\beta}{100}\right)^{n+1} \times X_{CVLN}^{n-2} \times \gamma_{ROCK}}{1Mj Disturbing work}$$
(52)
Equilibrium of units in Eq. (52) is as:

$$\frac{\frac{m^{4}}{MPa \times MPa} \times (MPa)^{2} \times \frac{MN}{m^{3}}}{1Mj} = \frac{MN \times m}{1Mj} = \frac{Mj}{Mj} = dimensionless$$

8.5. Rapid calculation of the safety factor through a strength reduction factor (by a simple approach in presence of minimum data via overall slope angle using seismic acceleration and disturbance factor)

Overall slope angle, seismic acceleration, disturbance factor and number of slope components can be help rapid computation of safety factor in emergency unseen cases as follows:

$$FS_{jd} = \left(H + D + \frac{\beta}{100}\right)^{n+2}$$
(53)

8.6. Calculating the RMi values for the assessing zone

RMI factor in the sections 8.1 and 8.2 is the acting RMI that rise from the interacting effects' intersecting of the RMI values of the slope's components. This value also obtained by averaging RMI values of the slope components. The RMI values for every component of the assessing zone calculated from correlation formulae between RMI and GSI, which proposed by G. Russo based on real field data and presented in Appendix 1.

In 2009 a new approach for a quantitative assessment of the Geological Strength Index (GSI) proposed by Russo. Based on the conceptual affinity of the GSI with the Joint Parameter (JP), which used in the RMi, (that introduced by Palmstrom in 1996). A relationship between the two indexes derived, exploited and proposed in order to obtain a reliable, quantitative assessment of the GSI by means of the basic input parameters for the determination of the RMi (i.e. the elementary block volume and the joint conditions). According to the RMi and GSI systems, Russo proposed below formulas [Adopted from 245]:

$$RMi: \ \sigma_{cm} = \sigma_c \times JP \tag{54}$$

$$GSI: \ \sigma_{cm} = \sigma_c \times s^a \tag{55}$$

Where s and a are the Hoek and Brown constants. Therefore, JP should be numerically equivalent to s^a and given that for undisturbed rock masses [245]:

$$s = exp. \left\lfloor (GSI - 100) / 9 \right\rfloor$$
(56)

$$a = (1/2) + (1/6) \times \left[exp.(-GSI/15) - exp.(-20/3) \right]$$
(57)

Then, a direct correlation between JP and GSI obtained from [245]: $\sum_{n=1}^{n} \frac{1}{(1-n)^{n}} \sum_{n=1}^{n} \frac{1}$

Therefore, we have the following relations according to Eqs. (42), (46), (54), (55), (56), (57) and (58):

$$RMi: \ \sigma_{cm} = \sigma_c \times JP = \sigma_c \times s^a = \sigma_c \times (exp.^{[(GSI-100)/9]})^{((1/2)+(1/6)\times[exp.^{[-GSI/15)}-exp.^{(-GSI/15)}-exp.^{(-GSI/15)}])}$$
(59)

$$e = ln\left(\frac{g_{0}}{X_{c-earth}}\right) = ln(\frac{1}{A.D}) = 1^{\frac{logS}{GSI-100}} = \sqrt{\frac{h_{OS} \times X_{CVLN} \times \left(H + D + \frac{\beta}{100}\right) \times (H + D)}{\overline{RMi} \times FS_{td} \times [\frac{(\alpha + \beta + \delta)}{100} + \frac{(H_{IR} + H_{BF})}{100} + \frac{(W_{GB} + W_{SP, B} + W_{SB})}{100}]}$$

$$= \sqrt{\frac{h_{OS}^{2} \times X_{CVLN} \times \left[\frac{(H + D + \frac{\beta}{100})}{RMi} + \frac{RA}{100}\right]^{2} \times (H + D) \times \overline{T} \times 0.888}}{\overline{RMi} \times FS_{td} \times RMi_{Sum} \times \overline{C}_{components of slopes(c of s)} \times \overline{\emptyset}_{c of s} \times C_{imin}}}}$$
(60)

Eq. (60) can use in reliability procedures of engineering decisions as a proof of relations in Section 8.1 and Section 8.2.

9. Validation of Recommended Formulas: Comparison of SLOPE/W Predictions and Experimental Formulations Results

All offered empirical equations can use to ensure the reliability of slope stability assessment by other methods. For this purpose, limit equilibrium SLOPE/W stability analysis of the slope carried out based on the section RS06 with contemplate various rock mass materials. FOS as a discriminant of stability limit help to determine suitable slope angles that slope consider actively fully drained in this study. The resulting FOS by SLOPE/W is high (1.97) indicating stable conditions for the slope stability analysis. Furthermore, safety factor values obtained from recommended equations in

this part are compared with acquired results from SLOPE/W software slope stability analysis (that was implemented by SRK consulting engineers and scientists group in 2008) for validation of presented formulas for the section RS06 of block five of Sungun copper mine with 37° overall slope angle.

Presence of water causes to create various instabilities conditions for slopes. This feature input and involves water-affecting factors in safety factor calculations. Effective fully drainage of the pit slopes will be the most critical parameter to ensure that the pit slopes remain stable. Block five of North sector (Section RS06) of Sungun Copper mine selected for the survey because this section is suitable for fully drained conditions.

Program SLOPE/W regulated and formularized in phrases of moment and force equilibrium of safety factor equations. Limit equilibrium methods used in this software includes Morgenstern-Price, General limit equilibrium, Spencer, Bishop, Ordinary, Janbu methods, etc. This program allows integration with other applications like finite element code [Adopted from 276]. Limit equilibrium analyses using the commercial software SLOPE/W carried out to reach follow scenario: • Slope fully drained to the toe. Standard deviation, percent of variance, variation coefficient, percent of variation coefficient, mean square error, root-mean-square error, correlation coefficient and percent of efficiency are used to comparison and validation of suggested empirical equations. The differences between obtained safety factor values through suggested equations correspond to the computational accuracy of the used parameters. Results of overall slope stability analysis of Sungun pit's north domain by SLOPE/W software demonstrate that the section RS06 is stable up to 37° overall slope angle.

Limit equilibrium SLOPE\W stability analysis model of overall pit slope on the section RS06 in block five of Sungun copper mine for fully drained condition by Janbu method and Mohr-Coulomb strength function with auto search of critical slip surface shown in Fig. 8. Table 5 presents the obtained FOS values from SLOPE/W software for the section RS06 of block five of Sungun copper mine for fully drained conditions and with β =37°. Rock mass input parameters for SLOPE/W stability analysis of the section RS06 in the selected block of Sungun copper mine reported in Table 6. The overall slope angles and slope configurations for north pit domain (section RS06) recommended as a result of SLOPE/W stability analysis, which is included in Table 7. Used factors in FOS calculations through recommended formulas (for the section RS06 in block five of Sungun copper mine) displayed in Table 8. Comparisons results between factors of safety values of recommended equations with SLOPE/W software stability analysis presents in Table 9. Statistical equations and factors that used in validation shown in Table 10.

In Table 6, must note that Sungun Porphyry, Dyke 1a, Dyke 1b and Faults are main geological units of the section RS06 (in its cross-section) [226]. Main geological units of the section RS06 of Sungun copper project in slope stability analysis cross-section SLOPE/W model of overall pit slopes for fully drained condition (which shown in Fig. 8) were colour coded. Sungun Porphyry (SP) was marked green. Dykes 1a (DK1a) were marked blue. Dykes 1b (DK1b) was marked yellow. Faults (FLT) were marked red. Skarn (SK) was marked aqua. Statistical measuring in Table 9 used to measure the differences between predicted values by suggested estimator formulas and the values actually observed by SLOPE/W software. The results of the SLOPE/W analysis for the north domain in the section RS06 for fully drained conditions are as Table 5:

Section	Pit Domain	OSA: Overall slope	Fully Drained Factor of Safety (FS)
		angle (°)	
			1.97
RS06	North	37	

Table 5: Factor of safety values obtained from SLOPE/W software.



Figure 8. Limit equilibrium Slope/W stability analysis model of overall pit slope for section RS06 in fully drained condition with Janbu method.

Domain / Lithology		Dyke 1a	Dyke 1b	Sungun Porphyry	Fault
		(DK1a)	(DK1b)	(SP: POT [®])	(FLT)
Mean uniaxial compressive		105	130	82	30
strength of the intact rock					
material, pieces and					
elements, UCS mean, $\overline{s\iota gc\iota}$,					
$\overline{\sigma_{c\iota}}$ (MPa)					
Geological strength index mean,		42	50	40	24
GSI Mean	Section				
Intact rock parameter: A	RS06	23	20	18	18
material constant of the intact					
rock, mi (Dimensionless)					
Disturbance Factor, D		0.7	0.7	0.7	0.7
Unit weight of rock, γ (MN/m ³)		0.024	0.024	0.023	0.023
Slope Height (m)		150	150	150	150
Cohesion, C (kPa)		931	1198	727	319
Friction angle, Ø - Phi (°)		44	48	39	24
Rock mass tensile strength, Sig_t -		-0.025	-0.072	-0.025	-0.004
σ_t (MPa)					
Uniaxial compressive strength		13.09	19.00	9.03	2.54
of the rock mass, Sig_{cm} . σ_{cm}					
(MPa)					
Rock mass young's modulus,		2743	5578	2142.6	512
E_{rm} (MPa)					

 Table 6: Rock mass input parameters for limit equilibrium SLOPE/W stability analysis.

^b Potassic altered (potassic feldspar, quartz, secondary biotite). The rock mass at Sungun strongly affected by hydrothermal alteration. Potassic alteration is of considerable volume and importance to affect significantly the rock mass strength [80].

Note: The strength of intact rock obtained through statistical analysis of uniaxial compressive strength data is derived from rock core's laboratory testing and simple field strength estimates. The intact rock strength refers to the strength of a finite piece of rock, free of any defects, such as veins, cemented or un-cemented joints, micro fissures, or faults.

This intact rock strength usually derived from testing the unconfined compressive strength (UCS) of hand specimen or drill core samples. Intact rock strength, for which average values have been calculated using field estimate backed up by a comprehensive laboratory testing programme of UCS and point load tests [281].

Table 7: Recommended overall slope angles and slope configurations for fully drained slope in section RS06

K30	0.
Pit Domain	North
Section	RS06
Max. Overall Slope Height (m)	300
Overall Slope Angle (°) - crest to	37
toe	
Inter Ramp Height (m)	100
Inter Ramp Slope Angle (°) crest	45
to toe	
Bench Face Angle	65
Geotechnical Berm Width (m)	50
Spill Berm Width (m)	24
Safety Berm Width (m)	14
Bench Height (m)	25

Table 8: Input factors of slope stability analysis for suggested computational formulas in section

DC
KO

	RS06.		
Factors	For Section RS06	Factors	For Section RS06
RMI	3.122937975 Mpa	Н	0.05
RMI _{sum}	740.1363 Mpa	D	0.7
h _{OS} =OSH	300 m	e	2.718281828
n	4 components	$\bar{T} = \bar{\sigma}_t$	0. 007 Mpa
\overline{C}_i	793.75 kPa	<i>X</i> ₁	0.000008832
C _{imax}	1198 kPa	$\alpha = IRA$	45 [°]
C _{imin}	319 kPa	$\delta = BFA$	65 [°]
$\overline{\emptyset}_i$	38.75°	$\beta = OSA$	37°
Ø _{imax}	48°	H _{IR}	100 m
Ø _{imin}	24°	H_{BF}	25 m
$\overline{\gamma}_{rock}$	0.0235 MN/m^3	W_{GB}	50 m
$\overline{V} = \frac{\overline{RM_1}}{}$		W _{SP.B}	24 m
RMI _{Sum}	$0.00421940928312798602095316768006 \cong 0.0042$		
$X_{C-Variable} = X_{CVLN}$	0.65133593217841457973135724564648 Mpa	W _{SB}	14 m

Table 9: Comparison and validation results between two calculated safety factors (slope stability analysis by new empirical formulas using RMi system and SLOPE/W stability assessment by Janbu method) for section RS06 with

	$\beta = 37$.											
Secti	Equatio	FS	FS	FS mean	Standa	Variance	Variation	Percent	Mean	Root	Coefficien	Efficien
on	ns	obtained	fro	(\overline{FS})	rd	(VAF or	coefficient	of	Square	Mean	t Of	cy %
		from	m		Deviati	Var)	(V)	Variatio	Error	Square	Correlatio	(Ē)
		suggested	Slop		on (SD			n	(MSE)	Error	\mathbf{n} (\mathbf{R}^2)	
		equations	e/w		or S)			coefficien	(112022)	(RMSE)		
		equations	C/ 11		01 5)			t (V%)		(ILVIOL)		
	Eq. (43)	1 9758354	1 97	1 972917	0.00291	8 51E-06	0.0014789	0.15%	3.41E-05	0.0058354	0 9941646	99.42%
	Lq. (43)	1.7750554	1.77	7	0.00271	0.51L-00	0.0014705	0.1570	5.41L-05	0.0050554	0.7741040	JJ. 4 270
				1	11							
	Eq. (47)	1.9701806	1.97	1.970090	9.03E-	8.16E-09	4.58E-05	0.0046%	3.26E-08	0.0001806	0.9998194	99.98%
-				3	05							
RS06	Eq. (48)	1.980957	1.97	1.975478	0.00547	3.00E-05	0.0027733	0.28%	1.20E-04	0.010957	0.989043	98.90%
	1 . ,			5	85							

Eq. (50)	1.9770906	1.97	1.973545	0.00354	1.26E-05	0.0017964	0.18%	5.03E-05	0.0070906	0.9929094	99.29%
			3	53							
Eq. (51)	1.9649595	1.97	1.967479	0.00252	6.35E-06	0.001281	0.13%	2.54E-05	0.0050405	0.9949595	99.50%
			7	03							
Eq. (52)	1.9706294	1.97	1.970314	3.15E-	9.90E-08	1.60E-04	0.02%	3.96E-07	0.0006294	0.9993706	99.94%
			7	04							
Eq. (53)	1.9738227	1.97	1.971911	0.00191	3.653E-06	9.69E-04	0.10%	1.461E-	0.0038227	0.9961773	99.62%
			3	13				05			

Mohammad Salehi Alashti.; East African Scholars Multidiscip Bull; Vol-1, Iss-3 (Dec, 2018): 61-121

Table 10: Used statistical equations and factors in validation.

Statistic parameters	Statistic formulas
Mean factor of safety	$\overline{FS} = \frac{1}{n} \sum_{i=1}^{n} FS_i$
Standard Deviation = $SD = S$	Standard Deviation = $\sqrt{Variance} = \sqrt{\overline{FS^2} - \overline{FS^2}}$
Variance = $VAF = Var = S^2$	$Variance = \overline{FS^2} - \overline{FS}^2$
Variation coefficient = Coefficient of variation = V = CV	Variation coefficient $= \frac{SD}{FS}$
Percent of variation coefficient = V%	Percent of variation coefficient $=\frac{SD}{FS} \times 100$
Mean Square Error = MSE	$MSE = \frac{1}{n} \sum_{i=1}^{n} \left(FS_{Suggested} - FS_{SLOPEW} \right)^2 $
Root Mean Square Error = RMSE	$RMSE = \sqrt{\frac{1}{n} \sum_{i=1}^{n} (FS_{Suggested} - FS_{SLOPEW})^2} $
Correlation Coefficient $=$ R ²	Correlation Coefficient = $1 - RMSE^{d}$
Efficiency = E	Efficiency = [Correlation Coefficient] \times 100 °

^e $FS_{suggested}$ is the factor of safety predicted by suggested empirical equations and FS_{SLOPEW} is the calculated (observed) safety factor by SLOPE/W software.

^d This is the correlation coefficient in accordance to root mean square error.

^eThis is the real efficiency which result by correlation coefficient.

Variance is a good criterion to show the dispersion and variability of calculated safety factor through proposed relations and SLOPE/W results in comparison with mean safety factor. In Fig. 9, minimum variances shown that demonstrates all safety factors are close to mean FOS. The mean square error (MSE) of the calculator relations evaluates the average of the squares of the "errors", which is the difference between the estimator formulas and SLOPE/W assessment results. Fig. 9 has shown a risk function that related to the anticipated value of the squared error loss or quadratic loss. Less mean square error or deviation values illustrate more accurate slope stability assessment because these empirical relations account the information that could produce a more efficient and precise estimation. Root-mean-square deviation (RMSD) is the second instant of the error between the supposed formulas and SLOPE/W estimation results which incorporates both the variance of the estimator formulas, SLOPE/W results and their bias.



Figure 8. Limit equilibrium Slope/W stability analysis model of overall pit slope for section RS06 in fully drained condition with Janbu method.

In Fig. 10, less root mean square error as a quantitative comparison demonstrates a good measure of accuracy of presented formulas. RMSE express the typical standard deviation of the differences between predicted values by relations and computed values by SLOPE/W. Standard deviation is a measure that is used to quantify the amount of variation or dispersion of computed FOS values by empirical relations and SLOPE/W software. Obtained standard deviations is close to zero, which indicates the computed safety factors through equations and SLOPE/W tend to be very close to the mean FOS values that are shown in Fig. 10.



Figure 10. Scatter curves for the standard deviation, variation coefficient and root mean square error according to comparison between suggested empirical relations and SLOPE/W results with that listed in Table 8.

The coefficient of variation (CV) which is also known as unitized risk is a standardized measure of dispersion of safety factors computed by relations and SLOPE/W results. Less percent of variation coefficient that defines low relative standard deviation (RSD) proves these relations are qualified for safety factor calculation, which displayed in Figs. 11(a) and 11(b).

Comparison between SLOPE/W results and mentioned equations illustrate less variation coefficients percent of them as Figs. 11(a) and 11(b) that presents high efficiency of offered formulas as shows as Fig. 12.

Efficiency is often determined the measure of desirability of suggested empirical relations. It measures empirical estimator formulas' optimality in slope stability procedures. More efficient estimator formulas of safety factor achieve a given performance of slope stability analysis.



Figure 11. Percent of variation coefficient stacked curve (A) and filled radar chart (B) according to comparison between suggested empirical relations and SLOPE/W result, which listed in Table 8.



Figure 12. Efficiency chart based on comparison between suggested empirical relations and SLOPE/W result that listed in Table 8.

The strength of association between proposed relations and SLOPE/W analysis result, which defined as correlation coefficient shown in Fig. 13. This curve indicates large amounts of correlation between new relations and SLOE/W results. All correlation coefficients are close to one, it would indicate that the relations are positively related and the scatter plot falls almost with a positive slope.



Figure 13. Correlation coefficients scatter curve and column charts based on comparison between suggested empirical relations and SLOPE/W result, which listed in Table 8.

High correlation between percent of variation coefficient and correlation coefficient with linear, logarithmic and polynomial trend line (regression) in Fig. 14 shows very high accuracy of recommended relations in calculations.





Figure 14. Correlation between percent of variation and correlation coefficients in accordance to comparison between suggested empirical relations and SLOPE/W result, which listed in Table 8, A. Linear regression, B. Polynomial regression, C. logarithmic regression.

High correlation coefficient and low CV indicate very negligible unitized risk of supposed relations and high reliability of obtained results that represented in Figs. 13 and 15.



Figure 15. 3-D column chart of variation coefficients based on comparison between suggested empirical relations and SLOPE/W result, which listed in Table 8.

10. Discussion of Results

In order to increase investments in mines' fields also heighten the open pit mines depth, whereby mines extraction tends to grow the factors of safety with a principle, effective and rational reason that is slopes stability problems intensification. Slope stability analysis through empirical relations require to quick analysis and primary estimates in different stages of design particular in primary design, design revision, modification, adjustment and optimization even encounter with unforeseen cases and emergency conditions. Besides, the deficiency of different classification systems in mentioned situations, it requires implements complete and quick analysis of slopes' safety factor through basis earth's strength limit specifications which covers all conditions or at least assess all-important and affecting conditions and items in calculations without any limitations toward to reduce error possibility. Moreover, the intended significant central importance effects of the earth strength limit functional factor's applications besides the covering some of the most important ground parameters therefore, using the RMi classification system in empirical safety factor calculator formulas lead to better computations for different reasons. Such as considering of the three-dimensional block volume in RMi determination that will universally improve the description of the features of the rock mass.

Strength limit of earth shows how fundamental factors of nature such as gravity also rock mass classification systems like RMi, impact on scrutinizing slopes stability assessments. This elementary factor and Napier's constant (or Euler's constant) uncovers the relationship between fundamental physics, geo-mechanics and rock slope engineering also other sciences too (For more information refers to: Salehi Alashti, 2017).

The slope stability against failure risk represented by safety factor, which is determined using recommended relations and limit equilibrium method by SLOPE/W software. It is found that the factor of safety in selected area is higher than 1.9 when encounters with fully drained condition and seismic acceleration effects, which also displays slope is still stable. The minimum value obtained for FOS is about 1.96748. The following discussions derived from the presented results:

- 1) The stability of the section RS06 is acceptable because of the minimum factor of safety is more than 1.95.
- Analysis the stability of rock slopes by these procedural equations' method provides detailed information and an independent approach for determining the safety factor of the selected section even based on both total and effective driving actions.
- 3) The case study that presented above illustrates that application of these relational processes made to reduce the risk of failure.
- 4) These easily calculable relations can use in uncertainties occurrence moments even help predict the appearance of them and uncover the silent risks.

5) The equations give some precise results for factors of safety calculation in comparison with the SLOPE/W limit equilibrium method predictably.

11. Summary and Conclusions

In this paper, previous slope stability investigations and approaches reviewed based on different methods, especially existing rock mass classification systems. This found mentioned methods contain various limitations and lacks evolutions. Therefore, these systematic procedures could improve through substantial factors like strength limit of earth, which can be including all aspects. In addition, better-improved techniques needed for slope stability evaluation via rock mass classification systems. In the direction of reaching presented purposes, some definitions of the earth's strength limit substantial factor described in detail also discussed precisely. Then, new empirical relations suggested by contemplating the beneficial roles of RMi classification system based on strength limit of earth fundamental factor which envisage most principal effectors to decrease errors and limitations for analysis of slope stability accurately, quickly, safely and efficiently, especially in un-seeming, un-detected and unforeseen conditions. This could influence slope stability engineering, stimulation, design and modelling also leads to entirely different approaches for rock engineering design particular rock slope engineering. Offered relations can apply in reliability analysis of slopes designs projects.

The explanation about cognizing the seismic wave's behaviour in rock mass and the connections between the rock mass parameters and the seismic factors are often vague. Strength limit, area density and gravity of earth help discovers the relations between the unit weight of rock as rock mass parameters and the seismic acceleration as seismic parameters on a slope.

Less percent of variance in recommended equations show that the results of calculated safety factors estimate with high precision. Low values of variation coefficient, root-mean-square error, standard deviation and variance illustrate errors reduction, which cause bettering safety factor calculations. High value of correlation coefficient tend to one indicates high accuracy and solidarity of these formulas. Low values of mean square error, percent of variance, standard deviation and percent of variation coefficient also high efficiency and correlation coefficient, represents that obtained factor of safety from recommended equations have high correlation and very low dispersion in its comparison with results of SLOPE/W software's limit equilibrium stability analysis. Therefore, engineers can be relying on the presented formulas, which has a high reliability degree.

Recommended formulas for safety factor calculation in fully drained conditions on section RS06 in block five of Sungun copper mine with 37° overall slope angle can apply to assess the reliability of the obtained factors of safety by other various methods. These applications have some benefits as follows:

(1) Specialist knowledge of rock engineering not required. However, as a minimum, the inspectors would need to be mine engineers or and engineering geologists or geotechnical engineers. (2) Quick (time saver) — typically takes no more times for calculations. (3) Cost effectiveness. (4) Safer. (5) Labor-saving. (6) Simple. (7) More environmentally friendly. (8) These applications have many practical advantages and includes precise approaches. (9) Easy for engineering geologists, mine engineer and geotechnical engineers to understand — no need for external training course, all necessary training information is available online and an engineering geologist or mine engineer or geotechnical engineer would be able to train themselves via self-directed learning. (10) Widely adopted. (11) Designed to accommodate both harder and weaker rock slopes. (12) The used RMi system recommended by Palmström also cited in other publications. (13) Using of RMi classification system cover most important ground parameters, which lead to better estimates. (14) Research into the system is ongoing which demonstrates its wide acceptance, viability and potential for long-term improvement. (15) This process does not require any equipment. (16) The process has been refined over many times to its present level of development. (17) This process does not require assumptions.

Finally, and briefly, can note that:

- 1. An empirical formulated procedure originally developed based on the fundamental parameter to assess (assessing) the stability of rock slopes such as mine's walls that has described to be capable of generating feasible rapid independent solutions for instability problems of rock slopes.
- 2. Importance of RMi introduces through earth's strength limit factor in rock slope engineering with precise simple equations.
- 3. A significant advantage of the suggested limit equilibrium empirical relations and independent equations compared with the analysis of SLOPE/W software, which have commonly applied for rock slope stability's problems that it is generally applicable, and will not require any assumptions to be made even about how forces act in the system.
- 4. The new ascertainable analytic formulas have discovered to provide suitable, acceptable, applicable, rational, albeit somewhat cautious safely estimates of the rock slopes stability.

5. New relations determine procedures that can provide solutions for fully drained conditions in a particular assessment mechanism of rock slope stability that have illustrated. When this process used in combination with other proposed valid methods and analysis procedures, these makes solutions that prosperously cover and bracket experimental data involved in design.

Strength Limit factor arise and enlarge considerable interests in various areas of sciences. Also, will using as a basis in new scientific discoveries. Moreover, this key substantial number applying for rock slopes engineering developments. Furthermore, this primordial fundament will be using as a bridge between fundamental applied physics and practical applications in the fields of geological engineering, geo-mechanics, rock mechanics, rock engineering, mining engineering, environmental hazards assessments and earth sciences.

Generally, application of strength limit of earth fundamental factor and RMi system in safety factor determination implies the unity, correlation, continuity and dependency all of the existence parts (like the universe and earth).

Highlights

- Earth's strength limit (ESL) and RMi open new gates between fundamental physics and Earth sciences.
- Earth's strength limit indicates the relationship between FOS and RMi.
- Strength Limit presents a rigorous mathematical approach to study slope's problems.
- RMi provides slope stability analysis accurately, quickly, safely and efficiently.
- Strength limit of earth (SLE) displays determinant role of RMi in scrutinizing slope stability assessment.

Acronyms, Nomenclature, Abbreviations and Notations

Symbols: Meanings, SI Units;

 $X_{c-earth}$: Strength limit number of earth, MPa & % & Dimensionless;

ESL: Earth's strength limit, MPa & % & Dimensionless;

SLE: Strength limit of earth, MPa & % & Dimensionless;

X_{C-Earth,Old}: Strength limit parameter of the earth according to old standard gravity, MPa & % & Dimensionless;

 $X_{C-Earth,Now}$: Strength limit parameter of earth or strength of existence global gravitation according to present standard gravity, MPa & % & Dimensionless;

 $X_{CV-Slope}$ or $X_{CV(Slope)}$ or $X_{CVLN-Slope}$: Cohesion variable limit number for slope, MPa;

X_{CCLN}: Cohesion constant limit number, MPa;

 g_0 : Standard gravitational acceleration or Earth's standard surface gravity, m/s²;

 $\mathbf{g}_{\mathbf{0}-\mathbf{0}\mathbf{ld}}$: Old standard gravity, m/s²;

 g_{0-Now} : Present value of standard gravity, m/s²;

AOE_{old}: Age of the earth according to old standard gravity, years & Billion years;

AOE_{Exact}: Accurate or exact age of the earth according to present standard gravity at present time, years & Billion years; **RMi**: Rock Mass index, MPa;

FS or FOS or SF or SOF: safety factor;

MEARS: Minimal effective action and reaction unit of strength, MPa;

D. **C**: Decay constant of gravitational particles;

b: Strength variations coefficient in 10^{10} year, $\frac{MPa}{Year} = \frac{MN}{m^2 Year} = \frac{Ggr}{Year m s^2}$;

 $\overline{V}_{Assessing Mine Block}$ or \overline{V} or \overline{V}_{MB} : Rock mass index values average to sum of the rock mass index values of the assessing mine block ratio, Dimensionless;

X₁: Detector parameter, Dimensionless;

FS mean (\overline{FS}): Mean factor of safety, Dimensionless;

FS_{fd}; Factor of safety for fully drained conditions, Dimensionless;

OSH or **H**_{0S} or **h**_{0S}: Overall slope height or is the maximum overall slope height, Meter (m);

n: Number of components in a slope (number of slope's components), Dimensionless;

RMI or **RMI**_(av): Average of rock mass index values for assessing mine block, MPa;

RMi_{Sum}: Sum of rock mass index values obtained from every approved components of a discovery boreholes in a slope for assessing mine block, MPa;

e: Napier number, Dimensionless;

H: Seismic acceleration, Dimensionless;

D: Disturbance factor, Dimensionless;

OSA or β : Overall slope angle, Degree;

IRA or α : Inter ramp angle, Degree;

BFA or Δ or δ : Bench face angle, Degree;

BFH or **H**_{**BF**}: Bench face height, Meter (m); **SP.BW** or **W**_{SP.B}: Spill berm width, Meter (m); **SBW** or **W**_{SB}: Safety berm width, Meter (m); **IRH** or **H**_{IR}: Inter ramp height, Meter (m); **GBW** or **W**_{**GB**}: Geotechnical Berm Width, Meter (m); C_{components of slopes (c of s)}: Average of cohesion values of slope components (average of a slope components cohesion values), MPa; **C**_{imin}:Minimum cohesion value of slope components, kPa; $\overline{\phi}_{c \text{ of } s}$: Average of internal friction angle values of slope components, Degree; $\overline{\mathbf{T}}$: Average of rock mass tensile strength in assessing mine block, MPa; γ : Unit weight of rock in assessing mine block, ΜN. $\overline{m^3};$ **X**_{CVLN}: Cohesion variable limit number, MPa; C: Cohesion, MPa; **Phi**: Friction angle, Degree; $\mathbf{g_h}$: Gravity at height h above sea level, m/s²; $\mathbf{g_0}$: Standard gravity, m/s²; **r**_e: Radius of earth; **h**: Height from sea level; **F**: Force between the masses, N; G: Gravitational constant or Newtonian gravitational constant or Newtonian constant of gravitation, m³kg⁻¹s⁻² & N $m^2 kg^{-2};$ \mathcal{M}_{\oplus} or E: Mass of the earth, Kg; M: Every point mass; **R**: Distance between the masses, Meter (m); **ED**: Earth density, Kg/m³ & gr/cm³; V_{earth} : Volume of earth, m³ & km³ & cm³ & cu mi; **A.D or** D_A **or S.D or** D_S : Area density, Ggr/m²; R'or R_{Earth-Dimensionless} or Dimensionless R: Ratio of every point mass M to mass m or Quantitative radius value of earth without unit, Dimensionless; $R'': m^2;$ **R'_{Earth}:** m²; \mathbf{g}_{V} : Variable standard gravity, m/s²; **E**_{*V*-*Earth***:**} Variable mass of the earth, Kg; $\mathbf{R}_{V-Earth}$: Variable distance between the masses, m; **ED**_{*V*-*Earth*}: Variable earth density, Kg/m³; $V_{V-Earth}$: Variable volume of earth, m³; $\mathbf{G}_{\mathbf{V}}$: Gravitational variation, m³kg⁻¹s⁻² & N m² kg⁻²; **AD**_V: Variable area density of earth, Ggr/m^2 ; **CV**: Coefficient of variation; MSE: Mean square error; **RMSD**: Root-mean-square deviation; **SD** or **S**: Standard Deviation; **VAF** or **Var** or S^2 : Variance; V or CV: Variation coefficient (Coefficient of variation); V%: Percent of variation coefficient; **R**²: Correlation Coefficient; E: Efficiency; NICICO: National Iranian Copper Industries Company; PGA: Peak ground acceleration; MCL: Maximum credible level; MDL: Maximum design level: **DBL**: Design basis level; CL: Construction Level; **PFPN**: Principle Fundamental Pivot Numbers; LEM: Limit equilibrium methods. Declarations

Interests

The author declares that he (author) is the 'owner' of all interests of this article forever.

Conflict of Interest

The author confirms that there are no known conflicts and competing of interest related to this publication.

Author's Descriptions

Mohammad Salehi Alashti (M.S.A) had the idea for the project, designed, performed and interpreted the research study. M.S.A analysed the data, drafted and wrote the paper also reviewed the manuscript. The author read and approved the final manuscript.

Author's Contributions Statement

No one.

Funding Statement

No financial support has been used for this work.

Supports Statement

No supports used for implementation of this research study.

License Statement

Open Access. This article published under the terms of the Author's open access private license.

References

- 1. Abbaszadeh Mohsen. Uncertainty and reliability analysis applied to slope stability: A Case Study From Sungun Copper Mine. Geotechnical and Geological Engineering 05/06/2011.
- 2. Abramson LW, Lee TS, Sharma S, Boyce GM. Slope Stability and Stabilization Methods, New York: John Wiley & Sons, 2002.
- 3. Adhikary DP, Dyskin AV, Jewell R.J. Modelling of flexural toppling failures of rock slopes. The University of Western Australia, 1994.
- 4. Adhikary DP, Dyskin AV, Jewell RJ. Analysis of fracture and deformation processes during flexural toppling in foliated rock slopes. The University of Western Australia, 1995.
- Adikari GSN, Donald IB, Parkin AK. Analysis of the construction behaviour of Darmouth dam. In: Proceedings of the Fourth International Conference on Numerical Methods in Geomechanics, 1982, Edmonton, Canada, Vol. 2. 1982.
- 6. Adhikary DP, Dyskin AV. Modelling of progressive and instantaneous failures of foliated rock slopes. Rock Mechanics and Rock Engineering 2007;40(4):49-362.
- 7. Albataineh Nermeen. Slope Stability Analysis Using 2D and 3D Methods. Master of Science Thesis, The Graduate Faculty of The University of Akron, August, 2006. [Dissertation]
- 8. Anagnosti P. Three dimensional stability of fill dams. In: Proceedings of the 7th International Conference on Soil Mechanics and Foundation Engineering, 1969, Mexico, Vol.2, 1969:275-280.
- Azzouz AS, Baligh MM. Design charts for three-dimensional stability of cohesive slopes subjected to surcharge loads. National Technical Information Service, U.S. Dept. Of Commerce, publication No. PB-275779, November 1976.
- 10. Azzouz AS, Baligh MM. Three-dimensional stability of slopes, National Technical Information Service, U.S. Dept. Of Commerce, Publication No. PB-285740, June 1978.
- 11. Azzouz AS, Baligh MM, Ladd CC. Three-dimensional stability analysis of four embankment failures. In: Proceedings of the 10th International Conference on Soil Mechanics and Foundation Engineering, 1981, Stockholm, Sweden, Vol. 3, 1981:343-346.
- 12. Azzouz AS, Baligh MM. Loaded areas on cohesive slopes. ASCE, Journal of the Geotechnical Engineering Division 1983;109(GT 5):726-729.
- 13. Baker R. Determination of the critical slip surface in slope stability computations. Int. J. Numer. Analyt. Meth. Geomech. 1980;4(4):333–359.
- 14. Baker R, Gaber M. Theoretical analysis of the stability of slopes. Geotechnique 1978;28(4):395–411.
- 15. Baker R, Leshchinsky D. Stability analysis of conical heaps. JSSMFE, Soils and Foundations December 1987;27(4):99-110.
- 16. Baker R, Shukha R, Operstein V, Frydman S. Stability charts for pseudo-static slope stability analysis. Soils dynamics and earthquake engineering 2006;26:813-823.

- 17. Baligh M, Azzouz AS. End Effects of stability of cohesive slopes. ASCE, Journal of the Geotechnical Engineering Division 1975;101(GT11):1105-1117.
- 18. Baligh MM, Azzouz AS, Ladd CC. Line loads on cohesive slopes, 1977.
- 19. Barton NR. Recent experiences with the Q-system of tunnel support design. In: Bieniawski, editor. Pro. Symp. On Exploration for Rock Engineering, Johannesburg, 1976, Rotterdam: publ. Balkema, 1976:107-117.
- Barton NR. Rock mass classification and tunnel reinforcement selection using the Q-system. In: Kirkaldie Louis, editor. Proc. Symp. Rock Classification Systems for Engineering Purposes, 1988, ASTM Special Technical Publication 984. Philadelphia: American Society for Testing and Materials, 1988:59-88.
- 21. Barton N. Some new Q-value correlations to assist in site characterization and tunnel design. International Journal of Rock Mechanics and Mining Sciences (Int. J. Rock Mech. & Min. Sci.) 2002;39:185-216.
- 22. Barton N, Grimstad E. The Q-system following twenty years of application in NMT support selection. Felsbau 1994;12(6):428-436.
- 23. Barton N. TBM performance estimation in rock using QTBM. Tunnels & Tunnelling September 1999:30-34.
- 24. Barton N, Grimstad E. The Q-system following thirty years of development and application in tunneling projects. In: Proc. ISRM Symp. EUROCK 2004, Salzburg, Austria, 2004:15-18.
- 25. Barton N, Lien R, Lunde J. Analysis of rock mass quality and support practice in tunnelling and a guide to estimating support requirements. Internal Report No. 106. Oslo, Norway: Norwegian Geotechnical Institute; 1974.
- 26. Barton NR, Lien R, Lunde J. Engineering classification of rock masses for the design of tunnel support. Rock Mechanics and Rock Engineering 1974;6(4):189–236, publ. Springer, Verlag.
- Barton N, Loset F, Lien R, Lunde J. Application of Q-system in design decisions concerning dimensions and appropriate support for underground installations. In: Bergman M, editor. Subsurface Space: Environmental Protection, Low Cost Storage and Energy Savings. Proceedings of the International Symposium Rock-store' 80, 1980, vol. 2. Oxford, UK: Pergamon Press, 1980:553-561.
- 28. Bessason B, Kaynia AM. Site amplification in lava rock on soft sediments. Soils dynamics and earthquake engineering 2002;22:525-540.
- 29. Bieniawski ZT. Engineering classification of jointed rock masses. Transaction of the South African Institute of Civil Engineers Dec. 1973;15(12):335-344.
- 30. Bieniawski ZT. Geomechanics classification of rock masses and its application in failure tunnelling. In: Proc. of the 3rd Int. Congress on Rock Mech., 1974, Denver, 1974:27–32.
- 31. Bieniawski ZT. Rock mechanics design in mining and tunneling. Rotterdam: A.A. Balkema, 1984:272 p.
- 32. Bieniawski ZT. Rock mass classifications in rock engineering. In: Bieniawski ZT, editor. Proceedings Symposium on Exploration for Rock Engineering, 1976, Rotterdam: A.A. Balkema, 1976:97-106.
- 33. Bierntowski ZT. European Conference on Soil Mechanics and Foundation Engineering, 1976, Vienna, Vol 1., 1976:1-1.
- 34. Bieniawski ZT. The geomechanics classification in rock engineering applications. In: Proceedings of the 4th international congress on rock mechanics (Proc. 4th. Congr. Int. Soc. Rock Mech.), 1979. Montreux, Vol 2, 1979:41-48.
- 35. Bieniawski ZT. Rock mass classification as a design aid in tunnelling. Tunnels & Tunnelling July 1988.
- Bieniawski ZT. Engineering Rock Mass Classifications: A Complete Manual for Engineers and Geologists in Mining, Civil and Petroleum Engineering. New York: John Wiley & Sons, Wiley-Interscience, 1989:205-219, 251 & 272, ISBN: 978-0-471-60172-2.
- Bieniawski von Preinl ZT, Celada Tamames B, M. Galera Fernández J, Hernández Álvarez Mario. Rock mass excavability (RME) indicator: New way to selecting the optimum tunnel construction method. Tunnelling and Underground Space Technology - TUNN UNDERGR SPACE TECHNOL. 2006;21:237-237. Doi: 10.1016/j.tust.2005.12.016.
- 38. Bishop AW. The use of pore pressure coefficients in practice. Geotechnique 1954;4:148–152.
- 39. Bishop AW. The use of slip circles in stability analysis of slopes. Géotechnique 1955;5:7-17.
- 40. Brown ET. editor. ISRM suggested methods, Rock characterization testing and monitoring, Oxford, UK: Pergamon Press; 1981.
- Brown DJ, Denby B, Singh RN. An expert system for slope stability assessment: Part II. International Journal of Surface Mining, Reclamation and Environment 1988;2(2):93-98, Taylor and Francis, Published online: 26 Apr 2007, <u>http://dx.doi.org/10.1080/09208118808944141</u>.
- 42. Brown DJ, Singh RN. An expert system for slope stability assessment: Part I. International Journal of Surface Mining, Reclamation and Environment 1987;1(3):173-178, Taylor and Francis, Published online: 25 Apr 2007, http://dx.doi.org/10.1080/09208118708944116.
- 43. Buhan PD, Freard J, Garnier D, Maghous S. Failure properties of fractured rock masses as anisotropic homogenized media. Journal of Engineering Mechanics 2002;128(8):869-875.

- 44. California Division of Mines and Geology. Guidelines for evaluating and mitigating seismic hazards in California. DMG special publication 117, 1997.
- 45. Cavounidis S. On the ratio of factor of safety in slope stability analyses. Geotechnique 1987;37(2):207-210.
- 46. Cavounidis S, Kalogeropoulos H. End effects on the stability of cuts in normally consolidated clays. Rivista Italiana Gi Geotechnica 1992;2:85-93.
- 47. Celestino TB, Duncan JM. Simplified search for non-circular slip surfaces. In: Proc., 10th Int. Conf. SMFE, 1981, Stockholm, Sweden, 1981:391–394.
- 48. Chen RH. Three-dimensional slope stability analysis. Joint Highway Research Project, Report JHRP-81-17, West Lafayette, IN, USA: Eng. Experiment station, Purdue University; 1981.
- 49. Chen RH, Chameau JL. Three-dimensional limit equilibrium analysis of slopes. Geotechnique 1982, 1983a;32(1):31-40.
- 50. Chen R, Chameau J. Discussion Three-dimensional Limit Equilibrium Analysis of Slopes. Geotechnique 1983b;33(1):215-216.
- 51. Chen Z. Recent developments in slope stability analysis. In: Proceedings of the 8th International congress of ISRM, 1995, Tokyo, 1995:1041-1048.
- 52. Chen TC, Lin ML, Hung JJ. Pseudostatic analysis of Tsao-Ling rockslide caused by Chi-Chi earthquake. Engineering Geology 2003;71:31-47.
- 53. Chen Z, Mi H, Zhang F, Wang X. A simplified method for 3D slope stability analysis. Canadian Geotechnical Journal 2003;40:675-683.
- 54. Chen Z, Wang J, Wang Y, Yin JH, Haberfield C. A three-dimensional slope stability analysis method using the upper bound theorem Part II: numerical approaches, applications and extensions. International Journal of Rock Mechanics & Mining Sciences 2001a;38:379-397.
- 55. Cheng Y, Yip C. Three-dimensional asymmetrical sloe stability analysis extension of Bishop's, Junbu's, and Morgenstern-Price's Techniques. Journal of Geotechnical and Geoenvironmental Engineering 2007;133(12):1544-1555.
- Clabon, Walter Allen, Part3: General astronomical constants. In: Cox Arthur N. ed. Allen's Astrophysical Quantities, Chapter 2: General Constants and Units. 4th edition. New York: AIP Press, Springer-Verlag, 2000:12, doi: 10.1007/978-1-4612-1186-0, ISBN: 0387987460.
- 57. Clough W, Woodward R.J. Analysis of embankment stresses and deformations. ASCE, J. S. M. F. E. 1967;1(93):SM4.
- 58. Cohen B. The Investigation of Difficult Things: Essays on Newton and the History of the Exact Sciences, in Honour of Whiteside D.T, In: Harman Peter M, Shapiro Alan E, eds. Cambridge U.K.: Cambridge University Press, 2002:353.
- 59. Collins IF, Gunn CIM, Pender MJ, Yan W. Slope stability analyses for materials with a non-linear failure envelope. International Journal for Numerical and Analytical Methods in Geomechanics 1988;12:533-550.
- 60. Cosar S. Application of rock mass classification systems for future support design of the Dim Tunnel near Alanya. Middle East Technical University; 2004.
- 61. Cotecchia V. Earthquake-prone environments. In: Anderson MG, Richards KS, eds. Slope Stability, Geotechnical Engineering and Geomorphology. New York: John Wiley & Sons, 1987:287-330.
- 62. Cottiss GI, Dowel RW, Franklin JA. A rock classification system applied in civil engineering, part 1. Civil Eng. and Public Works Review June 1971:611-614.
- 63. Cottiss GI, Dowel RW, Franklin JA. A rock classification system applied in civil engineering, part 2. Civil Eng. and Public Works Review July 1971:736-743.
- 64. Cruden DM, Varnes DL. Chapter 3: Landslide types and processes. In: Turner AK, Schuster RI, eds. Landslides: Investigation and Mitigation, Special Report 247, Transportation Research Board, National Research Council, Washington D.C.: National Academy Press, 1996:36-75.
- 65. Cruden DM. A Simple Definition of a Landslide. Bulletin of the International Association of Engineering Geology 1991;43:27-29.
- 66. Das SK. Slope stability analysis using genetic algorithm. Electron. J. Geotech. Engr. 2005;10, http://www.ejge.com/2005/Ppr0504/Abs0504.htm
- 67. Day AP, Seery JM. Monitoring of a large wall failure at Tom Price Iron Ore Mine. In: Potvin Y, editor. International Symposium on rock slope stability in open pit mining and civil engineering, 2007, Perth, 2007:333-340.
- 68. Deere DU. Technical description of rock cores for engineering purposes. Felsmechanik und Ingenieurgeologie 1963;1(1):16-22.
- 69. Deere DU. Technical description of rock cores for engineering purposes. Rock Mechanics and Engineering Geology 1964;1(1):17-22.

Available Online: http://www.easpublisher.com/easmb/

Mohammad Salehi Alashti.; East African Scholars Multidiscip Bull; Vol-1, Iss-3 (Dec, 2018): 61-121

- Deere DU. Rock quality designation (RQD) after 20 years. U.S. Army Corps of Engineers Contract Report GL-89-1. Vicksburg, MS, USA: Waterways Experimental Station, 1989:67.
- Deere DU, Deere DW. The rock quality designation (RQD) index in practice. In: Kirkaldie L, editor. Proc. Symposium of the Rock classification systems for engineering purposes, 1988, ASTM Special Technical Publication 984. Philadelphia: American Society for Testing Materials (Am. Soc. Test. Mat.), 1988:91-101.
- 72. Deere DU, Hendron AJ, Patton FD, Cording EJ. Design of surface and near surface constructions in rock. In: Fairhurst C, editor. Proc. 8th U.S. Symposium on Rock Mechanics, In Failure and breakage of rock, 1967, New York: Society of Mining Engineers (Soc. Min. Engrs), American Institute of Mining, Metallurgical, and Petroleum Engineers AIME (Am. Inst. Min. Metall. Petrolm Engrs.), 1967:237-302.
- 73. Deere DU, Merritt AH, Coon RF. Engineering classification of in-situ rock, Tech. ReDt. No. AFWL-TR-67144. Kirtland Air Force Base, New Mexico: 1969.
- 74. Deere DU, Miller DW. The rock quality designation (RQD) index in practice. Classification Systems for Engineering Purposes. In: Proc. Symp. Rock Classification Systems for Engineering Purposes, 1967, ASTM STP (Special Technical Publication). Philadelphia: American Society for Testing and Materials, 1967:91-101.
- 75. Dennhardt M, Forster W. Problems of three dimensional slope stability. In: Proceeding of the 11th International Conference in Soil Mechanics and Foundation Engineering, 1985, San Francisco, Part 2, 1985:427-431.
- 76. Drescher A, Christopoulos C. Limit analysis slope stability with nonlinear yield conditions. International Journal for Numerical and Analytical Methods in Geomechanics 1988;12:341-345.
- 77. Eberhardt Erik. Rock slope stability analysis Utilization of advanced numerical techniques, Geological Engineering, Earth and Ocean Sciences, University of British Colombia, Vancouver, Canada, April, 2003. p. 6.
- 78. Eberhardt E, Stead D, Coggan JS. Numerical analysis of initiation and progressive failure in natural rock slopes— the 1991 Randa rockslide. International Journal of Rock Mechanics & Mining Science 2004;41:69-87.
- 79. Ehlers J. Relativity, astrophysics and cosmology. In: Werner I. Dordrecht, Reidel D, eds. Survey of General Relativity Theory, Academic Publishing Company, 1973:16.
- 80. Einstein HH, Nelson RA, Bruhn RW, Hirschfeld RC. Model studies of jointed-rock behavior. In: Proc. 11th US Symposium, 1970, Berkeley, A.I.M.E.: New York, 1970:83-103.
- Elmo D, Yan M, Stead D, Rogers SF. The importance of intact rock bridges in the stability of high rock slopes toward a quantitative investigation using an integrated numerical modelling; discrete fracture network approach. In: Potvin Y, editor. International Symposium on rock slope stability in open pit mining and civil engineering, 2007, Perth, 2007:253-266.
- 82. Farshchi A, Samadani K, Heidari M. Optimization in Sungun copper mine, Iran, Application of Computers and Operations Research in the Mineral Industry. In: Proceedings of the 32nd International Symposium on the Application of Computers and Operations Research in the Mineral Industry (APCOM), 30 March 1 April, 2005, Tucson, USA.
- 83. Federal Highways Administration. Rockfall hazard rating system, participants' manual for NHI Course No. 130220. Publication No. FHWA SA -93-057, 1993.
- 84. Fellenius W. Calculations of the stability of Earth dams. In: Proceedings of the Second Congress of Large Dams, 1936, Washington DC., Vol. 4., 1936:445-463.
- 85. Franklin JA. Observations and tests for engineering description and mapping of rocks. In: Proc. 4th Cong. of ISRM, 1970, Belgrade, Vol. 1, 1970:1-3, 6 p.
- 86. Franklin JA. Safety and economy in tunneling. In: Proc., 10th Canadian Rock Mechanics Symposium, 1975, Queens Univ., Kingston, Ontario, Vol. 1, 1975:27-53.
- 87. Franklin JA. An observation approach to the selection and control of rock tunnel linings. In: Proc., Conference on Shotcrete for Ground Support, 1976, Easton, MD, 1976:556-596.
- 88. Franklin JA. Size-strength system for rock characterization. Karmis M., ed., In: Proc. Symp. Applications of Rock Characterization Techniques to Mine Design, Am. Soc. Min. Eng. Ann. Mtg., 1986, New Orleans, La, New York: AIME, 1986:11-16.
- 89. Franklin JA, Broch E, Walton G. Logging the mechanical character of rock. Tran. Inst. Min. Metall. 1971;A80:A1-A9.
- 90. Franklin JA, Louis C, Masure P. Rock material classification. In: Proc. 2 Int. Cong. Eng Geol., IAEG, 1974, Sao Paulo, 1974:325–341.
- 91. Gens A, Hutchinson JN, Cavounidis S. Three-dimensional analysis of slides in cohesive soils. Geotechnique 1988;38:1–23.
- 92. Ghose AK, Raju NM. Characterization of rock mass vià application of rock bolting in Indian coal measures. In: Einstein HH, editor. Proceedings of the 22nd US Symposium on Rock Mechanics (Proc. 22nd U.S. Symp. Rock Mech.), 1981, MIT, Cambridge, Boston, MA, 1981:422-427.
- 93. Giger MW, Krizek RJ. Stability analysis of vertical cut with variable corner angle. Soils and Foundation, Japanese Society for Soil Mechanics and Foundation Engineering June, 1975;15(2):63-71.

- 94. Giger Max W, Krizek Raymond J. Stability of vertical corner cut with concentrated surcharge load. ASCE, Journal of the Geotechnical Engineering Division Jan., 1976;102(GT1).
- 95. Goh ATC. Genetic algorithm search for critical slip surface in multiple-wedge stability analysis. Can. Geotech. J. 1999;36(2):382–391.
- 96. Goldscheider M, Dahmen D, Karcher C. Consideration of earthquakes in stability calculations for deep underwater final slopes. In: World of Mining-Surface & Underground, Vol. 62. Germany, 2010:252-261.
- 97. Gonzalez de Vallejo L.I., A new rock classification system for underground assessment using surface data. In: Proceedings of International Symposium Engineering Geology and Underground Construction, 1983, LNEC, Lisbon, Portugal, Vol. 1, 1983:II.85–II.94.
- 98. González de Vallejo LI. Tunnelling evaluation using the Surface Rock Mass Classification System (SRC). In: ISMR Symp. The Role of Rock Mechanics in Excavations for Mining and Civil Works, 1985, Soc. Mexicana de Mec. Rocas, Zacatecas, México, vol. 1, 1985:458–46 6.
- 99. Goodman RE. Introduction to Rock Mechanics, 2nd edition. New York: Wiley, 1989.
- 100.Goodman RE, Kieffer DS. Behavior of rock in slope. Journal of Geotechnical and Geoenviromental Engineering 2000;126(8):675-684.
- 101.Goodman RE, Shi G. Block Theory and its Application to Rock Engineering, NJ.: Prentice Hall, Englewood Cliffs, 1985.
- 102.Grasselli G, Egger P. Constitutive law for the shear strength of rock joints based on three-dimensional surface parameters. International Journal of Rock Mechanics & Mining Science 2003;40:25-40.
- 103.Greco VR. Efficient Monte Carlo technique for locating critical slip surface. J. Geotech. Engrg. 1996;122 (7):517–525.
- 104.Groten E. The geodesist's handbook 2012 part 5: general information; geodetic standards and conventions. Edited by International Association of Geodesy (IAG) Office and IAG Communication and Outreach Branch (COB), Springer Journal of Geodesy October 2012;86(10):962-963, available at: (http://link.springer.com/content/pdf/10.1007%2Fs00190-012-0584-1.pdf).
- 105.Hack, Robert (H.R.G.K), Slope stability probability classification (SSPC), 2nd edition (2nd Version). Published by Netherlands: International institute for aerospace survey and earth sciences (ITC), ITC Publication, Number 43., 1998:24-32, 40, 42, ISBN: 90-6164-154-3.
- 106.Hack Robert. An evaluation of slope stability classification, Keynote Lecture. Dinis da Gama C, Ribeira e Sousa L, Eds., In: Proceeding of the ISRM EUROCK 2002, November 2002, Funchal, Madeira, Portugal: Sociedade Portuguesa de Geotecnia, Av. do Brasil, 101, 1700-066 Lisboa, Portugal, 2002:1,11,3-32. ISBN: 972-98781-2-9.
- 107.Hack R, Alkema D, Kruse GAM, Leenders N, Luzi L. Influence of earthquakes on the stability of slopes. Engineering Geology 2007;91:4-15.
- 108.Hack HRGK, Price DG. A rock mass classification system for the design and safety analysis of slopes, In: e Sousa Ribeiro, Grossmann, editors. Proc. Int. Conf. Eurock '93, Rotterdam: Balkema, 1993:803-810.
- 109.Hack R, Price D, Rengers, N. A new approach to rock slope stability a probability classification (SSPC). Bulletin of Engineering Geology and the Environment 2003;62:167-184 Springer Verlag & erratum: pp. 185-185, article: DOI 10.1007/s10064-002-0155-4 & erratum: DOI 10.1007/s10064-002-0171-4 (printed-on-line, paper in print).
- 110. Hains A, Terbrugge PJ. Preliminary estimation of rock slope stability using rock mass classification systems. In: Proc 7th Int. Cong Rock Mech., 1991, Vol. 2, 1991:887–892.
- 111.Hajjarian Ahmad. Production Scheduling of Sungun Copper Mine with NPV Scheduler and Whittle 4X Softwares. M.SC. Dissertation in mining engineering, in Persian (FARSI, PARSI) language, Tehran, Iran: Department Of Mining Engineering, Faculty of Technical and Engineering (FTE), Sciences and Researches Branch (SRB), Islamic Azad University (IAU), 2007. [Dissertation] [In Persian]
- 112.Harman J, Hormazabal E, Martínes C. Fact and fiction about pit slope depressurization. In: Potvin Y, editor. International Symposium on rock slope stability in open pit mining and civil engineering, 2007, Perth, 2007:361-370.
- 113.Hartmann F, Katz C. Structural Analysis with Finite Elements, Berlin: Springer-Verlag, 2007.
- 114. Hoek E. Strength of rock and rock masses. ISRM News Journal 1994;2(2):4-16.
- 115.Hoek E, Bray JW. Rock Slope Engineering, 3rd edition, London: Institute of Mining and Metallurgy, 1981:402 p.
- 116.Hoek E, Brown ET. Empirical strength criterion for rock masses. ASCE, Journal of the Geotechnical Engineering Division (J. Geotech. Engng Div.) 1980a;106(GT 9):1013-1035.
- 117.Hoek E, Brown ET. Underground Excavations in Rock, London: Institution of Mining and Metallurgy, 1980:527 p.

- 118.Hoek E, Brown ET. Practical estimates of rock mass strength. International Journal of Rock Mechanics & Mining Sciences & Geomechanics Abstracts 1997;34(8):1165-1186.
- 119.Hoek E, Carranza-Torres C, Corkum B. Hoek-Brown failure criterion-2002 edition. In: Proceedings of the 5th North American rock mechanics society meeting and symposium, 2002, Toronto. Symp., 17th Tunnelling Association of Canada Conf.: NARMS-TAC, University of Toronto, 2002:267–271.
- 120.Hoek E, Kaiser PK, Bawden WF. Support of Underground Excavations in Hard Rock, Rotterdam, Brookfield: A.A. Balkema, 1995.
- 121.Hoek E, Marinos P, Benissi M. Applicability of the geological strength index (GSI) classification for very weak and sheared rock masses. The case of the Athens schist formation. Bull. Eng. Geol. Env. 1998;57:151-160.
- 122.Hoek E, Read J, Karzulovic A, Chen ZY. Rock slopes in civil and mining engineering. In: Proceedings of the International Conference on Geotechnical and Geological Engineering, GeoEng2000, Melbourne, 2000:19-24.
- 123.Hovland HJ. Three-dimensional slope stability analysis method. ASCE, Journal of the Geotechnical Engineering Division 1977;103(9):971-986.
- 124.Huang CC, Lee YH, Liu HP, Keefer DK, Jibson RW. Influence of surface-normal ground acceleration on the initiation of the Jih-Feng-Erh-Shan landslide during the 1999 Chi-Chi, Taiwan, earthquake. Bulletin of Seismological Society of America 2001;91(5):953-958.
- 125.Huang CC, Tsai CC. New method for 3D and asymmetrical slope stability analysis. Journal of Geotechnical and Geoenvironmental Engineering 2000;126(10):917-927.
- 126.Huang CC, Tsai CC, Chen YH. Generalized method for three-dimensional slope stability analysis. Journal of Geotechnical and Geoenvironmental Engineering 2002;128(10):836-848.
- 127.Hudson JA. Rock Engineering Systems, Theory and Practice, Chichester, England: Ellis Horwood Ltd., 1992:185 pp.
- 128. Hungr O. An extension of Bishop's Simplified Method of slope stability analysis to three dimensions. London, Geotechnique 1987;37(1):113-117.
- 129.Hungr O, Salgado FM, Byrne PM. Evaluation of a three-dimensional method of slope stability analysis. Canadian Geotechnical Journal 1989;26:679-686.
- 130.Hutchinson JN, Sarma SK. Discussion of three-dimensional limit equilibrium analysis of slopes, by RH Chen and JL Chameau. London, Geotechnique 1985;35(2):215-216.
- 131.Hynes-Griffin MR, Franklin AG. Rationalizing the seismic coefficient method. U.S. Army Corps of Engineers Report WES/MP/GL-84-13. Vicksburg, USA: 1984.
- 132. Jaeger JC. Friction of rocks and stability of rock slopes. Geotechnique 1971;21(2):97-134.
- 133.Janbu N. Application of composite slip surface for stability analysis. European Conference on Stability of Earth Slopes, 1954, Stockholm.
- 134.Janbu N. Earth pressure and bearing capacity calculations by generalized procedure of slices. In: Proceedings of 4th International Conference on Soil Mechanics and Foundation Engineering, 1957, Vol. 2, 1957:207-212.
- 135.Janbu N. Slope stability computations, embankment dam engineering-Casagrande volume. N.Y: John Wiley and Sons, 1973.
- 136.Janbu N, Bjerrum L, Kjaernsli B. Soil mechanics applied to some engineering problems. Publication 16. Norway: Norwegian Geotechnical Institute, 1956:5-26.
- 137.Jiang JC, Yamagami T, Three-dimensional slope stability analysis using an extended Spencer method. Journal of the Japanese Geotechnical Society of Soils and Foundations 2004;44(4):127-135.
- 138. Jianping Sun, Jiachun Li, Qingquan Liu. Search for critical slip surface in slope stability analysis by splinebased genetic algorithm GA method. Technical Notes, ASCE, Journal of Geotechnical and Geo-environmental Engineering February 1, 2008;134(2):ISSN: 1090-0241.
- 139.Kalatehjari Roohollah, Nazri Ali. A review of three-dimensional slope stability analyses based on limit equilibrium method. Electronic Journal of Geotechnical Engineering (EJGE) 2013;18(A):119-134.
- 140.Kendorski F, Cummings R, Bieniawski ZT, Skinner E. Rock mass classification for block caving mine drift support. In: Proceedings of 5th International Congress on Rock Mechanics, 1983, Melbourne, Vol. 1. ISRM, 1983:B51–B63.
- 141. Kenyon IR. General Relativity, oxford: Oxford University Press, sec. 7.2, 1990.
- 142.Kirsten HAD. A classification for excavation. Civ. Eng S Afr. 1982.;7:293.
- 143.Kirsten HAD. Significance of the probability of failure in rock slopes. The Civil Engineer in South Africa 1983;25(1):17-27.
- 144.Kopacsy J. Three-dimensional stress distribution and slip surfaces in Earth works at rupture. In: Proceeding of the 4th International Conference on Soil Mechanics and Foundation Engineering, 1957, London, England, 1957:339-342.
- 145.Kovari K. Erroneous concepts behind NATM. lecture given at Rabcewicz Colloquium, Salzburg, Zurich: Published by Swiss Federal Institute of Technology, October 1993.

Mohammad Salehi Alashti.; East African Scholars Multidiscip Bull; Vol-1, Iss-3 (Dec, 2018): 61-121

146.Kramer SL. Geotechnical Earthquake Engineering, New Jersey: Prentice Hall, 1996.

- 147.Kulhawy F, Duncan J. Stresses and movements in Oroville dam. ASCE, J. S. M. F. D. 1972;98:SM7.
- 148.Ladanyi B, Archambault G. Simulation of shear behavior of a jointed rock mass. In: Proceedings 11th US Symposium on Rock Mechanics, 1970, Berkeley, New York: A.I.M.E., 1970:105-125.
- 149.Ladanyi B, Archambault G. Evaluation de la résistance au cisaillement d'un massif rocheux fragmenté. In: Proc. 24th Int. Geological Congress, 1972, Montreal, Canada, Vol. 18, 1972:249-260.
- 150.Lam L, Fredlund D. A general limit equilibrium model for three-dimensional slope stability analysis. Canadian Geotechnical Journal 1993;30:905-919.
- 151.Laubscher DH. Geomechanics classification of jointed rock masses Mining applications. Trans. Inst. Min. Metall. (Section A) 1977;86:A1-A8.
- 152.Laubscher DH. Design aspects and effectiveness of support systems in different mining situations. Transactions of the Institute of Mining and Metallurgy 1977;86:A1-A7.
- 153.Laubscher DH. Design aspects and effectiveness of support systems in different mining conditions. Trans Inst. Min. Metall. 1984;93:A70-A82.
- 154.Laubscher D.H. A geomechanics classification system for rating of rock mass in mine design. Journal South African Inst. of Mining and Metallurgy 1990;90(10):257-273.
- 155.Laubscher DH, Page CH. The design of rock support in high stress or weak rock environments. In: Proceedings of the 92nd Canadian Institute of Mining and Metallurgy, 1990, Ottawa.
- 156.Laubscher DH, Taylor HW. The importance of geomechanics classification of jointed rock masses in mining operations. In: Bieniawski, Z.T., ed. Volume 1: Exploration for Rock Engineering. Cape Town: Balkema, 1976:119-128.
- 157.Lauffer H. Gebirgsklassifizierung für den Stollenbau. Geol. Bauwesen 1958;24(1):46-51.
- 158.Leshchinsky D. Slope stability analysis: generalized approach, closure. ASCE, Journal of Geotechnical Engineering Feb. 1992;118(2):358-359.
- 159.Leshchinsky D, Baker R, Silver ML. Three-dimensional analysis of slope stability," International Journal for Numerical and analytical Methods in Geomechanics 1985;9(2):199-223.
- 160.Leshchinsky D, Baker R. Three-dimensional slope stability: end effects. JSSMFE, Soils and Foundations 1986;26(4):98-110.
- 161.Leshchinsky D, Huang CC, Generalized slope stability analysis: Interpretation, Modification, and Comparison. ASCE, Journal of Geotechnical Engineering October 1992;118(10):1559-1576.
- 162.Leshchinsky D, Huang C. Generalized three dimensional slope stability analysis. ASCE, Journal of the Geotechnical Engineering Division November 1992;118(11):1748-1764.
- 163.Leshchinsky D, Mullett T. Design charts for vertical cuts. Journal of Geotechnical and Geoenvironmental Engineering 1988a;114(3):337-344.
- 164.Leshchinsky D, Mullett T. Stability of vertical corner cuts. In: Proceeding of the Sixth International Conference on Numerical Methods in Geomechanics, 1988, Innsbruck, Austria, Rotterdam: AA Balkema, 1988b:1249-1256.
- 165.Lewin W. Work, energy, and universal gravitation, MIT Course 8.01: Classical Mechanics, Lecture 11, Videotape, (Cambridge, MA USA: MIT OCW), Event occurs at 1:21-10:10, October 4, 1999, Retrieved December 23, 2010, [In English-ogg].
- 166.Li An jui. Two and Three Dimensional Stability Analyses for Soil and Rock Slopes. PhD Dissertation, Australia: School of Civil and Resources Engineering, The University of Western Australia, August 2009: 4, 13, 19. [Dissertation]
- 167.Li KS, White W. Rapid evaluation of the critical slip surface in slope stability problems. Int. J. Numer. Analyt. Meth. Geo-mech. 1987;11(5):449–473.
- 168.Lin Y. An introduction of the Chinese standard for engineering classification of rock masses (GB50218-94). Advances in Rock Mechanics, World Scientific Publishing, Lin Ed. 1998;317-327.
- 169.Ling HI, Cheng AHD. Rock sliding induced by seismic force. International Journal of Rock Mechanics & Mining Science 1997;34(6):1021-1029.
- 170.Ling HI, Leshchinsky D, Mohri Y. Soil slopes under combined horizontal and vertical seismic acceleration. Eaethquake engineering and structure dynamics 1997;26:1231-1241.
- 171.Liu GR, Quek SS. The finite element method: a practical course, Elsevier Science Ltd., 2003.
- 172.Loukidis D, Bandini P, Salgado R. Stability of seismically loaded slopes using limit analysis. Geotechnique 2003;53(5):463-479.
- 173.Luo HY, Zhou W, Huang SL, Chen G. Earthquake-induced landslide stability analysis of the Las Colinas Landslide in El Salvador. In: I.J.o.R.M.M. Sciences, editor. SINOROCK2004 Symposium, Vol.41. Elsevier Ltd., 2004:1-6.

- 174.Malkawi AIH, Hassan WF, Sarma SK, Global search method for locating general slip surface using Monte Carlo techniques. J. Geotech. Geoenviron. Eng. 2001;127(8):688–698.
- 175.Maps of Google. Available at: https://maps.google.com, [Date accessed: 12 March 2015].
- 176.Marinos P, Hoek E. GSI: A geologically friendly tool for rock mass strength estimation. In: Proceedings GeoEng2000 Conference, Melbourne, 2000:1422-1442.
- 177. Marinos P, Hoek E. Estimating the geotechnical properties of heterogeneous rock masses such as Flysch. Bull. Engng. Geol. Env. 2001;60:85-92.
- 178.Marinos V, Marinos P, Hoek E. The geological strength index: applications and limitations. Bull. Eng. Geol. Env. 2005;64:55-65.
- 179.Martens PN, Katz T, Ahmad S, Fuchsschwanz M. Investigations on stabilization of hard coal waste dump in Vietnam. In: World of Mining-Surface & Underground, 2011, Vol. 63. Germany, 2011:265-274.
- 180.Matthews Carol, Farook Zeena, Helm Peter. Ground engineering May 2014;22, Technical Paper.
- 181.Matula M, Holzer R. Engineering topology of rock masses. In: Proc. of Felsmekanik Kolloquium, Grundlagen ung Andwendung der Felsmekanik, 1978, Karlsruhe, Germany, 1978:107-121.
- 182.McCombic P, Wilkinson P. The use of the simple genetic algorithm in finding the critical factor of safety in slope stability analysis. Comput. Geotech. 2002;29(8):699–714.
- 183.Mazzoccola DF, Hudson JA. A comprehensive method of rock mass characterization for indicating natural slope instability. Quarterly Journal of Engineering Geology. 1996;29:37-56.
- 184.Merifield RS, Lyamin AV, Sloan SW. Limit analysis solutions for the bearing capacity of rock masses using the generalised Hoek-Brown criterion. International Journal of Rock Mechanics & Mining Sciences 2006;43:920-937.
- 185. Michalowski RL. Three-dimensional analysis of locally loaded slopes. Geotechnique 1989;39(1):27-38.
- 186.Mobin Mining & Construction Company. Available at: <u>http://www.mobinco.com/en/projects/sungun/</u> [Date accessed: 15 March 2015].
- 187.Mohr PJ, Taylor BN, Newell DB. 2010. The 2010 CODATA recommended values of the fundamental physical constants (Web Version 6.0), this database was developed by J. Baker, M. Douma, and S. Kotochigova. Available at: <u>http://physics.nist.gov/constants</u> [Thursday, 02-Jun-2011 21:00:12 EDT], National Institute of Standards and Technology, Gaithersburg, MD 20899, 2011.
- 188.Mohr PJ, Taylor BN, Newell DB. CODATA recommended values of the fundamental physical constants: 2010. Reviews of Modern Physics 13 November 2012;84, National Institute of Standards and Technology, Gaithersburg, Maryland 20899-8420, USA, 2011 doi: http://dx.doi.org/10.1103/RevModPhys.84.1527, [PACS numbers: 06.20.Jr, 12.20.-m] available at: (http://physics.nist.gov/cuu/Constants/Preprints/Isa2010.pdf & http://physics.nist.gov/cuu/Constants/index.html).
- 189.Mohr Peter J, Taylor Barry N, Newell David B. CODATA recommended values of the fundamental physical constants: 2010 (2012 adjustment of 2010 CODATA values). J. Phys. Chem. Ref. Data. 28 December 2012;41(4), National Institute of Standards and Technology, Gaithersburg, Maryland 20899-8420, USA, doi:http://dx.doi.org/10.1063/1.4724320 arXiv:1203.5425 https://arxiv.org/abs/1203.5425.
- 190.Mohr PJ, Taylor BN, Newell DB. CODATA recommended values of the fundamental physical constants: 2014. Gaithersburg, Maryland 20899-8420, USA: National Institute of Standards and Technology, July 30 2015, arXiv:1507.07956v1 [physics.atom-ph].
- 191.Molinda G, Mark C. The coal mine roof rating (CMRR)-A practical rock mass classification for coal mines. US Bureau of Mines Information Circular (USBM IC) IC 9387. 1994:83 pp.
- 192.Moon BP, Selby MJ. Rock mass strength and scarp forms in Southern Africa. Geografisika Annaler. Ser. A 1990;65:135-145.
- 193.Moreno, Talion E. Comparison and application of the geomechanics classification schemes in tunnel construction. In: Proc. Tunneling '82, Institute of Mining and Metallurgy, London, 1982:241-246.
- 194.Morgenstern NR, Price V.E. The analysis of the stability of general slip surfaces. Geotechnique 1965;15:77-93.
- 195.Morgenstern NR, Price VE. A numerical method for solving the equations of stability of general slip surfaces. Great Britain, The Computer Journal February 1967;9(4):388-393.
- 196.Müller L. Removing misconceptions on the new Austrian tunnelling method. Tunnels and Tunnelling Oct. 1978, 1978a:29-32. Also in Tunnels and Tunnelling, Special NATM edition, Summer 1990:15–18.
- 197.Muller L, Pacher F. Bearing capacity of abutments of dams, especially arch dam in rock. 8th Congress on large Dams, 1964, Edinburgh, Vol.1, 1964:935-954.
- 198.Nakao K, Iihoshi S, Koyama S. Statistical reconsiderations on the parameters for geomechanics classification. In: Proceedings of the 5th International Congress on Rock Mechanics, 1983, ISRM, Melbourne, Vol. 1, 1983:B13-B16.
- 199.National Geoscience Database (NGDIR) of IRAN 2011. Available from: http://www.ngdir.com [Date accessed: 12 March 2015].

Mohammad Salehi Alashti.; East African Scholars Multidiscip Bull; Vol-1, Iss-3 (Dec, 2018): 61-121

- 200.Newman DA. A modified version of the geomechanics classification for use in coal mines. In: Chugh, editor. Proc. 2nd Conf. on Ground Control in the Illinois Basin, 1985, S. Ill. Univ., Carbondale, 1985:64-72.
- 201.Newman D.A. The Design of Coal Mine Roof Support and Yielding Pillars for Longwall Mining in the Appalachian Coalfield. PhD dissertation, The Pennsylvania State University, 1985: 392 p. [Dissertation]
- 202.Newman DA, Bieniawski ZT. A modified version of the geomechanics classification for roof span design in underground coal mines. AIME- SME (Society for Mining, Metallurgy, and Exploration) Transactions, Littleton, CO, 1986;280:2134-2138. 280:2134-2138.
- 203. Newmark NM. Effects of earthquakes on dams and embankments. Geotechnique 1965;15(2):139-160.
- 204.Newton Isaac. Author. The Principia: Mathematical Principles of Natural Philosophy, In: Cohen B, Whitman A. translators. Preceded by a guide to Newton's principia by Cohen I Bernard, California: University of California Press, Proposition 75, Theorem 35, 1999:956.
- 205.Newton's Laws of Motion. Wikipedia, Available at: https://en.wikipedia.org/wiki/Newton%27s_laws_of_motion. [Date accessed: 12 March 2015].
- 206.Nieto AS. Some geologic factors in the location design and construction of large underground chambers in rock. In: Proc. Rapid Exc. & Tunn. Conf., 1983, AIME, 1983:569-596.
- 207.Nicholson D.T. Deterioration of Excavated Rock Slopes: Mechanisms, Morphology, and Assessment. PhD Thesis, University of Leeds, 2000. [Dissertation]
- 208.Nicholson DT. Identification of deterioration hazard potential for quarried rock slopes. Quarry Management Jan. 2002;29(1):43-52.
- 209.Nicholson DT. Breakdown mechanisms and morphology for man-made rock slopes in North West England. North West Geography 2003;3(1):12-26.
- 210.Nicholson DT. Assessment of excavated rock slope hazard due to progressive, weathering-related deterioration. Quarterly Journal of Engineering Geology and Hydrogeology 2003.
- 211.Nicholson DT. Q J Eng Geol Hydrogeol. 2004;37:327-346.
- 212.Nicholson DT, Hencher SR. Assessing the potential for deterioration of engineered rock slopes. In: Marinos PG, Koukis GC, Tsiambaos GC, Stournaras GC, editors. Proceedings of the IAEG symposium, 1997, Athens, Engineering Geology and the Environment, Rotterdam: Balkema, 1997:911-917.
- 213.Nicholson DT, Lumsden AC, Hencher SR. 2000. Excavation induced deterioration of rock slopes. In: Bromhead E, Dixon N, Ibsen M-L, editors. Landslides: Landslides in research, theory and practice, Proceedings of the 8th International Symposium, 2000, Cardiff, Wales, Vol.3. London: Thomas Telford, 2000:1105-1110.
- 214.Nuric Adila, Nuric Samir, Kricak Lazar, Husagic Resad. Numerical methods in analysis of slope stability. International Journal of Science and Engineering Investigations March 2013;2(14), ISSN: 2251-8843.
- 215.Nutakor D. Mitigation of the Cplus failure in the Q7 cut at Rio Tinto's Bingham canyon mine. Mining engineering 2012;64(6):107-111, USA.
- 216.Olivier HJ. New engineering-geological rock durability classification. Engineering Geology (Engr. Geolo.) 1979;14:255-279.
- 217.Pacher F. The development of the New Austrian Tunnelling Method and the main features in design work and construction. In: 16th Symp. On Rock Mechanics, 1975, Minneapolis, 1975:223-232.
- 218.Pacher F, Rabcewicz L, Golser J. Zum der seitigen Stand der Gebirgsklassifizierung in Stollen und Tunnelbau. Proc. XXII Geomech. colloq., 1974, Salzburg, 1974:51 -58.
- 219.Pakalnis R, Brady TM, Hughes P, Caceres C, Ouchi AM, MacLaughlin MM. Weak rock mass design for underground mining operations. In: Mark C, Pakalnis R, Tuchman RJ, editors. Proceedings of the International Workshop on Rock Mass Classification in Underground Mining, 2007, Pittsburgh, PA, U.S. Department of Health and Human Services, Public Health Service, Centers for Disease Control and Prevention, National Institute for Occupational Safety and Health, May 2007:119-127.
- 220.Palmstrom A. RMi-A Rock Mass Characterization System for Rock Engineering Purposes. PhD Thesis. Oslo, Norway: Department of Geology, University of Oslo; 1995:400.
- 221.Palmström A. Characterizing rock masses by the RMi for use in practical rock engineering; Part1: The development of the rock mass index. Journal of Tunnelling and Underground space Technology 1996;11(2):1, 2, 7, 8, 9, Available at: <u>http://www.rockmass.net</u>.
- 222.Palmström A. Characterizing rock masses by the RMi for use in practical rock engineering; Part2: Some practical applications of the rock mass index. Journal of Tunnelling and Underground space Technology 1996;11(3):1, Available at: <u>http://www.rockmass.net</u>.
- 223.Palmström A. On classification system. In: Proceedings of Workshop on Reliability of Classification Systems a Part of the International Conference "GeoEng-2000", 2000, Melbourne.
- 224.Palmström A. Recent developments in rock support estimates by the RMi. Journal of Rock Mechanics and Tunnelling Technology May 2000;6(1):11, Available at: <u>http://www.rockmass.net</u>.

- 225.Palmstrom A, Broch E. Use and misuse of rock mass classification systems with particular reference to the Q-system. Tunnelling and Underground Space Technology 2006;21(6):575–593. doi:10.1016/j.tust.2005.10.005.
- 226.Pantelidis Lysandros. Rock slope stability assessment through rock mass classification systems. International Journal of Rock Mechanics and Mining Sciences 2009.;46(2):315–325. doi:10.1016/j.ijrmms.2008.06.003.
- 227.Patching TH, Coates DF. A recommended rock classification for rock mechanics purposes. CIM Bulletin 1968:1195-1197.
- 228.Petri R, Stein W. Opencast mine slopes Stability of slopes in opencast lignite mines, North Rhine-Westphalia. In: World of Mining-Surface & Underground, 2012, Vol. 64. Germany, 2012:114-125.
- 229.Pierson LA, Davis SA, Van Vickle R. Rockfall hazard rating system: Implementation manual. Federal Highway Administration (FHWA) Report FHWA-OR—EG-90-01. USA: FHWA, U.S. Department of Transportation; 1990.
- 230.Plastino Angel R, Muzzio Juan C. On the use and abuse of Newton's second law for variable mass problems. Celestial Mechanics and Dynamical Astronomy 1992;53(3):227–232, Kluwer Academic Publishers, Netherlands.
- 231.Pradel D, Smith PM, Stewart JP, Raad G. Case history of landslide movement during the Northridge Earthquake. Journal of Geotechnical and Geoenviromental Engineering 2005;131(11):1360-1369.
- 232.Rabcewicz Lv. The new Austrian tunneling method. Water Power, part 1 November 1964:511-515, Part 2 January 1965, 1964/65:19 -24.
- 233.Rabcewicz Lv, Golser J. Principles dimensioning the support system for the new Austrian tunnelling method. Water Power, March 1973:88-93.
- 234.Rahul. Dump Slope Rating for Coal Mine and Prediction of Factor of Safety of Dump Slope by Artificial Neural Network. Unpublished, M. Tech Thesis, Varanasi: Department of Mining Engineering, IIT (BHU), 2011. [Dissertation]
- 235.Rai Rajesh. Classification systems in slope stability analysis; Slope stability classification systems. p. 1-13.
- 236.Ritter W. Die Statik der Tunnelgewölbe, Berlin: Springer, 1879.
- 237.Robertson AM. Estimating weak rock strength. In: Sastry KVS, editor. Proceedings of the SME Annual meeting, Society of Mining Engineering (AIME-SME Annual Meeting), 1988, Phoenix, AZ., 1988:1-5, Preprint #pp. 88-145.
- 238.Rock mass classification. Wikipedia, Available at: http://en.wikipedia.org/wiki/Rock_mass_classification [Date accessed: 12 March 2015].
- 239.Romana M. New adjustment rating for application of the Bieniawski classification to slopes. In: Proceedings of the International Symposium on the Role of Rock Mechanics in Excavations for Mining and Civil Works (Proc. Int. Symp. Rock Mechanics Mining Civ. Works), International Society of Rock Mechanics (ISRM), 1985, Zacatecas, Mexico City, 1985:49-68.
- 240.Romana, M., SMR classification, Wittke W, Ed.. In: Proc. 7th Congress on Rock Mechanics, ISRM, 1991, Aachenm, Germany, Vol. 2. Rotterdam: Balkema, 1991:955–960.
- 241.Romana M. Geomechanical classification for slopes: slope mass rating. In: Hudson John A, ed. Vol. 3: Comprehensive Rock Engineering. Pergamon Press, 1993:575–600.
- 242.Romana M. The SMR geomechanical classification for slopes: A critical ten years review. In: Chacon Irigaray, Fernandez editors. Proc. 8th Int. Conf. Landslides, 1995, Granada, Spain, Rotterdam: AA Balkema, 1995:255–267.
- 243.Romana M. The geomechanical classification SMR for slope correction. In: Proc. 8th Int. Congress Rock Mechanics, 1995, Tokyo, Vol. 3. ISRM, 1995:1085-1092.
- 244.Romana M, Seron JB, Montalar E. SMR Geomechanics classification: Application, experience and validation. In: Proceedings of the international symposium on the role of rock Mechanics, 2003, ISRM, Technology roadmap for rock mechanics, South African Institute of Mining and Metallurgy, 2003:1–4.
- 245.Russo G. A new rational method for calculating GSI. Tunneling and Underground Space Technology 2009;24:103-111.
- 246.Salehi Alashti M. Accurate age of the earth calculation by a new fundamental parameter. Open Access Library Journal 2015;2:e1296. <u>http://dx.doi.org/10.4236/oalib.1101296</u>. Available at OALib: http://www.oalib.com/paper/3135188 & http://dx.doi.org/10.4236/oalib.1101296 & http://www.oalib.com/paper/pdf/3135188 and Available at Scientific research publishing (scirp): http://www.scirp.org/journal/PaperInformation.aspx?PaperID=68071 & http://dx.doi.org/10.4236/oalib.1101296 & http://www.oalib.com/paper/pdf/3135188
- 247.Salehi Alashti, M., Strength limit of earth: a new fundamental phenomenal factor for universe's precise scientific perceptions and descriptions. Open Access Library Journal Dec 19, 2017;4:e4034, http://www.scirp.org/Journal/PaperInformation.aspx?PaperID=81158 & https://doi.org/10.4236/oalib.1104034.

& http://www.oalib.com/paper/pdf/5290768 in Scientific Research Publishing Website, Downloads at OALib: http://www.oalib.com/articles/5290768 & http://dx.doi.org/10.4236/oalib.1104034

- 248.Sandbak LA. Roadheader drift excavation and geomechanics rock classification. In: Proc. Rapid Excav. Tunneling Conf., 1985, AIME, Vol. 2. New York: AIME, 1985:902-916.
- 249.Saranaathan SE. Different research techniques and models in rock mass rating and slope stability analysis. Review Article, Journal of Chemical and Pharmaceutical Research 2015;7(7):160-168. http:// www.jocpr.com.
- 250.Sarma SK. Stability analysis of embankment and slopes. Géotechnique 1973;23:423-33.
- 251.Seed HB. Considerations in the earthquake-resistant design of earth and rockfill dams. Geotechnique 1979;29(3):215-263.
- 252.Seed RB, Mitchell JK, Seed HB. Kettlemen Hills waste landfill slope failure, II: stability analysis. ASCE, Journal of the Geotechnical Engineering 1990;116(4):669-690.
- 253.Selby M.J. A rock mass strength classification for geomorphic purposes: with tests from Antarctica and New Zealand. Zeitschrift für Geomorphologie 1980;23:31–51.
- 254.Selby MJ. A rock mass strength classification for geomorphic purposes: with tests from Antarctica and New Zealand. J Geomorph. NF 1980;24(1):31–51.
- 255.Selby MJ. Controls on the stability and inclination of hill slopes formed on hard rock. Earth Surface Processes and Landforms 1982;7(5):449 467.
- 256.Selby MJ. Hill Slope Materials and Processes, Oxford: Oxford University Press, 1982:264 pages.
- 257.Selby MJ. Rock Slopes in Slope Stability. In: Anderson MG, Richards KS. eds. New York: John Wiley, 1987.
- 258.Sepúlveda SA, Murphy W, Petley DN. Topographic controls on co-seismic rock slides during the 1999 Chi-Chi earthquake, Taiwan. Quarterly Journal of Engineering and Hydrogeology 2005a;38:189-196.
- 259.Sepúlveda SA, Murphy W, Jibson RW, Prtley DN. Seismically induced rock slope failures resulting from topographic amplification of strong ground motions: The case of Pacoima Canyon, California. Engineering Geology 2005b;80:336-348.
- 260.Serafim JL, Pereira JP. Considerations of the geomechanics classification of Bieniawski. In: Proceedings of the International Symposium on Engineering Geology and Underground Const., 1983, LNEC, Lisbon, Portugal, Vol. 1, 1983:33-42.
- 261.Sharma Rahul, Rai Rajesh, Shrivastva BK. Dump slope rating for Indian coal mining. Research Journal of Mining 2017;1(1):12-26.
- 262. Sheorey PR. Empirical Rrock Failure Criteria. Rotterdam, Netherlands: A.A. Balkema, 1997.
- 263.Sjöberg J. Analysis of Large Scale Rock Slopes. PhD Dissertation, LULEÅL University of Technology, 1999. [Dissertation]
- 264.Shuk T. Key elements and applications of the natural slope methodology (NSM) with some emphasis on slope stability aspects. In: Proceedings of the 4th South American Congress on Rock Mechanics 2, 1994, Santiago de Chile. ISRM, Rotterdam: Balkema, 1994:255-266.
- 265.Siad L. Seismic stability analysis of fractured rock slopes by yield design theory. Soils dynamics and earthquake engineering 2003;23:203-212.
- 266.Singh A. FRHI-a system to evaluate and mitigate rock fall hazard instable rock excavations. Journal Division Civil Engineering Instrument Engineering (J. Div. Civ. Eng. Inst. Eng.) 2004;85:62-75, India.
- 267.Singh A, Connolly MJ. VRFSR: an empirical method for determining volcanic rock excavation safety on construction sites. Journal of the Institution of Engineers 2003;84:176-191, India, Civil Engineering Division, Institution of Engineers, Calcutta, INDIA.
- 268.Singh B, Goel RK. Rock Mass Classification: A Practical Approach in Civil Engineering, 1st edition. Amsterdam, Netherlands/New York, US/Oxford, UK: The Elsevier Science Ltd.; 1999, 2000: 1-267, ISBN-13: 978-0-08-043013-3, ISBN-10: 0-08-043013-9.
- 269.Singh B, Goel R.K. Tunnelling in Weak Rocks, 1st edition. Vol 5. Netherlands: Elsevier Science Ltd., Elsevier Geo-engineering Book Series, 2006:489, ISBN 13: 978-0-08-044987-6, ISBN 10: 0-08-044987-5.
- 270.Singh RN, Brown DJ, Denby B, Croghan JA. The development of a new approach to slope stability assessment in U.K. surface coal mines. In: Symposium on Ground Control in Coal Mines, Au S. I. M. M., 1986:57–63.
- 271.Singh RN, Denby B, Brown DJ. The characteristics of coal measures slope instability in British surface coal mines. In: 26th Symposium on Rock Mechanics, Rotterdam: Balkema, 1985:41–48.
- 272.Singh RN, Gahrooee DR. Application of rock mass weakening coefficient for stability assessment of slopes in heavily jointed rock mass. International Journal of Surface Mining, Reclamation and Environment 1989;3(4):207-219, Taylor and Francis, Published online: 25 Apr 2007.
- 273.Singh RN, Reed SM, Hughes DB. Research into the stability of restored opencast coal mine sites in the north east of England. International Journal of Surface Mining, Reclamation and Environment 1987;1(4):257-266, Taylor and Francis, Published online: 25 Apr 2007, http://dx.doi.org/10.1080/09208118708944128.

- 274.Singh TN. Assessment of coal mine waste dump behaviour using numerical modeling. In: Fuenkajorn & Phienwej, eds. Rock Mechanics 2011:25-36.
- 275.Slope stability analysis 2015. Wikipedia, Available at: http://en.wikipedia.org/wiki/Slope_stability_analysis [Date accessed: 12 March 2015].
- 276.SLOPE/W Slope Stability Analysis 2009. Calgary, Canada: Geo-Slope International, retrieved 20 July 2009. Available at: <u>http://www.geo-slope.com</u> [Date accessed: 12 March 2015].
- 277.Smith HJ. Estimating rippability of rock mass classification. In: Proceedings of The 27th U.S. Symposium on Rock Mechanics, 1986, University of Alabama, 1986:443-448.
- 278.Sonmez H, Ulusay R, Gokceoglu C. A practical procedure for the back analysis of slope failures in closely jointed rock masses. International Journal of Rock Mechanics & Mining Science 1998;35(2):219-233.
- 279.Soren Masa. Effects of Faults on Stability of Slopes In Open Cast Mines. Bachelor of Technology Thesis in Mining Engineering, Rourkela, India: Department of Mining Engineering, National Institute of Technology, 2010:7. [Dissertation]
- 280.Spencer E. A method of analysis of the stability of embankments assuming parallel inter-slice forces. Geotechnique 1967;17(1):11-26.
- 281.SRK Consulting Engineers and Scientists, Final report of Sungun Copper project; mining, geotechnics and slope design studies. December 2008:1, 2, 4, 5, 11, 18, 19, 29, 44, 46, 53, 54, 55, 65, 77, 106, Appendix B3-6.
- 282.Stacey TR. Stability of rock slopes in mining and civil engineering situations. Pretoria, South Africa: National Mechnical Engineering Research Institute; 1973.
- 283.Stacey TR, Xianbin X, Armstrong R, Keyter GJ. New slope stability considerations for deep open pit mines. The Journal of The South Africa Institute of Mining and Metallurgy 2003;373-380.
- 284.Standard gravity 2015. Wikipedia, Available at: <u>http://en.wikipedia.org/wiki/Standard_gravity</u>, [Date accessed: 12 March 2015].
- 285.Stead D, Eberhardt E, Coggan JS. Developments in the characterization of complex rock slope deformation and failure using numerical modeling techniques. Engineering Geology 2006;83:217-235.
- 286.Stead D, Eberhardt E, Coggan J, Benko B. Advanced numerical techniques in rock slope stability analysis applications and limitations. Landslides 2001;615-624.
- 287.Stewart DP, Adhikary DP, Jewell RJ. Studies on the stability of model rock slopes. Report G:1108. Australia: The University of Western Australia; 1994.
- 288.Stille H, Groth T, Fredriksson A. FEM-analysis of rock mechanical problems with JOBFEM. Report no. 307. Stiftelsen Bergteknisk Forskning – BeFo, Stockholm, Sweden, 1/82, 1982.
- 289.Sun G, Zheng H, Jiang W. A global procedure for evaluating stability of three-dimensional slopes. Natural Hazards 2012;61(3):1083-1098.
- 290. Taheri A, Tani K. Rock slope design using slope stability rating (SSR)? Application and field verifications. In: Proceedings of the 1st Canada-US Rock Mechanics Symposium-Rock Mechanics Meeting Society's Challenges and Demands, May 2007, Two Volume Set. Taylor&Francis, 2007:215–221.
- 291.Taheri A, Tani K. Assessment of the stability of rock slopes by the slope stability rating classification system. Rock Mechanics and Rock Engineering 2010;43(3):321–333.
- 292. Tahmasebi P, Hezarkhani A, Mortazavi M. Application of discriminant analysis for alteration separation; Sungun Copper Deposit, East Azerbaijan, Iran. INSInet Publication, Australian Journal of Basic and Applied Sciences, 2010;6(4):565, ISSN: 1991-8178, available at: <u>http://amirkabir.academia.edu/PejmanTahmasebi/Papers/178690/Application_of_Discriminant_Analysis_for_Al</u> <u>teration_Separation_Sungun_Copper_Deposit_East_Azerbaijan_Iran</u>.
- 293.Terzaghi K. Rock defects and loads on tunnel supports. In: Proctor, RV, White TL, eds., Vol. 1: Rock Tunneling With Steel Supports. Youngstown, OH: Commercial Shearing and Stamping Company, 1946:17-99.
- 294. The international system of units, 2001edition. In: Taylor Barry N, ed. The International System of Units (SI), Vol. 330. NIST Special Publication, 2001:29, 57, Available at: www.physics.nist.gov/document/sp330.html
- 295.Ugai K. Three-dimensional stability analysis of vertical cohesive slopes. Soils and Foundations 1985;25(3):41-48.
- 296.Ugai K. Three dimensional slope stability analysis by slice methods. In: Proceedings of the 6th International Conference on Numerical Methods in Geomechanics, 1988, Innsbruck, Austria, Rotterdam: Balkema, 1988:1369-1374.
- 297.Unal E. Design Guidelines and Roof Control Standards for Coal Mine Roofs. Ph.D. Dissertation, University Park, PA: Pennsylvania State University, 1983:355. [Dissertation]
- 298. Ünal E. Modified rock mass classification: M-RMR system. Milestone in rock engineering: A Jubilee Collection; Bieniawski ZT, editor. Rotterdam, Balkema, 1996:203-223.

Available Online: http://www.easpublisher.com/easmb/

- 299. Varnes DJ. Slope movement types and processes. In: Schuster RL, Krizek RJ. editors. Landslides: Analysis and Control, National Research Council Transportation Research Board Special Report, Special Report 176, Washington, D.C.: National Research Council; 1978. p. 11- 33.
- 300.Varnes DJ. Landslide hazard zonation: a review of principles and practice. Natural Hazards 1984;3:63 pages, UNESCO, Paris.
- 301. Vecchia O. A simple terrain index for the stability of hillsides or scarps. Large Ground Movements and Structures. Geddes JD, ed. publ. Pentech Press Ltd., 1978:p. 449-461.
- 302. Veiga Pinto A, Neves EM. Prediction of Beliche dam behaviour during reservoir filling. In: Proceeding of The Eleventh International Conference on Soil Mechanics and Foundation Engineering, 1985, Vol. 4.
- 303.Venkateswarlu V. Geomechanics Classification of Coal Measure Rocks Via Roof Supports. Ph.D. Thesis, Dhanbad: Indian school of mines, 1986: 251. [Dissertation]
- 304. Verma Dhananjai, Thareja Rahul, Kainthola Ashutosh, Singh TN. Evaluation of open pit mine slope stability analysis. International Journal of Earth Sciences and Engineering August 2011;04(04):590-600, ISSN 0974-5904.
- 305.Wang C, Tannant DD, Lilly PA. Numerical analysis of stability of heavily jointed rock slopes using PFC2D. International Journal of Rock Mechanics & Mining Sciences 2003;40:415-424.
- 306.Wang J, Tan W, Feng S, Zhou R. Reliability analysis of an open pit coal mine slope. International Journal of Rock Mechanics & Mining Sciences 2000;37:715-721.
- 307.Weaver J. Geological factors significant in the assessment of rippability. Transactions of the South African Institute of Civil Engineers (S. Afr. Civ. Eng.) December 1975;17(12):313–316.
- 308. Weinberg Steven. Gravitation and cosmology. New York: John Wiley, 1972, sec. 2.8.
- 309.Wickham GE, Tiedemann HR, Skinner EH. Support determinations based on geologic predictions. In: Lane KS, Garfield LA, editor. Proceedings of the 1st North American Rapid Excavation and Tunnelling Conference (RETC), 1972, Chicago, Vol. 1. New York: American Institute of Mining, Metallurgical and Petroleum Engineers (AIME), 1972:43–64.
- 310.Wickham GE, Tiedemann HR, Skinner EH. Ground support prediction model RSR concept. In: Proc. 2nd North American Rapid Excavation & Tunnelling Conference (RETC), 1974, San Francisco, Vol 1. New York: American Institute of Mining, Metallurgical and Petroleum Engineers (AIME), 1974:691-707.
- 311.Williams David R. Earth fact sheet (NASA Goddard Space Flight Center, Greenbelt, MD 20771), 1 September 2004 Last Updated: 19 May 2016, Last accessed: May 2016, Available at: http://nssdc.gsfc.nasa.gov/planetary/factsheet/earthfact.html.
- 312. Williamson DA. Uniform rock classification for geotechnical engineering purposes. Transportation Research Record 783, National Academy of Sciences, Washington D.C, 1980:pp. 9-14.
- 313. Williamson DA. Unified rock classification system. Bull Assoc. Eng. Geol. 1984;XXI(3):345–354.
- 314.Williamson DA, Kuhn CR. The unified classification system. Rock Engineering Systems for Engineering Purposes, In: Proc. Symp. Rock Classification Systems for Engineering Purposes, 1988, ASTM STP (Special Technical Publication) 984. Philadelphia: American Society for Testing Materials, 1988:7-16.
- 315.Witthaus H. Entwicklung Einer Planungs-methode Für Ankerausbau im Deutschen Steinkohlen-bergbau unter Berücksichtigung der Gebirgsstruktur und der Gebirgsbeanspruchung. Aachen, Germany: Verlag Mainz, 2006. [Dissertation] (In German).
- 316. Wyllie DC, Mah CW. Rock Slope Engineering. 4th edition. London: Spon Press, 2004.
- 317.Xing Z. Three-dimensional stability analysis of concave slopes in plan view. Journal of Geotechnical Engineering 1988a;114(6):658-671.
- 318.Xing Z. Closure: three-dimensional stability analysis of concave slopes in plan view. Journal of Geotechnical Engineering 1988b;116(2):345-346.
- 319. Yamagami T, Jiang JC. Determination of the sliding direction in three-dimensional slope stability analysis. In: Proceedings of the 2nd International Conference on Soft Soil Engineering, 1996, Part 1. Nanjing: Hohai University Press, 1996:567-572.
- 320. Yamagami T, Jiang JC. A search for the critical slip surface in three-dimensional slope stability analysis. Soils and Foundations 1997;37(3):1-16.
- 321.Yang XL. Seismic displacement of rock slopes with nonlinear Hoek-Brown failure criterion. International Journal of Rock Mechanics & Mining Sciences 2007;44:948-953.
- 322. Yang XL, Li L, Yin JH. Stability analysis of rock slopes with a modified Hoek–Brown failure criterion. International Journal for Numerical and Analytical Methods in Geomechanics 2004a;28:181-190.
- 323.Yang XL, Li L, Yin JH. Seismic and static stability analysis for rock slopes by a kinematical approach. Geotechnique 2004b;54(8):543-549.
- 324. Yang XL, Zou JF. Stability factors for rock slopes subjected to pore water pressure based on the Hoek-Brown failure criterion. International Journal of Rock Mechanics & Mining Science 2006;43:1146-1152.

- 325.Yarahmadi Bafghi AR, Verdel T. Sarma-based key-group method for rock slope reliability analyses. International Journal for Numerical and Analytical Methods in Geomechanics 2005;29:1019-1043.
- 326.Yu MH, Zan YW, Zhao J, Yoshimine M. A unified strength criterion for rock material. International Journal of Rock Mechanics & Mining Science 2002;39:975-989.
- 327.Yudhbir Lemanza W, Prinzl F. An empirical failure criterion for rock masses. In: Proceedings of fifth congress of ISRM, Vol.1, 1983:10-18.
- 328.Zanbak C. Design Charts for rock slopes susceptible to toppling. Journal of Geotechnical Engineering ASCE 1983;190(8):1039-1062.
- 329.Zheng H. Eigenvalue problem from the stability analysis of slopes. Journal of Geothechnical and Geoenvironm ental Engineering 2009;35(5):647-656.
- 330.Zienkiewicz OC, Taylor RL. The Finite Element Method, 5th edition. Vol. 1: The Basis, Oxford: Butterworth-Heinemann, 2000.
- 331.Zolfaghari AR, Heath AC, McCombie PF. Simple genetic algorithm search for critical noncircular failure surface in slope stability analysis. Comput. Geotech. 2005;32(3):139–152.

APPENDIXES

Table A.1: Calculated RMi values for assessing zone via borehole 15 (GT15).

From	to	Length	GSI	S	а	$JP = S^a$	σ	RMi	RMi
		U					č		Classifications
0	1.6	1.6	0	1.49453E-05	0.666454561	0.000608173	82	0.04987	Low
1.6	3.3	1.7	0	1.49453E-05	0.666454561	0.000608173	82	0.04987	Low
3.3	4.2	0.9	23	0.000192473	0.535757075	0.010217029	82	0.837796	Medium
4.2	4.8	0.6	14	7.08068E-05	0.565328015	0.004507452	82	0.369611	Medium
4.8	6.4	1.6	20	0.000137913	0.543720751	0.007962014	82	0.652885	Medium
6.4	6.6	0.2	24	0.000215092	0.533437314	0.011058214	82	0.906774	Medium
6.6	7.2	0.6	61	0.013123729	0.502643629	0.113253978	82	9.286826	High
7.2	8.4	1.2	26	0.000268617	0.529236969	0.012887416	82	1.056768	High
8.4	9.9	1.5	15	7.91279E-05	0.561101135	0.004995147	82	0.409602	Medium
9.9	11	1.1	14	7.08068E-05	0.565328015	0.004507452	82	0.369611	Medium
11	12.1	1.1	0	1.49453E-05	0.666454561	0.000608173	82	0.04987	Low
12.1	12.7	0.6	15	7.91279E-05	0.561101135	0.004995147	82	0.409602	Medium
12.7	13.5	0.8	16	8.8427E-05	0.557146859	0.005516358	82	0.452341	Medium
13.5	14.7	1.2	20	0.000137913	0.543720751	0.007962014	82	0.652885	Medium
14.7	16.1	1.4	28	0.000335463	0.525560938	0.014928435	82	1.224132	High
16.1	17.4	1.3	29	0.000374886	0.523898757	0.016035038	82	1.314873	High
17.4	19.2	1.8	39	0.001138803	0.512166824	0.03107496	82	2.548147	High
19.2	19.8	0.6	30	0.000418942	0.522343775	0.01720296	82	1.410643	High
19.8	21.4	1.55	34	0.000653392	0.517064082	0.022554893	82	1.849501	High
21.4	22	0.6	31	0.000468176	0.520889078	0.018435393	82	1.511702	High
22	23.3	1.35	42	0.001589327	0.509922905	0.037396812	82	3.066539	High
23.3	24.7	1.4	38	0.001019045	0.513020216	0.029183866	82	2.393077	High
24.7	25.6	0.9	28	0.000335463	0.525560938	0.014928435	82	1.224132	High
25.6	26.2	0.6	12	5.66977E-05	0.574676055	0.003628069	82	0.297502	Medium
26.2	27	0.8	23	0.000192473	0.535757075	0.010217029	82	0.837796	Medium
27	28.1	1.05	26	0.000268617	0.529236969	0.012887416	82	1.056768	High
28.1	29.7	1.6	31	0.000468176	0.520889078	0.018435393	82	1.511702	High
29.7	30.9	1.25	38	0.001019045	0.513020216	0.029183866	82	2.393077	High
30.9	31.2	0.3	39	0.001138803	0.512166824	0.03107496	105	3.262871	High
31.2	33.6	2.4	45	0.002218085	0.508085739	0.044825935	105	4.706723	High
33.6	35.2	1.55	37	0.000911882	0.513932441	0.027391375	105	2.876094	High
35.2	36.1	0.9	36	0.000815988	0.514907553	0.025692277	105	2.697689	High
36.1	37.2	1.15	34	0.000653392	0.517064082	0.022554893	105	2.368264	High
37.2	38	0.8	36	0.000815988	0.514907553	0.025692277	105	2.697689	High
38	39.6	1.6	34	0.000653392	0.517064082	0.022554893	105	2.368264	High
39.6	40.3	0.65	30	0.000418942	0.522343775	0.01720296	105	1.806311	High
40.3	41.2	0.95	35	0.000730178	0.515949889	0.024081659	105	2.528574	High

Mohammad Salehi Alashti.; East African S	cholars Multidiscip Bull	; Vol-1, Iss-3 (Dec	, 2018): 61-121
--	--------------------------	---------------------	-----------------

41.2	41.6	0.4	27	0.000300185	0.527337709	0.013880161	82	1.138173	High
41.6	43.1	1.5	39	0.001138803	0.512166824	0.03107496	82	2.548147	High
43.1	43.2	0.1	0	1.49453E-05	0.666454561	0.000608173	105	0.063858	Low
43.2	44.8	1.55	62	0.014666017	0.502459454	0.119852259	105	12.58449	Very High
44.8	46.3	1.55	57	0.008414677	0.503516356	0.090203301	105	9.471347	High
46.3	47.8	1.45	65	0.020468076	0.501975182	0.141971959	105	14.90706	Very High
47.8	49.3	1.55	55	0.006737947	0.50404815	0.080440238	105	8.446225	High
49.3	50.8	1.5	61	0.013123729	0.502643629	0.113253978	105	11.89167	Very High
50.8	52.3	1.5	48	0.003095587	0.506581595	0.053561975	105	5.624007	High
52.3	52.4	0.1	27	0.000300185	0.527337709	0.013880161	105	1.457417	High
52.4	53.8	1.4	26	0.000268617	0.529236969	0.012887416	82	1.056768	High
53.8	55.3	1.45	36	0.000815988	0.514907553	0.025692277	82	2.106767	High
55.3	56.5	1.2	29	0.000374886	0.523898757	0.016035038	82	1.314873	High
56.5	58	1.55	16	8.8427E-05	0.557146859	0.005516358	82	0.452341	Medium
58	59.5	1.5	22	0.000172232	0.538236758	0.009421899	82	0.772596	Medium
59.5	60	0.5	21	0.00015412	0.540887388	0.008670843	82	0.711009	Medium
60	61.4	1.35	21	0.00015412	0.540887388	0.008670843	82	0.711009	Medium
61.4	62.4	1	14	7.08068E-05	0.565328015	0.004507452	82	0.369611	Medium
62.4	62.5	0.15	22	0.000172232	0.538236758	0.009421899	82	0.772596	Medium
62.5	63.9	1.35	26	0.000268617	0.529236969	0.012887416	82	1.056768	High
63.9	65.2	1.35	26	0.000268617	0.529236969	0.012887416	82	1.056768	High
65.2	66.6	1.35	37	0.000911882	0.513932441	0.027391375	82	2.246093	High
66.6	68.2	1.65	30	0.000418942	0.522343775	0.01720296	82	1.410643	High
68.2	69.9	1.65	22	0.000172232	0.538236758	0.009421899	82	0.772596	Medium
69.9	71.4	1.5	39	0.001138803	0.512166824	0.03107496	82	2.548147	High

Table A.1: Continued.

From	to	Length	GSI	S	а	$JP = S^a$	σ	RMi	RMi
									Classifications
71.4	73.1	1.7	37	0.000911882	0.513932441	0.027391375	82	2.246093	High
73.1	74.8	1.7	42	0.001589327	0.509922905	0.037396812	82	3.066539	High
74.8	76.4	1.65	23	0.000192473	0.535757075	0.010217029	82	0.837796	Medium
76.4	77.2	0.75	13	6.33607E-05	0.569846292	0.004052124	82	0.332274	Medium
77.2	79.2	2.05	0	1.49453E-05	0.666454561	0.000608173	82	0.04987	Low
79.2	79.6	0.35	0	1.49453E-05	0.666454561	0.000608173	82	0.04987	Low
79.6	80.4	0.85	28	0.000335463	0.525560938	0.014928435	82	1.224132	High
80.4	80.8	0.4	37	0.000911882	0.513932441	0.027391375	82	2.246093	High
80.8	81.1	0.25	0	1.49453E-05	0.666454561	0.000608173	82	0.04987	Low
81.1	81.8	0.75	24	0.000215092	0.533437314	0.011058214	82	0.906774	Medium
81.8	83.4	1.6	33	0.000584681	0.518255087	0.021107619	82	1.730825	High
83.4	84.2	0.8	17	9.88188E-05	0.553447606	0.006072306	82	0.497929	Medium
84.2	85.6	1.35	31	0.000468176	0.520889078	0.018435393	82	1.511702	High
85.6	86.8	1.2	56	0.007529784	0.503773393	0.085188218	82	6.985434	High
86.8	88.8	2	64	0.018315639	0.502125972	0.134189287	82	11.00352	Very High
88.8	89.8	1.05	47	0.002770053	0.507049947	0.050490969	82	4.140259	High
89.8	91.9	2.1	60	0.011743628	0.502840501	0.107008536	82	8.7747	High
91.9	92.8	0.9	50	0.00386592	0.50573356	0.060227219	82	4.938632	High
92.8	94.8	2	33	0.000584681	0.518255087	0.021107619	82	1.730825	High
94.8	95.8	1	48	0.003095587	0.506581595	0.053561975	82	4.392082	High
95.8	97.6	1.8	45	0.002218085	0.508085739	0.044825935	82	3.675727	High
97.6	98.8	1.2	37	0.000911882	0.513932441	0.027391375	82	2.246093	High
98.8	100	1.4	34	0.000653392	0.517064082	0.022554893	82	1.849501	High
100	102	1.5	39	0.001138803	0.512166824	0.03107496	82	2.548147	High
102	104	1.95	35	0.000730178	0.515949889	0.024081659	82	1.974696	High
104	105	1.15	35	0.000730178	0.515949889	0.024081659	82	1.974696	High

Mohammad Salehi Alashti.; East African Scholars Multidiscip Bull; Vol-1, Iss-3 (Dec, 2018): 61-121

105	107	1.0	22	0.000584681	0 518255087	0.021107610	82	1 730825	High
103	107	2.25	40	0.000384081	0.51136847	0.021107019	82	2 711755	High
107	109	1.8	40	0.001272634	0.51136847	0.033070186	82	2.711755	High
109	111	1.0	40	0.001272034	0.51150847	0.033070180	82	2.711733	High
111	115	2.4	51	0.00198483	0.508057785	0.042214911	82	5.22501	High
115	115	1.0	24	0.004520239	0.505550100	0.00364136	82	1.840501	High
115	110	0.8	34 72	0.000633392	0.517064082	0.022334893	82	1.849301	High Marrie High
110	11/	1.15	13	0.049787068	0.5010/1059	0.222414354	82	18.23798	Very High
11/	119	2	40	0.001272634	0.51136847	0.0330/0186	82	2./11/55	High
119	120	1	49	0.003459377	0.506143449	0.050804152	82	4.05/939	High
120	122	2	44	0.00198483	0.508657785	0.042214911	82	3.461623	High
122	123	1	48	0.003095587	0.506581595	0.053561975	82	4.392082	High
123	124	1.1	31	0.000468176	0.520889078	0.018435393	82	1.511702	High
124	125	1.6	50	0.00386592	0.50573356	0.060227219	82	4.938632	High
125	127	1.65	43	0.001776104	0.509269267	0.039740994	82	3.258762	High
127	129	1.65	49	0.003459377	0.506143449	0.056804132	82	4.657939	High
129	130	1.5	27	0.000300185	0.527337709	0.013880161	82	1.138173	High
130	132	1.5	46	0.002478752	0.507550586	0.047581868	82	3.901713	High
132	133	1.5	38	0.001019045	0.513020216	0.029183866	82	2.393077	High
133	135	1.55	45	0.002218085	0.508085739	0.044825935	82	3.675727	High
135	136	1.55	37	0.000911882	0.513932441	0.027391375	82	2.246093	High
136	138	1.4	47	0.002770053	0.507049947	0.050490969	82	4.140259	High
138	139	1.2	37	0.000911882	0.513932441	0.027391375	82	2.246093	High
139	139	0.35	45	0.002218085	0.508085739	0.044825935	82	3.675727	High
139	140	0.7	31	0.000468176	0.520889078	0.018435393	82	1.511702	High
140	141	0.75	24	0.000215092	0.533437314	0.011058214	82	0.906774	Medium
141	141	0.6	30	0.000418942	0.522343775	0.01720296	82	1.410643	High
141	142	0.9	32	0.000523195	0.519528199	0.019735738	82	1.618331	High
142	144	1.5	35	0.000730178	0.515949889	0.024081659	82	1.974696	High
144	145	1.6	38	0.001019045	0.513020216	0.029183866	82	2.393077	High
145	147	1.4	46	0.002478752	0.507550586	0.047581868	82	3.901713	High
147	149	2.05	62	0.014666017	0.502459454	0.119852259	82	9.827885	High
149	150	0.95	49	0.003459377	0.506143449	0.056804132	82	4.657939	High
150	151	1.3	41	0.001422193	0.510621603	0.035175401	82	2.884383	High
151	153	1.7	36	0.000815988	0.514907553	0.025692277	82	2.106767	High
153	154	1.6	46	0.002478752	0.507550586	0.047581868	82	3.901713	High
154	155	1.15	28	0.000335463	0.525560938	0.014928435	82	1.224132	High
155	157	1.95	34	0.000653392	0.517064082	0.022554893	82	1.849501	High
157	159	1.3	47	0.002770053	0.507049947	0.050490969	82	4.140259	High
159	160	1.5	36	0.000815988	0.514907553	0.025692277	82	2.106767	High
160	162	1.5	59	0.01050866	0.503050945	0.101096828	82	8.28994	High

Table A.1: Continued.

From	to	Length	GSI	S	Α	$JP = S^{a}$	σ	RMi	RMi
									Classifications
162	164	2	48	0.003095587	0.506581595	0.053561975	82	4.392082	High
164	166	2.1	31	0.000468176	0.520889078	0.018435393	82	1.511702	High
166	168	1.8	39	0.001138803	0.512166824	0.03107496	82	2.548147	High
168	169	1.55	53	0.005395326	0.504655794	0.071688517	82	5.878458	High
169	171	1.5	44	0.00198483	0.508657785	0.042214911	82	3.461623	High
171	172	1.6	41	0.001422193	0.510621603	0.035175401	82	2.884383	High
172	174	1.45	37	0.000911882	0.513932441	0.027391375	82	2.246093	High
174	176	2	41	0.001422193	0.510621603	0.035175401	82	2.884383	High
176	177	1	42	0.001589327	0.509922905	0.037396812	82	3.066539	High
177	178	1.6	42	0.001589327	0.509922905	0.037396812	82	3.066539	High
178	180	1.45	48	0.003095587	0.506581595	0.053561975	82	4.392082	High
180	181	1.6	43	0.001776104	0.509269267	0.039740994	82	3.258762	High

Mohammad Salehi Alashti.; East African Scholars Multidiscip Bull; Vol-1, Iss-3 (Dec, 2018): 61-121

181	183	1.35	45	0.002218085	0.508085739	0.044825935	82	3.675727	High
183	184	1.5	35	0.000730178	0.515949889	0.024081659	82	1.974696	High
184	186	1.5	33	0.000584681	0.518255087	0.021107619	82	1.730825	High
186	187	1.55	52	0.00482795	0.504991382	0.067658156	82	5.547969	High
187	189	1.5	57	0.008414677	0.503516356	0.090203301	82	7.396671	High
189	190	1.5	39	0.001138803	0.512166824	0.03107496	82	2.548147	High
190	192	1.45	50	0.00386592	0.50573356	0.060227219	82	4.938632	High
192	193	0.9	48	0.003095587	0.506581595	0.053561975	82	4.392082	High
193	193	0.6	69	0.031922492	0.5014632	0.177770456	130	23.11016	Very High
193	195	1.43	55	0.006737947	0.50404815	0.080440238	130	10.45723	Very High
195	195	0.07	40	0.001272634	0.51136847	0.033070186	82	2.711755	High
195	197	2	36	0.000815988	0.514907553	0.025692277	82	2.106767	High
197	198	1	36	0.000815988	0.514907553	0.025692277	82	2.106767	High
198	199	1.25	45	0.002218085	0.508085739	0.044825935	82	3.675727	High
199	200	0.75	40	0.001272634	0.51136847	0.033070186	130	4.299124	High
200	201	1	91	0.367879441	0.500174376	0.606424905	130	78.83524	Very High
201	203	2.7	77	0.077649082	0.500770706	0.278107553	130	36.15398	Very High
203	206	2.7	53	0.005395326	0.504655794	0.071688517	82	5.878458	High
206	208	1.7	49	0.003459377	0.506143449	0.056804132	82	4.657939	High
208	210	1.9	50	0.00386592	0.50573356	0.060227219	82	4.938632	High
210	212	2	40	0.001272634	0.51136847	0.033070186	82	2.711755	High
212	213	1.75	41	0.001422193	0.510621603	0.035175401	82	2.884383	High
213	215	1.2	32	0.000523195	0.519528199	0.019735738	82	1.618331	High
215	217	1.95	29	0.000374886	0.523898757	0.016035038	82	1.314873	High
217	218	1.1	34	0.000653392	0.517064082	0.022554893	82	1.849501	High
218	219	1	37	0.000911882	0.513932441	0.027391375	82	2.246093	High
219	220	1	27	0.000300185	0.527337709	0.013880161	82	1.138173	High
220	221	1.05	42	0.001589327	0.509922905	0.037396812	82	3.066539	High
221	222	1.5	53	0.005395326	0.504655794	0.071688517	82	5.878458	High
222	223	1.2	35	0.000730178	0.515949889	0.024081659	82	1.974696	High
223	224	0.55	30	0.000418942	0.522343775	0.01720296	82	1.410643	High
224	226	1.8	34	0.000653392	0.517064082	0.022554893	82	1.849501	High
226	227	0.8	22	0.000172232	0.538236758	0.009421899	82	0.772596	Medium
227	228	1.2	24	0.000215092	0.533437314	0.011058214	82	0.906774	Medium
228	228	0.25	18	0.000110432	0.54998693	0.006664289	82	0.546472	Medium
228	230	1.5	34	0.000653392	0.517064082	0.022554893	82	1.849501	High
230	231	1.25	28	0.000335463	0.525560938	0.014928435	82	1.224132	High
231	232	1	22	0.000172232	0.538236758	0.009421899	82	0.772596	Medium
232	234	1.9	35	0.000730178	0.515949889	0.024081659	82	1.974696	High
234	235	1.5	38	0.001019045	0.513020216	0.029183866	82	2.393077	High
235	237	1.5	36	0.000815988	0.514907553	0.025692277	82	2.106767	High
237	238	1.4	35	0.000730178	0.515949889	0.024081659	82	1.974696	High
238	240	1.6	38	0.001019045	0.513020216	0.029183866	82	2.393077	High
240	241	1	21	0.00015412	0.540887388	0.008670843	82	0.711009	Medium
241	242	1.7	22	0.000172232	0.538236758	0.009421899	82	0.772596	Medium
242	243	0.6	38	0.001019045	0.513020216	0.029183866	82	2.393077	High
243	244	1.2	31	0.000468176	0.520889078	0.018435393	82	1.511702	High
244	246	1.5	46	0.002478752	0.507550586	0.047581868	82	3.901713	High
246	247	1.5	29	0.000374886	0.523898757	0.016035038	82	1.314873	High
247	249	1.5	19	0.00012341	0.546749443	0.007293695	82	0.598083	Medium
249	250	1.2	28	0.000335463	0.525560938	0.014928435	82	1.224132	High
250	251	1.4	26	0.000268617	0.529236969	0.012887416	82	1.056768	High
251	253	1.55	41	0.001422193	0.510621603	0.035175401	82	2.884383	High
253	254	1.35	30	0.000418942	0.522343775	0.01720296	82	1.410643	High
			1		-			-	

	Table A.1: Continued.									
From	to	Length	GSI	S	а	$JP = S^{a}$	σ	RMi	RMi	
									Classifications	
254	256	1.5	26	0.000268617	0.529236969	0.012887416	82	1.056768	High	
256	257	0.9	21	0.00015412	0.540887388	0.008670843	82	0.711009	Medium	
257	257	0.7	31	0.000468176	0.520889078	0.018435393	82	1.511702	High	
257	259	1.9	32	0.000523195	0.519528199	0.019735738	82	1.618331	High	
259	261	1.55	28	0.000335463	0.525560938	0.014928435	82	1.224132	High	
261	262	1.05	24	0.000215092	0.533437314	0.011058214	82	0.906774	Medium	
262	263	1.1	23	0.000192473	0.535757075	0.010217029	82	0.837796	Medium	
263	264	0.8	31	0.000468176	0.520889078	0.018435393	82	1.511702	High	
264	265	1.5	30	0.000418942	0.522343775	0.01720296	82	1.410643	High	
265	266	0.8	19	0.00012341	0.546749443	0.007293695	82	0.598083	Medium	
266	267	0.8	28	0.000335463	0.525560938	0.014928435	82	1.224132	High	
267	268	1	33	0.000584681	0.518255087	0.021107619	82	1.730825	High	
268	269	1	32	0.000523195	0.519528199	0.019735738	82	1.618331	High	
269	270	11	28	0.000335463	0.525560938	0.014928435	82	1.224132	High	
270	271	1.05	31	0.000468176	0.520889078	0.018435393	82	1.511702	High	
271	272	0.75	25	0.000240369	0.531267162	0.011947582	82	0.979702	Medium	
272	272	0.85	27	0.000300185	0.527337709	0.013880161	82	1.138173	High	
272	274	1.15	33	0.000584681	0.518255087	0.021107619	82	1.730825	High	
274	275	1.10	37	0.000911882	0 513932441	0.027391375	82	2 246093	High	
275	276	1.1	33	0.000584681	0.518255087	0.021107619	82	1 730825	High	
276	270	1.1	24	0.000215092	0 533437314	0.011058214	82	0.906774	Medium	
277	278	1.5	29	0.000374886	0.523898757	0.016035038	82	1 314873	High	
278	279	11	33	0.000584681	0.518255087	0.021107619	82	1.730825	High	
279	280	0.4	47	0.002770053	0 507049947	0.050490969	82	4 140259	High	
280	281	1.5	43	0.001776104	0 509269267	0.039740994	82	3 258762	High	
281	282	0.9	29	0.000374886	0.523898757	0.016035038	82	1.314873	High	
282	283	14	26	0.000268617	0 529236969	0.012887416	82	1.056768	High	
283	284	0.7	30	0.000418942	0 522343775	0.01720296	82	1 410643	High	
284	285	0.9	28	0.000335463	0.525560938	0.014928435	82	1 224132	High	
285	286	1.45	35	0.000730178	0.515949889	0.024081659	82	1.974696	High	
286	287	0.7	19	0.00012341	0 546749443	0.007293695	82	0 598083	Medium	
287	289	1.45	23	0.000192473	0.535757075	0.010217029	82	0.837796	Medium	
289	207	1.45	29	0.000172475	0.523898757	0.016035038	82	1 31/1873	High	
207	291	0.8	29	0.000374886	0.523898757	0.016035038	82	1 31/873	High	
290	291	1.4	29	0.000574880	0.525698757	0.022554803	82	1.314873	High	
291	292	1.4	28	0.0000335463	0.525560038	0.022334893	82	1.049301	Ligh	
292	293	1.1	20	0.000333403	0.523427214	0.014928433	82	0.006774	Madium	
293	294	0.95	24	0.000213092	0.535457514	0.011038214	82	1.224122	Ligh	
294	293	1.2	20	0.000333403	0.525500958	0.014926433	02 92	1.224132	Iliah	
293	290	0.93	29	0.0003/4880	0.323698/3/	0.010033038	02	1.3148/3	Tigli Ulah	
290	298	1.4	30 25	0.000720170	0.51490/553	0.023092277	82 82	2.100/0/	пign Ui-t	
298	298	0.8	33	0.000/301/8	0.515949889	0.024081659	82	1.9/4696	riign	
298	300	1.35	31	0.000468176	0.520889078	0.018435393	82	1.511702	High	
300	300	0.3	36	0.000815988	0.514907553	0.025692277	82	2.106767	Hıgh	

Mohammad Salehi Alashti.; East African Scholars Multidiscip Bull; Vol-1, Iss-3 (Dec, 2018): 61-121